# The prevalence, selected risk factors and burden of primary headaches in lecturers at a University of Technology (UoT) in KwaZulu-Natal during the COVID-19 pandemic

By Deidre Chetty

Dissertation submitted in partial compliance with the requirements for the Master's Degree in Health Sciences: Chiropractic Durban University of Technology

I, Deidre Chetty, do declare that this dissertation is representative of my own work in both conception and execution (except where acknowledgements indicate to the contrary).

Deidre	Chetty
	,

06/03/24

Date

Approved for Final Submission

Dr Penelope Orton PhD Nursing Supervisor 06/03/24

Date

Dr Laura O'Connor

M.Tech Chiropractic

Collaborator

06/03/24

Date

## DEDICATION

I dedicate this to my dad, mum, brother and sister. I would have never been able to accomplish my goals without your love and unyielding support. I am forever grateful to have such a phenomenal support structure. Thanks for everything — I love and appreciate you all dearly.

## ACKNOWLEDGEMENTS

I would like to express my heartfelt gratitude to the following individuals:

- My supervisor, Dr Penny Orton, thank you for being a beam of light and making this process so much easier. Your invaluable support, guidance and encouragement did not go unnoticed. You have always gone above and beyond to assist me with my dissertation. I appreciate everything; thank you.
- My collaborator, Dr Laura O'Connor thank you for your expertise and guidance, both in the clinic and my dissertation. You have played an integral role in my life over the past six years. I will forever be grateful to you.
- Dr Gill Hendry, thank you for your assistance with the statistical analysis.
- Ms Helen Bond, thank you for the proof reading and editorial expertise for this study.
- To all the research participants, thank you for filling out the questionnaire and participating in this study.
- Thank you to my extended family and friends for their continuous support and encouragement.

## ABSTRACT

**Background:** Headaches are a common disabling condition that affect many people worldwide. Headaches have been under-studied in developing countries and there is a paucity of literature documenting headaches in lecturers. The COVID-19 pandemic has forced lecturers to adapt to online teaching, which may have resulted in headaches. It is envisaged that this research will be beneficial to chiropractors, other healthcare providers and lecturers.

**Aim of Study**: The aim of this study was to determine the prevalence, selected risk factors and burden of primary headaches in lecturers at a university of technology (UoT) in KwaZulu-Natal during the COVID-19 pandemic.

**Methodology**: A quantitative, descriptive cross-sectional design, where 685 lecturers employed at the Durban University of Technology, through self-selection, were selected to participate, using consecutive sampling. A minimum sample size of 247 was calculated, using a confidence level of 95% and confidence interval of 5%. Lecturers from all six faculties were invited to participate. The headache-attributed restriction, disability, social handicap and impaired participation (HARDSHIP) questionnaire was administered electronically via QUESTIONPRO®. Each participant was required to initial a consent form prior to answering the questionnaire. The diagnostic algorithm was utilised to diagnose headaches according to the diagnostic questions in the HARDSHIP questionnaire. All data were captured on an Excel spreadsheet and subsequently analysed using SPSS version 28.

**Results**: There was a total of 255 completed questionnaires. The mean age of respondents was 45.66 years old. Of the total sample (n=255), 51.8% were headache sufferers. Most respondents suffered from probable tension-type headache (41.9%; n=54), followed by migraine-type headache (27.9%, n=36). Social risk factors revealed a significant relationship between alcohol and headaches, and sleep deprivation and headaches. Medial risk factors revealed a significant relationship between a head injury and headache and sinusitis and headache. Although there was a relationship between psychological risk factors and headache, it was not significant. Many work-related risk factors were significantly related to headache. There was a greater number of headache sufferers who

tested positive for COVID-19. Many respondents experienced loneliness and their workload increased during the pandemic. Most of the respondents' headaches interfered with work, and they usually took medication to treat their headache.

**Conclusion**: Headaches are common in lecturing staff with, tension type headaches being most prevalent. There are many risk factors which are strongly associated with headaches and its impact caused an interference with work and negatively affected the suffers mental health. Family, social or leisure activities are also neglected when a headache occurred. This study adds to the current literature on headache prevalence in the lecturer population and increases awareness.

# TABLE OF CONTENTS

DEDICATION	II
ACKNOWLEDGEMENTS	III
ABSTRACT	IV
TABLE OF CONTENTS	VI
LIST OF APPENDICES	XI
LIST OF TABLES	XII
LIST OF FIGURES	XIII
LIST OF ABBREVIATIONS	XIV
DEFINITIONS	XV
1.1 INTRODUCTION AND BACKGROUND	1
1.2 RESEARCH PROBLEM	
1.3 AIM OF THE STUDY	4
1.4 OBJECTIVES OF THE STUDY	4
1.5 SIGNIFICANCE OF THE STUDY	4
1.6 FLOW OF THE DISSERTATION	5
1.7 CONCLUSION	5
CHAPTER TWO LITERATURE REVIEW	7
2.1 INTRODUCTION	7
2.2 CLASSIFICATION OF HEADACHES	7
2.3 TYPES OF PRIMARY HEADACHES	8
2.3.1 Tension-Type Headache	8
2.3.2 Migraine-Type Headache	9
2.3.3 Trigeminal Autonomic Cephalalgias	10
2.4 PREVALENCE OF PRIMARY HEADACHES	

	2.5 RISK FACTORS FOR HEADACHES	. 13
	2.5.1 Social History	. 13
	2.5.2 Medical History	. 13
	2.5.3 Psychological	. 14
	2.5.4 Work-Related	. 14
	2.5.5 COVID-19	. 16
	2.6 CLINICAL CHARACTERISTICS OF PRIMARY HEADACHES	. 18
	2.6.1 Tension-Type Headache	. 18
	2.6.2 Migraine Type Headache	. 19
	2.6.3 Cluster Headache	. 19
	2.7 DIAGNOSIS	. 20
	2.8 MANAGEMENT OF PRIMARY HEADACHES	. 20
	2.9 BURDEN OF HEADACHES	. 22
	2.10 CONCLUSION	. 23
C	2.10 CONCLUSION	. 23 . <b>24</b>
C	2.10 CONCLUSION	. 23 . <b>24</b> . 24
C	2.10 CONCLUSION	. 23 <b>. 24</b> . 24 . 24
C	<ul> <li>2.10 CONCLUSION</li> <li>CHAPTER THREE METHODOLOGY</li> <li>3.1 INTRODUCTION</li> <li>3.2 STUDY DESIGN</li> <li>3.3 STUDY POPULATION AND SETTING</li> </ul>	. 23 . <b>24</b> . 24 . 24 . 24
C	<ul> <li>2.10 CONCLUSION</li> <li>CHAPTER THREE METHODOLOGY</li> <li>3.1 INTRODUCTION</li> <li>3.2 STUDY DESIGN</li> <li>3.3 STUDY POPULATION AND SETTING</li> <li>3.4 PARTICIPANT RECRUITMENT</li> </ul>	. 23 . <b>24</b> . 24 . 24 . 24 . 25
C	<ul> <li>2.10 CONCLUSION</li> <li>CHAPTER THREE METHODOLOGY</li> <li>3.1 INTRODUCTION</li> <li>3.2 STUDY DESIGN</li> <li>3.3 STUDY POPULATION AND SETTING</li> <li>3.4 PARTICIPANT RECRUITMENT</li> <li>3.5 PERMISSION</li> </ul>	. 23 . <b>24</b> . 24 . 24 . 24 . 25 . 25
C	2.10 CONCLUSION	. 23 . <b>24</b> . 24 . 24 . 24 . 25 . 25 . 25
C	2.10 CONCLUSION	. 23 . <b>24</b> . 24 . 24 . 25 . 25 . 25 . 25
C	2.10 CONCLUSION	. 23 . <b>24</b> . 24 . 24 . 25 . 25 . 25 . 25 . 25
C	2.10 CONCLUSION CHAPTER THREE METHODOLOGY	. 23 . <b>24</b> . 24 . 24 . 25 . 25 . 25 . 25 . 25 . 25
C	2.10 CONCLUSION CHAPTER THREE METHODOLOGY 3.1 INTRODUCTION 3.2 STUDY DESIGN 3.3 STUDY POPULATION AND SETTING 3.4 PARTICIPANT RECRUITMENT 3.5 PERMISSION 3.6 SAMPLING 3.6.1 Sample Size 3.6.2 Sample Characteristics 3.6.3 Sample Strategy 3.7 DATA COLLECTION.	. 23 . 24 . 24 . 24 . 25 . 25 . 25 . 25 . 25 . 26 . 26

	3.7.2 Administration of the Tool	26
	3.7.3 The Hardship Questionnaire	26
	3.7.4 Classification of Headaches	27
	3.8 VALIDATION OF THE QUESTIONNAIRE	27
	3.8.1 Focus/Expert Group	27
	3.8.2 Outcome of the Focus Group	28
	3.8.3 Pilot study	29
	3.8.4 Outcome of the Pilot Study	30
	3.9 DATA ANALYSIS	30
	3.9.1 Statistical Analysis	30
	3.10 DATA COLLECTION PROCEDURE	31
	3.11 ETHICAL CONSIDERATIONS	31
	3.12 CONCLUSION	32
C	HAPTER FOUR RESULTS	33
C	4.1 INTRODUCTION	<b>33</b> 33
C	4.1 INTRODUCTION         4.2 SAMPLE SIZE AND RESPONSE RATE	<b>33</b> 33 33
C	4.1 INTRODUCTION 4.2 SAMPLE SIZE AND RESPONSE RATE 4.3 DEMOGRAPHICS	<b>33</b> 33 33 33
C	<ul> <li>4.1 INTRODUCTION</li></ul>	<b>33</b> 33 33 33 33
C	<ul> <li>4.1 INTRODUCTION</li> <li>4.2 SAMPLE SIZE AND RESPONSE RATE.</li> <li>4.3 DEMOGRAPHICS</li> <li>4.3.1 Age</li> <li>4.3.2 Gender.</li> </ul>	33 33 33 33 33 33 34
C	<ul> <li>4.1 INTRODUCTION</li> <li>4.2 SAMPLE SIZE AND RESPONSE RATE.</li> <li>4.3 DEMOGRAPHICS</li> <li>4.3.1 Age</li> <li>4.3.2 Gender.</li> <li>4.3.3 Race</li> </ul>	33 33 33 33 33 33 34 34
C	<ul> <li>4.1 INTRODUCTION</li></ul>	33 33 33 33 33 33 34 34
C	<ul> <li>4.1 INTRODUCTION</li> <li>4.2 SAMPLE SIZE AND RESPONSE RATE.</li> <li>4.3 DEMOGRAPHICS</li> <li>4.3.1 Age</li> <li>4.3.2 Gender.</li> <li>4.3.3 Race</li> <li>4.3.4 Marital Status</li> <li>4.4 PREVALENCE OF HEADACHES.</li> </ul>	33 33 33 33 33 34 34 34 35
C	<ul> <li>4.1 INTRODUCTION</li> <li>4.2 SAMPLE SIZE AND RESPONSE RATE</li> <li>4.3 DEMOGRAPHICS</li> <li>4.3.1 Age</li> <li>4.3.2 Gender</li> <li>4.3.3 Race</li> <li>4.3.4 Marital Status</li> <li>4.4 PREVALENCE OF HEADACHES</li> <li>4.5 SCREENING</li> </ul>	33 33 33 33 33 33 34 34 34 35 36
C	<ul> <li>4.1 INTRODUCTION</li> <li>4.2 SAMPLE SIZE AND RESPONSE RATE.</li> <li>4.3 DEMOGRAPHICS</li> <li>4.3.1 Age</li> <li>4.3.2 Gender</li> <li>4.3.3 Race</li> <li>4.3.4 Marital Status</li> <li>4.4 PREVALENCE OF HEADACHES</li> <li>4.5 SCREENING</li> <li>4.6 DIAGNOSTIC QUESTIONS.</li> </ul>	33 33 33 33 33 34 34 34 35 35 36 37
C	<ul> <li>4.1 INTRODUCTION</li> <li>4.2 SAMPLE SIZE AND RESPONSE RATE.</li> <li>4.3 DEMOGRAPHICS</li> <li>4.3.1 Age</li> <li>4.3.2 Gender.</li> <li>4.3.3 Race</li> <li>4.3.4 Marital Status.</li> <li>4.4 PREVALENCE OF HEADACHES.</li> <li>4.5 SCREENING.</li> <li>4.6 DIAGNOSTIC QUESTIONS.</li> <li>4.7 IMPACT ON LIFE</li> </ul>	33 33 33 33 33 33 34 34 34 35 36 37 41

4.8.1 Social History Results.	
4.8.2 Medical History Results	6
4.8.2.1 Chronic illnesses	
4.8.3 Psychological Results.	
4.8.4 Work-Related Risk Fac	tors
4.8.5 COVID-19 Related Ris	k Factors51
4.9 CONCLUSION	
CHAPTER FIVE DISCUSSION	
5.1 DEMOGRAPHIC FACTOR	S 53
5.1.1 Age	
5.1.2 Sex	
5.1.3 Race	
5.1.4 Marital Status	
5.2 PREVALENCE OF HEADA	CHES 54
5.3 RISK FACTORS	
5.3.1 Social Risk Factors	
5.3.1.1 Smoking and Head	aches 55
5.3.1.2 Alcohol and Heada	ches 55
5.3.1.3 Sleep and Headacl	nes 56
5.3.1.4 Exercise and head	aches 56
5.3.2 Medical Risk Factors	
5.3.2.1 Head Injury and He	eadache57
5.3.2.2 Glasses and Head	ache 57
5.3.2.3 Blood pressure and	d headache 57
5.3.2.4 Sinusitis and Head	ache 58

5.3.4 Work-Related Risk Factors5	59
5.3.5 COVID-19 Related Risk Factors 5	59
5.4 BURDEN OF HEADACHE	59
5.5 CONCLUSION	30
CHAPTER SIX CONCLUSION, LIMITATIONS AND RECOMMENDATIONS 6	51
6.1 INTRODUCTION6	51
6.2 CONCLUSION	51
6.3 LIMITATIONS	32
6.4 RECOMMENDATIONS	32
6.4.1 Recommendation for Research into Headaches	32
6.4.2 Recommendation for the Durban University of Technology	33
REFERENCES6	<b>54</b>
APPENDICES9	<b>}7</b>

# LIST OF APPENDICES

Appendix 1: Permission from the Director of Research	97
Appendix 2: Ethics Clearance Certificate	98
Appendix 3: Permission Letter from Professor Timothy Steiner	99
Appendix 4: Letter of Information	100
Appendix 5: Informed Consent	102
Appendix 6: Questionnaire Focus Group	103
Appendix 7: Final Questionnaire	108
Appendix 8: Diagnostic Algorithm	114

# LIST OF TABLES

Table 2.1: Risk factors for migraine	10
Table 2.2: ICHD3: headache attributed to systemic viral infection	18
Table 4.1: Age of respondents	33
Table 4.2: Gender of respondents	34
Table 4.3: Race of respondents	34
Table 4.4: Marital status of respondents	35
Table 4.5: Headache sufferers	35
Table 4.6: Respondents who had a headache during the last 12 months	35
Table 4.7: Diagnosis of headaches	41
Table 4.8: Social risk factors	45
Table 4.9: Medical risk factors	46
Table 4.10: Psychological risk factors	48
Table 4.11: Work-related risk factors	49
Table 4.12: COVID-19 related risk factors	51

# LIST OF FIGURES

Figure 4.1: Screening questions and results	37
Figure 4.2: The relationship between the duration of headache and medication	38
Figure 4.3: Severity of headache	39
Figure 4.4: Cluster-type headache question	39
Figure 4.5: Cluster-type headache question	39
Figure 4.6: Cluster-type headache question	40
Figure 4.7: Type of pain	40
Figure 4.8: Type of pain	41
Figure 4.9: The impact of headaches	42
Figure 4.10: Missed workdays/family activities due to headaches	43
Figure 4.11: Missed workdays/family activities due to headaches	43
Figure 4.12: Relieving a headache at home VS at work	.44

# LIST OF ABBREVIATIONS

CAR:	Chimeric antigen receptor
CNS:	Central nervous system
Co2:	Carbon dioxide
COVID-19:	Coronavirus disease 2019
CRS:	Cytokine release syndrome
CTH:	Cluster-type headache
CTTH:	Chronic tension type headache
ETTH:	Episodic tension type headache
GBD:	Global Burden of Diseases
ICHD-3:	International Classification of Headache Disorders 3rd Edition
IHS:	International Headache Society
MTH:	Migraine-type headache
MP:	Mobile phone
NSAIDs:	Non-steroidal anti-inflammatory drugs
TAC:	Trigeminal autonomic cephalgia
TNF:	Tumour necrosis factor
TTH:	Tension-type headache
UoT:	University of Technology
WHO:	World Health Organization
WFH:	Working from home

## DEFINITIONS

#### Definition of a lecturer

The Cambridge Advanced Learners Dictionary defined a lecturer as an individual who teaches at a college or university (Murungi and Gitonga, 2015). According to Maboe (2019: 137), a lecturer is someone who is trained to formally facilitate teaching and learning for the purpose of acquiring knowledge. For the purposes of this study, the first definition will be used.

#### Definition of a headache

Headaches are said to be the most common medical complaint known to man (Farooq and Williams, 2008). The International Headache Society (IHS) (2018) defined a headache as pain located in the head, above the orbitomeatal line and/or nuchal ridge.

## **CHAPTER ONE INTRODUCTION**

### **1.1 INTRODUCTION AND BACKGROUND**

This chapter provides an overview of the background, research problem, aims and objectives. The significance of this study is then followed by a synopsis of what each chapter entails.

The IHS (2018) defined a headache as pain located in the head, above the orbitomeatal line and/or nuchal ridge. Headaches may also be defined as head pain which occurs in the head or neck area (Shuaibi *et al.*, 2021). It is said that headaches are one of the most common complaints in daily practice (Kurt and Kaplan, 2008), since they affect almost everyone at least once in their lifetime (Falavigna *et al.*, 2010).

Headache disorders are ranked in the top ten of the most disabling conditions according to the World Health Organisation (WHO) (Stovner *et al.*, 2007). It is estimated that almost half of the adult population have experienced a headache at least once within the last year (WHO, 2016). The global prevalence over periods of one month to a lifetime was found to be 58.4% (Abu-Arafeh *et al.*, 2010). The WHO ranks headache disorders in the top ten most disabling conditions for males and females and the top five for females only (Stovner *et al.*, 2007).

Headaches can be divided into primary and secondary headaches (Aurora, 2004; Ahmed, 2012). Secondary headaches are a result of a condition which causes inflammation or traction on pain-sensitive structures (Rizzoli and Mullally, 2017). Primary headaches (PHA) comprising of migraine, tension and cluster-type headaches (National Institute of Neurological Disorders and Stroke, 2016), are the most common and the most costly. These headache disorders are a diverse group of neurologic conditions which cause headaches but do not have an obvious cause (Robbins and Lipton, 2010). In adults, migraine type headaches (MTH) affect 12% of this population and are the most disabling, whereas tension-type headaches (TTH) affect 38% of the population (Robbins, 2021). Primary headaches make up approximately 90% of all headaches and are not life-threatening (Dell'Isola *et al.*, 2021). For the purposes of this study, only literature on primary headaches will be presented.

For adults under the age of 50 years, MTH is the leading cause of years lived with disability (Steiner *et al.*, 2018). A clinical picture of MTH includes four phases: prodrome, aura, headache phase and postdrome (Rizzoli and Mullally, 2017). More common than other headaches, TTHs is but not as severe as MTH and has a lifetime prevalence of up to 80% in the general population. Although TTH is not as severe as MTH, there is a significant

socioeconomic impact as there is a degree of disability and high frequency (Stovner *et al.,* 2007). Of the trigeminal autonomic cephalgias (TACs), cluster-type headaches (CTH) are the most common but are rare in the group of primary headache disorders (Villar-Martinez, Puledda and Goadsby, 2020).

Risk factors, such as gender, family history, smoking, climate and body mass index, have been associated with primary headache disorders (Xie *et al.*, 2020). Psychosocial risk factors have been shown to greatly affect the impact of headaches in various ways (Malmberg-Ceder *et al.*, 2019). A population-based study in the Netherlands found that psychiatric conditions, sleep problems and a history of head or neck injury were risk factors for chronic headaches (Diener *et al.*, 2019). The risk factors that will be discussed in the literature include social, medical, psychological, work-related and the coronavirus disease 19 (COVID-19) related factors.

There are many studies which have focused on the physical factors accounting for headaches (Winter *et al.*, 2011; Ahmed, 2012; Lucas *et al.*, 2014; Mahmoud *et al.*, 2018 Raucci *et al.*, 2020; Xie *et al.*, 2020). However, research has shown that biological factors alone do not account for all aspects of headache and disability; it is unfortunate that psychological factors are only given importance when patients present with major psychopathology (Nicholson *et al.*, 2007). During the COVID-19 pandemic, high stress levels, depression and anxiety were reported among teachers, which negatively impacted their teaching (Al Lily *et al.*, 2020), all of which may lead to headaches.

The burden that headaches impose on individuals include disability, pain, reduced quality of life, impairment of participation in work related and social activities, and heavy financial costs (Yu *et al.*, 2012). The diagnosis rate of headaches is 40%, leaving 60% undiagnosed (Steiner *et al.*, 2011). Most primary headache clients can be treated effectively by primary healthcare services, thus making an improvement in the diagnosis and treatment rate which has the potential to significantly reduce the burden of headaches (Vosoughi *et al.*, 2019).

The COVID-19 disease is a pathogenic viral infection, which is highly transmittable and caused severe acute respiratory syndrome coronavirus 2 (SARS-CoV-2). This virus caused a global pandemic that led to loss of human life worldwide (Shereen *et al.*, 2020). In South Africa, the first case was reported in March 2020 (Broadbent, Combrink and Smart, 2021). To curb the spread of the rapid transmission, a lockdown was implemented, which caused businesses and learning institutions to close. The impact of the COVID-19 pandemic forced South African schools and universities to adapt to online learning (Mpungose, 2020) during the acute phase of the pandemic. This transition to an online environment, working from home (WFH), and 'adapting to a new normal' may have resulted in anxiety, headaches and

other health concerns, which could have impacted the prevalence of PHA in university lecturers.

The higher education setting provides an occupational environment that requires teachers to perform and respond to different roles; these include, but are not limited to, teaching, mentoring, hosting internships, conducting research and organising seminars. All of these require a high level of personal, professional and scientific development (Souto *et al.*, 2018), adding a burden on the university lecturer. University lecturers experience pressure from heavy workloads and high levels of stress (Donnelly, 2014). There is a positive correlation between TTH and stress (Ashina, Bendtsen and Ashina, 2005).

The quality of higher education depends on various factors, such as student-staff ratio, funding from the government and private institutions, preparedness of learners, resource availability, student support services and the number of lecturers among others, all of which has been a challenge for South Africa (Tewari and Ilesanmi, 2020). The performance of students in South African universities is 74%, compared to 80% which is the desired national benchmark (RSA, 2013). Between 2009 and 2017, the average graduation rate was 16.43%, which was far below the target of 25% (Tewari and Ilesanmi, 2020). This places pressure on South African lecturers to make sure that students pass and graduate.

Ergonomically, WFH has many negative aspects which may have led to headaches in lecturers during the COVID-19 pandemic. This includes but is not limited to improper chair height, increased screen exposure, poor laptop/desk positioning and no footrest/armrest (Yeow, Ng and Lim, 2021). Compared to universities in developed countries, South African universities are still struggling with access to the internet (Van Deursen and Van Dijk, 2019). This may have been one of the major struggles that South African lecturers had to deal with during the transition to online teaching and learning, which may have led to or aggravated headaches.

### **1.2 RESEARCH PROBLEM**

Headaches are considered treatable, but many headaches are under-recognised, underdiagnosed and often under-treated (Takeshima and Kikui, 2013). Despite the high prevalence and public health impact, headaches have only been partially documented, with most research focusing on migraine headaches (Dowson, 2015). There are limited studies done on headaches in university lecturers, yet they play such an integral part in the lives of students. University lecturers are required to perform various tasks, which puts them under a lot of pressure due to heavy workload and stress. The COVID-19 pandemic increased this pressure as lecturers were forced to adapt to online teaching. This transition brought about poor ergonomic factors, amongst other things, such as inadequate workspaces, increased time spent on laptops/computers, poor seating and desk height which could contribute to headaches. These headaches may have led to absenteeism, decreased productivity and a reduced quality of life. A high prevalence and disease burden of migraine headaches has been associated with productivity losses and employer costs (Shimizu *et al.*, 2021). According to a previous study, among 2458 employees, 17% had migraine and had significantly missed more work days compared to non-headache sufferers or those with tension headaches (Shimizu *et al.*, 2021).

The PHA prevalence in developing countries has been insufficiently investigated (Li *et al.*, 2020). From the literature presented, it is evident that there is a great need to conduct further studies on headaches, focusing on the prevalence, selected risk factors and burden of headaches, due to contradictory findings in the literature, and to assist health care providers (Kurt and Kaplan, 2008).

## **1.3 AIM OF THE STUDY**

The aim of this study was to determine the prevalence, selected risk factors and burden of primary headaches in lecturers at an university of technology (UoT) in KwaZulu-Natal during the COVID-19 pandemic.

## **1.4 OBJECTIVES OF THE STUDY**

- 1) To determine the prevalence and types of primary headaches suffered by lecturers at an university of technology.
- 2) To determine the socio-demographics, medical history, work-related and COVID-19 related risk factors for the primary headaches.
- 3) To determine the personal impact of the primary headaches in terms of work, family and social life.

## **1.5 SIGNIFICANCE OF THE STUDY**

Headaches have been under-studied in developing countries and there is a paucity of literature documenting headaches in university lecturers. The COVID-19 pandemic has forced lecturers to adapt to online teaching which may have increased their susceptibility to headaches.

This research utilised the HARDSHIP questionnaire, which has not been utilised in a South African context, to determine the prevalence, risk factors and burden of primary headaches.

By conducting this research, information has been produced to assist chiropractors and other healthcare providers, who treat lecturers, understand the extent of the selected risk factors which lead to headaches in this population. Recommendations will be made to the institution to encourage employee assistance programmes to help staff manage and improve their quality of life (Al-Hashel *et al.*, 2021).

### **1.6 FLOW OF THE DISSERTATION**

The following chapters are structured as follows:

Chapter Two provides a detailed review of the current literature on primary headaches. This literature focuses on the prevalence, pathogenesis and selected risk factors associated with these sub-types. This chapter also describes the impact of headaches in terms of work, family and social life.

Chapter Three details the methodology of the study.

Chapter Four presents a detailed report on the results obtained from the analysis of the data collected.

Chapter Five provides an analysis and discussion of the results, which is relative to the current literature.

Chapter Six concludes the study and provides recommendations for future studies. Limitations are also identified and noted.

## **1.7 CONCLUSION**

Headaches are a common disabling condition that affect many people worldwide. Most individuals who present with a headache to a primary care provider for an evaluation have a primary headache disorder (Rizzoli and Mullally, 2017). Primary headaches are more common than secondary headaches but are not life-threatening. However, these headaches place a significant burden on the lives of individuals. For the purposes of this study, primary headache disorders have been discussed. Headaches have been understudied in developing countries and there is a paucity of literature documenting headaches in lecturers.

The COVID-19 pandemic has forced lecturers to adapt to online teaching which may have resulted in headaches. This study aimed to determine the prevalence, selected risk factors and burden of primary headaches in lecturers at a university of technology (UoT) in KwaZulu-Natal during the COVID-19 pandemic. This research may be beneficial to chiropractors, other healthcare providers and lecturers.

The next chapter is the literature review which will present the various objectives in detail.

## **CHAPTER TWO LITERATURE REVIEW**

### **2.1 INTRODUCTION**

This chapter provides a detailed account on the current and relatable literature on the epidemiology of headaches in lecturers. Information on the prevalence, risk factors and burden of headaches is highlighted in this chapter. Numerous search engines, such as Google Scholar, PubMed, the Durban University of Technology Library and the Institutional Repository, were used to gather information.

## **2.2 CLASSIFICATION OF HEADACHES**

Headaches can be classified broadly as primary and secondary headaches (Aurora, 2004; Ahmed, 2012). According to the International Classification of Headache Disorders (ICHD-III), headaches are classified as the following:

- Primary headaches, which include tension, migraine and cluster.
- Secondary headaches, which include potentially life-threatening aetiologies, traumatic brain injury and vascular disorders.
- Cranial neuropathies, such as trigeminal neuralgia (IHS, 2018).

Primary headaches, which constitute most headaches (Ahmed, 2012), are composed of multiple entities that cause chronic and episodic head pain, in the absence of any underlying pathology or traumatic injury (Mier and Dhadwal, 2018). These headaches have no clear etiology on examination of patients, nor on structural imaging (Bahra, 2020) and are not life-threatening. According to the Mayo Clinic (2015), nerves, blood vessels and muscles of the head and neck all play a role in the development of primary headaches. Primary headaches are divided into tension-type headaches (TTH), migraine headaches (MTH) and trigeminal autonomic cephalgias (TACs) (Benoliel and Eliav, 2013).

Secondary headaches occur secondarily, caused by another disease or clinical condition (IHS, 2004). These headaches are not as common as primary headaches; however, they may be life-threatening (Ahmed, 2012).

For the purposes of this study, the focus will be on primary headaches.

## 2.3 TYPES OF PRIMARY HEADACHES

The TTH, MTH and CTH are discussed as follows.

#### 2.3.1 Tension-Type Headache

Of the primary headaches, TTHs are the most common. It is said to be the most prevalent neurological disorder (Ashina *et al.*, 2021). According to the IHS (2018), the lifetime prevalence for TTH, in various studies ranges, is between 30% and 78% in the general population. These headaches were previously referred to as 'muscle contraction headaches' due to the role of stress and emotional and/or mental conflict which triggers pain, and then causes contractions of the muscles in the face, scalp, jaw and neck (Hassan and Asaad, 2020).

These headaches are further divided into infrequent episodic TTH (less than one headache a month); frequent episodic TTH (this includes two to fourteen headaches per month); and chronic TTH (which includes 15 headache days a month or more) (Derry *et al.*, 2015). Infrequent episodic TTH occurs in almost everyone and generally does not require medical attention, whereas frequent episodic TTH may be associated with disability, and one may need treatment, such as expensive drugs. Chronic TTH is the most serious and affects quality of life (IHS, 2018).

The etiology of TTHs is not well understood. Previously, it was considered to be primarily psychogenic but numerous studies since ICHD-1 suggest a neurobiological basis (IHS, 2018). Nutritional, muscular, environmental and genetic factors may play a role in the development of TTH (Shah and Hameed, 2021). In a randomised control trial, 100 patients with chronic TTH were compared with a control group of 100 people. The results showed that almost 70% of patients with chronic TTH had vitamin D deficiency, and a quarter people in the control had a vitamin D deficiency (Prakash et al., 2017). Stress and posture are two significant factors in TTHs (Kratěnová et al., 2007), but the pathophysiology is poorly understood. It is said that poor posture, like neck flexion for a prolonged period while watching a computer screen, leads to an increased amount of stress placed on the atlantoaxial joint. The shoulders will then try to compensate, leading to muscular imbalances and tightness in some areas causing TTHs (Shah and Hameed, 2021), which lecturers will be prone to due to the prolonged screen time. A population-based study of 11 199 twin pairs concluded that genetic factors were said to play a role in TTH and frequent episodic TTH, whereas infrequent episodic TTH is caused mainly by environmental factors (Russell, Saltyte-Benth, Levi, 2006).

According to Rudra, Chatterjee and Ray (2015), TTH is characterised by a pressing or tightening sensation which has a mild to moderate intensity. TTH usually occurs bilaterally and may last between thirty minutes and seven days (Blumenfeld, Schim and Brower 2010).

The risk factors for tension type headaches include emotional stress, anxiety, depression, panic disorders, and changes in sleep patterns, skipping meals and exposure to challenging environments, such as noise, heat and poor lighting (Black *et al.*, 2004). Factors that aggravate TTH include stress, anxiety, fatigue, lack of sleep, temperature and certain foods, whereas factors that relieve TTH include massage, rest, sleep, medication and good posture (Haque *et al.*, 2012).

A TTH can be differentiated from a MTH by the absence of photophobia, phonophobia, nausea and other neurological symptoms (Martami *et al.*, 2018).

#### 2.3.2 Migraine-Type Headache

A MTH is the second most common headache (Ahmed, 2012). The word "migraine" is from the term "hemicranias", that means "half of the head" and many people experience pain in half of the head, however individuals may feel pain on both sides (Khan *et al.*, 2021). Each year across the world, more than one billion individuals are affected by MTH. These headaches are one of the most common neurologic disorders, with a high morbidity and prevalence (Amiri *et al.*, 2021).

According to Ruschel and Jesus (2022), migraines are separated into subtypes: migraine with aura, migraine without aura and chronic migraine. The IHS (2018) describes migraine with aura as being characterised primarily by the transient focal neurological symptoms, which usually precede, or sometimes accompany the headache. MMigraine without aura is classified as a clinical syndrome, which is characterised by a headache with specific features and symptoms. Many patients complain more about the discomfort associated with the aura, rather than the actual MTH (Manzoni and Torelli, 2008). A chronic migraine is a headache occurring 15 or more days a month, for more than three months. Categorising MTHs are different phases, including a premonitory phase, transient neurological symptoms, intense headache attack and a postdrome phase (Andreou and Edvinsson, 2019).

For the etiology of MTHs, it is important to note that there is a strong genetic component associated with these headaches (Ruschel and Jesus, 2022). This is complex and there is uncertainty about which genes and loci are implicated in the pathogenesis (de Vries *et al.*, 2016). According to Blumenfeld, Schim and Brower (2010), MTHs occur as a result of the activation of the trigeminal nerve fibres and inflammation of the meningeal blood vessels.

A MTH is characterised by a pulsating/throbbing sensation and is usually unilateral (Blanda and Wright, 2009 and Chawla, 2009). MTH causes the patient to rest as it is worsened by physical activity (Chawla, 2009). These headaches may be associated with syndromes, such as cyclic vomiting, abdominal migraine, somnambulism, benign paroxysmal torticollis, benign paroxysmal vertigo and confusional migraine, all of which have a different duration and presentation (Straube and Andreou, 2019). There are many risk factors for MTHs as shown in Table 2.1 (Bigal and Lipton, 2006; Lipton and Bigal, 2005; Amiri *et al.*, 2021).

#### Table 2.1: Risk factors for migraine

		Estrogen dysregulation	Cortisol dysregulation
	imbalances		Advancing age
		Demographic factors	Female sex
		Matakalia fastara	Obesity
Piological factors			Dyslipidemia
Biological factors			Diabetes
			Hypertension
		Genetic and epigenetic	MTDH gene
			MEF2D gene
			PRDM16 gene
			Phobia
Psychological factors		Anxiety	Panic
			Stress
Miscellaneous		Lower level of education	Lower socioeconomic status

#### Migraine risk factors

<sup>\*</sup>These conditions need to be furtherly assessed by future studies.

#### 2.3.3 Trigeminal Autonomic Cephalalgias

Trigeminal autonomic cephalgias (TACs) are categorised as cluster-type headaches (CTH), paroxysmal hemicranias (PH) and short-lasting unilateral neuralgiform headache attacks with conjunctival injection and tearing (SUNCT) (Lambru and Matharu, 2013; IHS, 2018). For the purposes of this study, only cluster-type headaches will be discussed as CTH is the most common headache among TACs (Ahmed, 2012).

A CTH is the most prevalent headache among TACs (Ahmed, 2012), but these headaches are relatively rare and are considered as one of the most severe types of headaches (Kandel and Mandiga, 2022). These headaches occur in 0.1% of the general population (Kandel and Mandiga, 2022). The onset of CTH is usually between 20 to 40 years (IHS, 2004). CTH is divided into two forms: episodic and chronic. Episodic CTH indicates that

there is a remission period, whereas chronic CTH sufferers experience daily or near-daily headaches for more than one year without remission (Rudra, Chatterjee and Ray, 2015)

The exact aetiology of CTH is not known but studies have shown a genetic component, and multiple loci were associated (Harder *et al.*, 2021; O'Connor *et al.*, 2021). Although the mode of inheritance is not clear, it is known that first-degree family members are 18 times more likely to have CTHs (Kandel and Mandiga, 2022). An association between the trigeminovascular system, hypothalamus and the parasympathetic nerve fibers involved in the trigeminal autonomic reflex is known (Kandel and Mandiga, 2022). There is also a link between a pain attack and vasodilation.

These headaches are unilateral and short-lasting, with at least one autonomic symptom, such as nasal congestion, lacrimation, aural fullness or conjunctival injection (Drummond, 2006). Risk factors include family history, alcohol consumption, male gender, age greater than 30 and prior brain surgery or injury (Waung *et al.*, 2020).

### 2.4 PREVALENCE OF PRIMARY HEADACHES

Headache disorders are among the most prevalent conditions worldwide according to the Global Burden of Disease (GBD) (Stovner *et al*, 2022). According to Rizzoli and Mullally (2017), the lifelong prevalence of headache is 96% and there is a female dominance. Primary headache disorders have shown to have a lifetime prevalence of 91.3% (Steiner *et al.*, 2014). Approximately one half of the adult population worldwide is affected, at any time, by a headache disorder (Hainer and Mathesor, 2013). Among the adult population, prevalence studies revealed that TTHs affect 38%, whereas MTHs affect 12% (Robbins, 2021). A study on the general population, age group 18-65, in China, has shown that a one-year prevalence was 23.8% (Yu *et al.*, 2012).

The most common type of primary headaches are TTHs (Blanda and Wright, 2009; IHS, 2018), affecting 78% of the general population (Jensen, 2003). A Denmark populationbased study has shown the lifetime prevalence of TTH to be 78%, with a majority of episodic infrequent TTH (Lyngberg *et al.*, 2005). TTHs are more common in females than males (Derry *et al.*, 2015 and Chowdhury, 2012). According to Bayraktutan *et al.* (2014), studies and clinical experience have shown that TTH is more common in females (1: 6; male: female ratio) and occurs more commonly in the third and fourth decades. In both males and females, the prevalence of TTHs peaks between the ages of 30 and 39 years and decreases with an increasing age (Lyngberg *et al.*, 2005)

The second most common headaches are MTHs (Ahmed, 2012). In 2019, the global prevalence of MTH was estimated to be 1.1 billion (Safiri *et al.*, 2022). Over the last three

decades, the global prevalence of MTH has increased substantially (Safiri *et al.*, 2022). A study in Turkey revealed about 20% of sufferers have an attack once or less per month; 38% had one to four attacks per month; and 42% experienced migraines more than four times per month (Özdemir *et al.* 2014). MTHs occur mainly between 25 and 55 years (Robbins and Lipton, 2010). Females are more likely to suffer from MTHs compared to males and the prevalence in the general population ranged from 2.6% to 32%, according to a Denmark study. However, the prevalence of these headaches varies as it is dependent on the technique used to diagnose them in the different studies (Ali, Ahmed and Omer, 2022). In 2019, the global prevalence rate in females and males were 17 902.5 and 10 337.6 respectively (Safiri *et al.*, 2022).

Among TACs, CTH is the most prevalent headache (Ahmed, 2012) and they occur more frequently as a continuance of a disorder that began during young adulthood (Fowler, Capobianco and Dodick, 2004). There is a male predominance (Bahra, May and Goadsby, 2002; Wei, Yuan Ong and Goadsby, 2018), and the mean onset is in the third decade. Since the prevalence is lower compared to migraines, it is difficult to assess the prevalence in the community (Wei, Yuan Ong and Goadsby, 2018).

Information on headache prevalence is sparse in Africa (Oshinaike *et al.*, 2014). In developing countries, the prevalence of headaches is not investigated sufficiently (Li *et al.*, 2020) which denotes a need for studies in developing countries, like South Africa. Due to the insufficient literature, management of headaches may be affected. A study conducted in rural South Tanzania revealed that the one-year prevalence of headache was 23.1% (Dent *et al.*, 2004). The one-year prevalence of migraine in Ethiopia was 3%, with the peak age in the fourth decade (Tekle Haimanot *et al.*, 1995). In Nigeria, the prevalence ratio of migraine was said to be 5.3% (Osuntokun *et al.*, 1992). According to Charleston (2021), in the United States, race-based disparities exist in headache medicine, but more research is needed. Due to the diversity of South Africa, this research will be beneficial as all race groups are included so it will build on the existing body of knowledge.

There are socioeconomic disadvantages which are highly prevalent in those with headaches (Burch, Rizzoli and Loder, 2020). According to Mateen (2008), primary headaches are undertreated in many developing countries. The inconsistent literature on the prevalence of headaches is due to type of data collection, such as telephonic or face-to-face interviews (Kurt and Kaplan, 2008; Stovner *et al.*, 2007). Factors that may also affect the consistency of information include age, gender, cultural background, ethnicity, or a combination of these factors (Rasmussen, 1999 and Stovner *et al.*, 2007). Studies on the prevalence in different occupational groups is important to note due to the increased

prevalence and consequent decline in the ability to work (Li *et a*l., 2020). The literature on the prevalence in lecturers remain sparse.

## 2.5 RISK FACTORS FOR HEADACHES

There are several factors which influence our health — these are called determinants of health. This includes biology, genes, behaviour, diet and lifestyle factors, like smoking and exercise. There are also social determinants, which include our social and physical environment. The conditions that we live, work, play and learn in are our social determinants (Jswords, 2018).

### 2.5.1 Social History

According to Molarius *et al.* (2008), lifestyle factors which increase the prevalence of headaches include physical inactivity, occasional/current smokers, occasional heavy consumption of alcohol, missing breakfast and underweight or obese individuals.

Large population studies have shown strong evidence of an association between CTHs and cigarette smoking (Raucci *et al.*, 2020). According to Dueland (2015), 75% of MTH patients indicated alcohol as the most common cause for their attacks. Both excessive sleep and lack of sleep may trigger headaches (Raucci *et al.*, 2020).

### 2.5.2 Medical History

Headaches are the most common symptom after a head injury (Lucas *et al.*, 2014). According to the IHS (2013), headache attributed to a head injury (HAIH) is a headache that has no clinical characteristics and starts within seven days of injury. Individuals with a mild head injury have a higher headache severity and prevalence than those with a severe head injury (Theeler *et al.*, 2013; Solomon, 2009), but further investigation is needed.

The results for the studies examining headaches and reduced vision indicate a link between headaches and refractive errors (Gil-Gouveia and Martins, 2002; Gunes *et al.*, 2016). Headaches may be provoked by uncorrected vision and the need for glasses (Thorud, Aurjord and Falkenberg, 2021).

According to Lee *et al.* (2021), one of the most common causes of health-related productivity loss is headaches. The association of MTH with an increased risk of long-term cardiovascular and cerebrovascular events is mainly driven by an increased risk of stroke and myocardial infarction (MI) (Mahmoud *et al.*, 2018). In the early twentieth century, a correlation between headache and high blood pressure was first considered (Wang and Wang, 2021). A relationship between headache and an increased blood pressure were

repeatedly reported in literature (Adney *et al.,* 2021). Headache may be considered the most common symptom for arterial hypertension (Arnold, 2018).

### 2.5.3 Psychological

There is a significant association between headaches and anxiety/depression (Magnavita, 2015; Chu, 2018). Most of the research done on headaches has been on the role of biological factors in headache disability, but reliance on a purely biomedical model of headache does not account for all aspects involved in headache causation (Nicholson et al., 2007). Psychosocial risk factors have been shown to greatly impact headaches occurrence and frequency (Malmberg-Ceder et al., 2019). Stress is more common in headache populations, with depression and anxiety being associated with both migraine and other headaches. The relationship between depression/anxiety and headache is thought to be bidirectional (Malmberg-Ceder et al., 2019). There is evidence showing that neural circuits responsible for cognitive-affective phenomena is highly interconnected with the circuitry responsible for headache pain and, therefore, psychological factors can influence headache pain via various central nervous system pathways (Nicholson et al., 2007). Psychological risk factors, such as headache management, locus of control and selfefficacy, and negative affect/emotional states can alter the likelihood of a headache attack being triggered, the severity of pain, the impact on functioning treatment and prognosis. In addition, tension and stress is the leading trigger in TTH (Ashina, Bendtsen and Ashina, 2005).

Smith (2007) reported that there are high levels of stress with higher education systems. The teaching profession has always involved stress, due to excessive workloads, interpersonal communication problems, insufficient training and job insecurity (Ozamiz-Etxebarria, 2021). Higher education teachers are expected to perform and respond to different roles; these include, but are not limited to, teaching, mentoring, hosting internships, conducting research and organising seminars. All of these require a high level of personal, professional, and scientific development (Souto *et al.*, 2018). At universities, administrative duties have increased, together with increased demands placed on teaching and research responsibilities.

### 2.5.4 Work-Related

There are many work-related factors which have been attributed to headaches (Magnavita, 2022). The results from an Italian survey reported that headaches were one of the most common symptoms that office workers experienced. This was due to heat, drafts, lighting, odours and poor air quality in the working environment (Magnavita, 2014). Leadership styles

(Magnavita, 2022) and injustice in the workplace (Min *et al.*, 2014) may also play a role in headaches.

When South Africa experienced its first lockdown for COVID-19, lecturers were required to abruptly change their work environment and patterns and work from home. Many negative aspects of WFH have been described. Working remotely has led to an increased duration of screen exposure which may contribute to fatigue, headaches, and tiredness (Majumdar, Biswas and Sahu, 2020). WFH ergonomic problems that were raised included too much screen time, improper chair height, noise and distortion, poor lighting, poor laptop/desk positioning, no footrest/armrest and an increased key object distance (Yeow, Ng and Lim, 2021). Living alone and WFH without any face-face interactions may contribute to social isolation and depression as there is no opportunity to socialise with colleagues (Xiao *et al.*, 2021).

Excessive screen time is possibly a leading cause of headaches (Abou Hashish *et al.*, 2022) and these headaches are likely to be unilateral and occur towards the middle or end of the day (Mashige, 2014). According to Ranasinghe *et al.* (2016), the eyes and extraocular muscles endure constant shifting and accommodating for a prolonged period causing stress on the muscles and fatigue of the eyes which lead to headaches. The brightness of the screen light leads to migraine attack, and the migraine cascade threshold may reduce due to an increased screen exposure (Kowacs *et al.*, 2004). Working in the same position for prolonged hours, including computer work, is a risk factor for headaches (Houle *et al.*, 2021). Workers with headache disorders reported the use of computer monitors and occupational stress to be the most common risks (Vicente *et al.*, 2021).

According to Demir and Sumer (2019), headache duration and frequency of MTH is increased due to the use of smartphones. This is attributed to poor sleep quality and daytime sleepiness. A community-based cross-sectional study conducted in India concluded that headaches are associated with usage of a mobile phones (MP) — the prevalence of MP usage was reported to be 70% (Stalin *et al.*, 2016). In addition, headaches in MP users were increased in those who had longer calls and higher daily call frequency (Wang *et al.*, 2017).

The chronical postural strain from working in uncomfortable positions are a mechanical cause of TTH (Smith *et al.*, 2009). The severity of TTH increases because of the postural strain placing stress on the precranial muscles to their maximum tension (Monzani, Zurriaga and Espi Lopez, 2018).

#### 2.5.5 COVID-19

According to Harapan *et al.* (2020), an outbreak of the coronavirus disease 2019 (COVID-19) occurred in 2019 in Wuhan City, China, caused by a novel SARS-CoV-2. Evidence shows a zoonotic origin of COVID-19 (Lau *et al.*, 2020) and Wu *et al* (2021), as a bat being the natural host, which may be questionable. Within the first 50 days, it was noted that there were over 70 000 infections and more than 1 800 individuals had died (Shereen *et al.*, 2020). According to Broadbent *et al.* (2021), the first case was reported on 5 March 2020 in South Africa. A National State of Disaster was declared on 15 March and, from March 27, a full lockdown of the country was in place. This included closing most economic activity and prohibiting movement for non-essential purposes (Broadbent, Combrink and Smart, 2021). During this time, schools and universities were closed, and teaching and learning had to be done online.

According to the ICHD, headaches are accepted as a symptom of a systemic viral infection (Table 2.2) (IHS, 2013), although the exact mechanism is not known. Headaches in infectious diseases coexist with fever but may also be present without fever. It is indicated that systemic infections do not have this effect simply through fever and exogenous/endogenous pyrogens, but the mechanisms include direct effects of microorganisms and activation of immunoinflammatory mediators (IHS, 2013).

Indirect mechanisms that cause headaches in SARS-CoV-2, but are not disease specific, include dehydration, metabolic disturbances, systemic inflammation and hypoxia (Bobker and Robbins, 2020). Although more studies are yet to be done discussing the link between headaches and COVID-19, proposed links are discussed as follows.

The central nervous system (CNS) may directly be invaded by SARS-CoV-2 via the olfactory bulb (Bobker and Robbins, 2020). Angiotensin-converting enzyme 2 (ACE2) are the receptors for cell entry in SARS-CoV-2 and these receptors are mainly expressed in neurons, but also in the glia in the CNS (Togha *et al.*, 2022), the potential for direct viral infection causing headaches (Bobker and Robbins, 2020). SARS-CoV-2 may also cause organ damage due to the downregulation of ACE2 expression on the cells (Togha *et al.*, 2022).

Headaches may be due to vascular attack and trigeminovascular activation from high expression of ACE2 in the capillary endothelium (Abboud *et al.*, 2020). Trigeminovascular system activation leads to the release of neurotransmitters that are pain-generating, which may cause headaches presenting similarly to MTHs (Noseda and Burstein, 2013; Planchuelo-Gomez *et al.*, 2020).

According to Bobker and Robbins (2020), headaches in SARS-CoV-2 may be linked to cytokine release syndrome (CRS), a supraphysiological response following immunotherapy and associated with neurotoxicity quite often. Neurotoxic symptoms that are seen with chimeric antigen receptor (CAR) T cells include headaches which were present in 42% of people in immunotherapy (Rubin *et al.*, 2019). The role of cytokines such as tumour necrosis factor (TNF- $\alpha$ ) headache pathophysiology have been described (Wagner and Myers, 1996), including an increased level during migraine attacks (Perini *et al.*, 2005).

In South Africa, the use of face masks was recommended on 10 April 2020 (Rab, Javaid and Haleem, 2020). Stress and wearing masks were found to be the main triggers of headache (Uygun *et al.*, 2020). The pathogenesis may be due to mechanical compression, hypoxia, rebreathing of carbon dioxide (Co2), stress or an aggravation of a pre-existing headache (Gurnani and Kaur, 2021). Studies have focused on masks and headaches in healthcare workers (Rosner, 2020; Ramirez-Moreno *et al.*, 2020), but little or no research has been published on the effect of mask usage in lecturers. Once lecturers had returned to university, they had to adapt to lecturing with masks on, something that was new for everyone. Whether or not mask usage aggravated headaches in lecturers is worthy of attention.

An adverse event most frequently complained about after the COVID-19 vaccination included headaches (Castaldo *et al., 2022*). In both clinical trials and real-world data, headaches have been reported by almost half of the recipients of the vaccine (Zhu *et al.,* 2020).

Table 2.2: ICHD3: headache	attributed to	systemic viral	infection
----------------------------	---------------	----------------	-----------

Description	Headache caused by and occurring in association with other symptoms and/or clinical signs of a systemic viral infection, in the absence of meningitis or encephalitis.			
Diagnostic criteria	A.H d c	leadache of any luration fulfilling criterion C		
	B. B fo	Both of the ollowing:	1. 2.	Systemic viral infection has been diagnosed. No evidence of meningitic or encephalitic involvement.
	C. E c d a fo	Evidence of causation lemonstrated by at least two of the ollowing:	1. 2. 3. 4.	<ul> <li>Headache has developed in temporal relation to the onset of the systemic viral infection.</li> <li>Headache has significantly worsened in parallel with worsening of the systemic viral infection.</li> <li>Headache has significantly improved or resolved in parallel with the improvement in or resolution of the systemic viral infection.</li> <li>Headache has either or both of the following characteristics: <ul> <li>(a) Diffuse pain.</li> <li>(b) Moderate or severe intensity.</li> </ul> </li> </ul>
	D. N a a d	Not better accounted for by another ICHD-3 diagnosis.		

From a clinical point of view, and from the literature presented, lecturers are exposed to risk factors that may pre-dispose them to developing headaches. Therefore, since there are limited studies done in South Africa documenting risk factors in lecturers, this study will be of great benefit, not only to chiropractors and other health related professionals, but for lecturers as awareness will be raised.

### 2.6 CLINICAL CHARACTERISTICS OF PRIMARY HEADACHES

Each headache has unique clinical characteristics, and by knowing the characteristics it makes it easier to identify the type of headache that one is experiencing, thus directing treatment.

### 2.6.1 Tension-Type Headache

The TTHs are characterised by a diffuse, mild to moderate pain that is described as feeling a tight band around the head (Mayo Clinic, 2021). The pain is described as dull, pressure-like or constricting, or some say that there is a sense of fullness in the head (Chowdhury, 2012). The pain is tightening in quality and bilateral in location. The duration of these headache episodes

varies between 30 minutes and seven days, with the average duration being 4–6 hours (Shah and Hameed, 2021). These headaches usually start during the day and increases slowly, which may be aggravated in the evening in some people (Chowdhury, 2012). A TTH does not worsen with physical activity and is not usually associated with nausea/vomiting but photophobia or phonophobia may be present, but both should not be present in a TTH (Chowdhury, 2012). Patients with TTHs may also present with neck and/or shoulder tightness and sleep disturbances (Shah and Hameed, 2021).

#### 2.6.2 Migraine Type Headache

A MTH is characterised by severe paroxysmal unilateral pulsatile headaches (Li *et al*, 2020), which usually begins with warning signs (prodromes) and an aura. Migraine has two major types: with aura and without aura which is discussed as follows. Individuals may also experience a prodromal and/or postdromal phase, which includes symptoms such as fatigue, depression, craving certain food, yawning that is repetitive, hyperactivity or hypoactivity and neck stiffness/pain (IHS, 2018). Once the MTH develops, there is throbbing which intensifies with an increase in intracranial pressure. This is then associated with nausea, vomiting and abnormal sensitivity to noise, light and smell, which can also be accompanied with muscle tenderness and allodynia (Burstein *et al.*, 2015). The duration can be hours to days and the pain may be extremely severe (Mayo Clinic, 2021).

According to the IHS (2018), the following characteristics of the sub-types of migraines are described:

#### Migraine without aura

Recurrent headaches that have attacks which last 4–72 hours and these headaches have a unilateral location, moderate to severe intensity and pulsating quality, aggravated by physical activity. These headaches are associated with nausea and/or phonophobia and photophobia.

#### Migraine with aura

The attacks are recurrent and lasts for minutes. There are unilateral, fully reversible, sensory, visual or other central nervous system symptoms which gradually develop. This is then followed by headache and migraine associated symptoms.

#### 2.6.3 Cluster Headache

Cluster-type headaches present as a harrowing, unilateral headache (Sargeant and Blanda, 2009; Lambru and Matharu, 2013). These headaches are a form of sporadic, short-lived and agonising head pain associated with automatic dysfunction (Rudra, Chatterjee and Ray, 2015). Cluster-type headaches usually involve the first division of the trigeminal

nerve, the temple and over the peri and retro-orbital regions (Bahra, May and Goadsby, 2002). These headaches may be accompanied by symptoms such as ipsilateral conjunctival injection, sweating of the facial or forehead area, lacrimation, restlessness or agitation, rhinorrhoea, nasal congestion, miosis, ptosis and/or eyelid oedema (IHS, 2018).

## **2.7 DIAGNOSIS**

The HARDSHIP questionnaire utilised in this study has proven cross-cultural validity and incorporates demographic enquiry, diagnostic based questions and burden related questions. This instruments' modular design makes it flexible to adaptation and cultural sensitivities (Steiner *et al.*, 2014).

Strengths:

- The evolution of this instrument has come from a broad base of expert opinions.
- It has undergone testing in many settings and cultures.
- The successful use in many studies have built a collection of studies that had similar methodology in various regions.

Limitations:

• In all cultures and languages, the diagnostic questions are insensitive to TTH.

The diagnostic algorithm (Appendix 8) was utilised to diagnose headaches according to the diagnostic questions in the HARDSHIP questionnaire.

### 2.8 MANAGEMENT OF PRIMARY HEADACHES

Most headache disorders do not require input from a specialist as they can be treated effectively with a correct diagnosis, with no further investigations (Ahmed, 2012). A diagnosis, empathy and effective treatment may be very rewarding for the clinician as it can make a significant difference to the patient (Sinclair *et al.*, 2015). Educating the patient on lifestyle factors is a vital part in any headache management plan (Raucci *et al.*, 2020).

Physical therapy should always be considered in patients with TTHs. The most used nonpharmacological treatment is physical therapy, including active treatment strategies, such as relaxation, posture improvement, exercise regimen, in addition to ultrasound and electrical stimulation (Jensen and Roth, 2005). Cryotherapy, heat, massage, transcutaneous electrical nerve stimulation and changes to ergonomics are effective for acute TTH, but not episodic TTH (Fumal and Schoenen, 2008). Acupuncture, homeopathy, herbal medication, Indian head massage and cognitive behaviour therapy may help MTHs (Ahmed, 2012). If those affected with MTH are managed with acute and preventative strategies, then the disability and loss of function caused by this disease will be reduced (Jenkins, 2020).

Manual therapy is said to be more effective compared to care from a general practitioner in both short- and long-term reduction of symptoms for chronic TTH (Castien *et al*, 2011). Spinal manipulation is not effective in treating episodic TTH (Bove and Nilsson, 1998). However, manual therapy, such as spinal manipulation, myofascial manipulation or craniosacral techniques, is effective in treating chronic MTH (Espí-López, Bermell-Salvador and Cortés-Amador, 2017).

Psychological treatments used to treat headaches include relaxation training, EMG biofeedback and cognitive behavioural therapy (Chowdhury, 2012). Relaxation training and biofeedback have widely been accepted for treating headaches (Lee *et al.*, 2019), and have been used for more than four decades (Penzien, Rains and Andrasik, 2002). Muscle tension and autonomic arousal, which may result from headaches, are consciously reduced during relaxation training (Chowdhury, 2012). EMG biofeedback helps the patient recognise and control tension in the muscle, which is done by providing continuous feedback about activity in the muscle (Holroyd *et al.*, 2005). During biofeedback, the patient is helped to gain control over pericranial muscle tension (Chowdhury, 2012). Cognitive behavioural training involves learning to identify thoughts and beliefs which cause stress and aggravate headaches (Chowdhury, 2012). Both cognitive behaviour therapy and biofeedback training have shown to reduce stress (Crystal and Grosberg, 2009).

Medications may be helpful to treat headaches, but it is important to note that prolonged usage and overuse of medication to treat headaches may lead to another headache called medication-overuse headache. The majority of patients should have both pharmacological and non-pharmacological interventions for the headache treatment (Shah and Pandey, 2017). Many patients with TTH can take simple analgesia (Sinclair *et al.*, 2015). Ahmed (2012) indicates the use of over-the-counter analgesics, such as paracetamol and ibuprofen, for infrequent episodic TTH. Those with CTTH may use preventative treatments, such as amitriptyline or topiramate (Clarke *et al.*, 2009). For mild to moderate migraines, non-steroidal anti-inflammatory (NSAIDs) and acetaminophen are first line treatment, whereas triptans are used for moderate to severe migraines (Mayans and Walling, 2018). The most recommended treatments for CTHs are oxygen therapy and triptans (Kandel and Mandiga, 2022).
## 2.9 BURDEN OF HEADACHES

According to Jensen and Stovner (2008), headaches cause disability and loss of quality of life, which is known to be on the same level with other chronic disorders. Headaches are a major problem because they can limit daily activities, decrease quality of life and productivity (Rasmussen, 1999; Bigal *et al.*, 2001; Curry and Green, 2007; Stovner, Hagen and Jensen, 2007; Jensen and Stovner, 2008; Kurt and Kaplan, 2008). According to the Global Burden of Diseases (GBD), headaches are major causes of disability worldwide (GBD, 2016). Both personal life and work life may be affected equally in those who are headache sufferers (Bussone *et al.* 2004).

The MTH has been described as a burdensome disease and was said to have an impact on family relationships, work activities and on an individual's economic situation (Leonardi and Raggi, 2019). According to GBD (2019), in 2016, the second largest contributor to the disability-adjusted life-years globally was MTH.

The burden of TTH is greater than MTH with regard to the aspect of absenteeism from work (Oshinaike *et al.*, 2014). There is a paucity of literature in terms of the impact of headache disorders and the loss of productivity at the workplace (Shimizu *et al.*, 2021). Studies have revealed significant effects on partners of migraine sufferers, with 24% missing family days or social activities and 12% who avoid family days and social activities due to proband's migraine. Migraine sufferers have also reported the influence the headache has on good parenting. Some individuals had stated that they have avoided having children because of the headache (Hamelsky, Lipton and Stewart, 2005; Lipton *et al.*, 2003). CTHs have a negative impact on personal life, society and self-perceived health (Petersen *et al.*, 2022).

In South Africa, higher education is a force in national socioeconomic development against a backdrop of skill shortage, oppression, ignorance and perpetual dependence (Maharasoa and Hay, 2001). Compared to universities in developed countries, South African universities are still struggling with access to the internet and computers due to the intensity of factors, which has led to a digital divide (Van Deursen and Van Dijk, 2019), making it difficult for lecturers to teach online. A major problem that affects the quality of graduates in South Africa is a huge number of underprepared students entering higher institutions (Tewari and Ilesanmi, 2020). Since performance is dependent on the number of teachers, their quality, their devotion and duty and their effectiveness (Ahmad, 2016), they are given a major responsibility to produce good quality graduates and, therefore, have many tasks, increased workloads and stress.

Over the past two years (2020–2022), there have been significant changes in the lives of university lecturers due to the COVID-19 pandemic. South African universities have been

forced to transit to online learning due to the COVID-19 pandemic (Mpungose, 2020), which has added an extra burden on lecturers as they have had to adapt to online teaching. This added work stress may negatively affect university lecturers' health, leading to anxiety, headaches and other health concerns, which may influence the prevalence of primary headaches. Lecturers were forced to adapt to the "new normal" within a short space of time and reported having increased workloads, psychosomatic problems and exhaustion (Prado-Gascó *et al.*, 2020). The lockdown and social distancing policies implemented to control the spread of the virus have led to dramatic mental health impacts, bringing about greater social isolation and increased anxiety (Smith and Lim, 2020).

The burden of headaches may be reduced significantly by creating awareness, hospital visits, recognising precipitating factors and treatment (Oshinaike *et al.*, 2014). According to Statistics South Africa (2022), the unemployment rate was reported as 35.3%, the highest since 2008. Treating headaches is costly and due to the unemployment rate, many people will not be able to afford proper treatment, worsening the burden of headaches.

## 2.10 CONCLUSION

To date, there has been no literature published that documents the prevalence, risk factors and burden of headaches in university lecturers in South Africa.

From the literature presented, it is evident that headaches affect many people worldwide. For the purposes of this study, the focus was on primary headaches, which have the following subtypes: TTHs, MTHs and TACs. Cluster-type headaches are the most common of the TACs, as discussed previously.

The prevalence of headaches differs in populations and different countries. It is important to identify the prevalence in lecturers as there are limited studies, yet they play an integral role in the lives of students.

There are many risk factors for primary headaches, but only a few have been discussed. The factors discussed seemed to be the most relevant to the lecturer population. The COVID-19 pandemic changed the lives of lecturers as they were forced to transition to online teaching. Identifying the impact these headaches have in the lives of lecturers will assist in better management programmes and additional support.

Although headaches are extremely common, there is still information lacking, so these headaches remain underestimated, underrecognised and undertreated (WHO, 2016). There is a great need for more research on headaches, especially in developing countries.

The following chapter will outline the research methodology.

# CHAPTER THREE METHODOLOGY

## **3.1 INTRODUCTION**

Research methodology is a way to solve a research problem systematically (Patel and Patel, 2019). This is important as a systematic approach with diligent planning is required for the conduct of research (Garg, 2016). This chapter provides an outline of the research methodology, including a description of the study design, study population, questionnaire development, data collection and data analysis.

## **3.2 STUDY DESIGN**

This study utilised a quantitative approach with a descriptive cross-sectional design. This design was the most suitable for this study as it provided data for describing the status of phenomena at a fixed point in time (Ihudiebube-Splendor and Chikeme, 2020). Quantitative research is a research strategy which emphasises the quantification of the collection and analysis of data (Bryman 2012); hence, a quantitative approach with a descriptive cross-sectional design was appropriate for this study. Descriptive studies explain the concepts and their relationships, which is beneficial as it leads to further investigation for new research (Burns and Grove, 2009).

Surveys include the collection of information from a sample of people through their responses to questions (Check and Schutt, 2011). The benefits of surveys include having a big population and, therefore, greater statistical power, the ability to gather large amounts of information and having the availability of validated models (Jones, Baxter and Khanduja, 2013).

The quantitative paradigm for this study was based on positivism. Through a quantitative approach, this generally focuses on identifying explanatory associations or causal relationships, where findings from large samples are favoured (Park, Konge and Artino, 2020).

## **3.3 STUDY POPULATION AND SETTING**

The study population was lecturers (part-time, full-time and those on fixed term contracts), who were employed by the Durban University of Technology (DUT) during the time of this study. These lecturers were of different genders, age groups and ethnicities. The university has seven campuses across the two cities of Durban and Pietermaritzburg, which include: Steve Biko, Ritson, ML Sultan, Brickfield, City, Riverside and Indumiso campuses. There

are six faculties which comprise of Applied Sciences, Management Sciences, Engineering and The Built Environment, Accounting and Informatics, Arts and Design and Health Sciences, with approximately 33 000 students and a total of 685 lecturing staff.

## 3.4 PARTICIPANT RECRUITMENT

All lecturers lecturing at DUT were recruited for this study. In total, there are 685 lecturers employed at the seven campuses. Once approval from the Institutional Research Ethics Committee and Gatekeeper's permission was received, the focus and pilot studies had been conducted, and the questionnaire had been amended accordingly, an e-mail was sent out to all lecturing staff. The e-mail had a link which included a letter of information, informed consent and the questionnaire and was sent via the DUT server. No individuals were forced to participate in this study and no remuneration was offered.

## 3.5 PERMISSION

Permission to conduct this research was sought from the DUT Institutional Research Ethics Committee (Appendix 2) and the gatekeeper's permission from the DUT Director of Research (Appendix 1)

## 3.6 SAMPLING

According to Brink *et a*l. (2012), a sample is a subsection of a research population which makes up a research sample.

## 3.6.1 Sample Size

There were 685 lecturers employed at DUT when this study was conducted. To reach a suitable sample size for generalisability, a sample size calculation had been performed by a statistician using PASS Software. In an e-mail communication on 14 June 2022, the statistician, Dr Tonya Esterhuizen indicated that, using a 95% confidence interval, a power of 80 and an error margin of 5%, the study required a response rate of 247.

## 3.6.2 Sample Characteristics

## Inclusion criteria

- Registered lecturers (full-time, part-time and contract) employed at DUT.
- Informed consent that was initialled (Appendix 5).

## **Exclusion criteria**

- 1. Tutors and mentors.
- 2. Individuals who participated in the pilot study of the research.

### 3.6.3 Sample Strategy

The whole population was invited to participate in the study, and through self-selection, elected to participate, using consecutive sampling. This sampling strategy was chosen for the study as it included recruiting everyone in the accessible population who met the eligibility criteria over a certain period (Polit and Beck, 2012). This was achieved by providing a link to the questionnaire in an e-mail that was sent to lecturing staff. Following the initial e-mail requesting participation, three follow up e-mails were sent, with a three-week interval between each reminder e-mail. The data collection process took three months to complete.

## **3.7 DATA COLLECTION**

Data can be collected by means of questionnaires, interviews, observations, or scales (Polit and Beck, 2012). For the purposes of this study, a questionnaire was used.

### 3.7.1 Measurement Tool

The measurement tool utilised for this study was the HARDSHIP questionnaire. This questionnaire is a modular instrument with proven cross-cultural validity, which consists of questions related to demographics, diagnostic questions based on ICHD-3 beta criteria and headache burden related questions (Steiner *et al.*, 2014). The questionnaire was available on the internet for use, but permission was granted by Professor Steiner via e-mail (Appendix 3).

## 3.7.2 Administration of the Tool

The questionnaire was administered electronically by a link and was in English since DUT is an English medium university, therefore all lecturers are proficient in this language. The online survey management system QUESTIONPRO® was used. Online survey tools are useful for academic research as it is easy to design a questionnaire and collect data by sending e-mails to the research participants. All data were secured by TRUSTe and NORTRON for security purposes (Nagalakhmi and Trivedi, 2010).

## 3.7.3 The Hardship Questionnaire

The initial version, with adaptations that have been used in eighteen countries, has not yet been used in a South African context. The evolution of this instrument came from a broad base of expert opinions and has undergone testing in many settings and cultures. This instruments' modular design makes it flexible to adaptation and cultural sensitivities (Steiner *et al.*, 2014).

The HARDSHIP questionnaire, after permission was obtained (Appendix 3), was modified to suit the requirements of this study.

The questionnaire consisted of three sections:

- Section A was related to demographic characteristics, social history, medical history, work related characteristics and COVID 19.
- Section B included headache diagnostic related questions.
- Section C consisted of questions that helped establish the impact headaches had on the lecturers' personal life in terms of work, family and social life.

## 3.7.4 Classification of Headaches

The diagnostic algorithm for headaches according to the diagnostic questions in the HARDSHIP questionnaire was used to classify the headaches (Appendix 8). However, this algorithm did not include cluster-type headaches so diagnostic questions for cluster-type headaches were added to the questionnaire.

## **3.8 VALIDATION OF THE QUESTIONNAIRE**

To validate the questionnaire, a focus group and pilot study was done.

## 3.8.1 Focus/Expert Group

A focus group ensured that the participants were able to understand and answer the questionnaire without ambiguity. The purpose of this was to validate the questionnaire. The definition of validity is the extent to which a concept is measured accurately in a quantitative study (Heale and Twycross, 2015) and comprises of four types: content validity, construct validity, criterion validity and face validity. The focus group achieved face validity. Face validity is there to confirm that the content of the test is suitable to the aims (Middleton, 2019). The focus group consisted of five participants and each person received a letter of information (Appendix 4) and informed consent (Appendix 5). Although there were six criteria, five participants were deemed fit as most of the participants met more than one criteria.

## Inclusion criteria:

The focus group consisted of:

- 1. The researcher, supervisor and collaborator.
- 2. A person with experience in survey research.
- 3. A person familiar with statistical analysis of survey data.
- 4. At least one person who was a headache sufferer.

- 5. At least one person who was a lecturer at the Durban University of Technology.
- 6. At least one person who was medically trained to diagnose headaches.

#### **Exclusion criteria**

Participants who were unwilling to sign the letter of information and informed consent.

### 3.8.2 Outcome of the Focus Group

The focus group participants engaged in some robust discussion regarding each question, which resulted in some changes to the questionnaire. A selection of questions were excluded, whilst others were narrowed down to make them more specific and understandable. The number of questions was reduced due to the questionnaire taking longer than the expected time. Instructions for each section was changed to 'Choose an appropriate option' instead of 'Place an x where necessary' since the questionnaire was administered online.

For Section A, the following changes were made:

- Under the sub-section "Demographics":
  - The first question changed to "What is your age?" and "What is your sex?" moved to the second question.
  - "What is your marital status?" was added as the last question.
- Under the sub-section "Social History":
  - Two questions were added regarding exercise: "Do you exercise?" and "If yes, how many hours per week do you exercise for?".
- Under the sub-section "Medical History":
  - "In general, how would you describe your general health?" was added.
  - Under the list of medical conditions, two additional boxes were added sinusitis and hayfever.
- Under the sub-section "Work Related":
  - "You may choose more than one option" was added under the question "Describe the manner in which you have lectured over the last year".
  - "Now that things are slowly getting back to normal, how are you coping?" was added.

#### For Section B the following changes were made:

- Under the sub-section "Screening":
  - "On how many days in the last 3 months did you have a headache?" was reworded to "During the last 30 days, on how many of these days did you have a headache?".

- "Have your headaches worsened since COVID-19?" was changed to "Have your headaches worsened during the pandemic?".
- A question about family was added: "Does anyone in your immediate family suffer from similar headaches?".
- The sub-section "Daily headache questions" was removed and the questions that were under that section was moved to screening.
- The following question was added "Please think about your headaches. Do you think they are all of one type or are they of more than one type?".
- The following instruction was added before Diagnostic questions: "If you answered one, the next questions are to diagnose the headache. If you answered more than one, please focus upon the headache type that on the whole bothers you the most".
- Under the sub-section "Diagnostic Questions": "How long would it last if you did not take medication?" was split into two questions to make it easier to understand and answer – one requests the number of hours and the other requests the number of days. For the question "How bad is this type of headache?", an additional option was added (severe).

### For Section C the following changes were made:

- Under the sub-section "Impact Questions":
  - A question was added "Do you think your headaches have had a negative effect on your mental health?".

## 3.8.3 Pilot study

A pilot study aids in identifying potential barriers to the methodology of a study, so they can be addressed to ensure the success of the larger scale study (Jairath, Hogerney and Parsons 2000). This consisted of four participants. The participants were required to have the same characteristics as those in the main study.

### Inclusion criteria

- 1. Lecturers employed at the Durban University of Technology.
- 2. Have completed and signed the letter of information and informed consent.

### Exclusion criteria

- 1. Prospective participants that were unwilling to sign the letter of information and informed consent.
- 2. Persons who attended the focus group.

### 3.8.4 Outcome of the Pilot Study

The following questions were changed/phrased differently after the questionnaire was answered online. An option of <1 hour was added for many questions.

### Section A

- Under the sub-section "Psychological":
  - "Are you usually highly stressed?" was changed to "How would you describe your stress levels?" with options.
- Under the sub-section "Work Related":
  - More options were provided instead of <2years or >2years.
  - "Did you experience any difficulty working from home?" was changed to "Did you experience any difficulty with the transition to working from home during COVID-19"?

### Section B

- Under the sub-section "Screening":
  - "Have you ever had a headache?" was changed to "Are you a headache sufferer?".
- Under the sub-section "Diagnostic Questions":
  - "Is your last answer with or without medication?" was changed to "With regards to your last answer, is this when you have taken medication?".
  - -The word "cyclical" was changed to "cycles" and "occurs in a routine pattern" was added.

## 3.9 DATA ANALYSIS

Data analysis involves classifying, collecting, controlling and summarising the information, followed by describing them in significant terms (Brink *et al.*, 2012).

## 3.9.1 Statistical Analysis

The responses from the survey were automatically coded in QUESTIONPRO® and, on completion of the survey, the data were imported into IBM SPSS version 28. Descriptive statistics provide information such as the average age or the number of males or females (Greasley, 2008). Descriptive analysis included counts and frequencies for categorical data and means and standard deviations for interval data. Inferential statistics was used to determine patterns and relationships in the data. Univariate analysis using the chi-square goodness-of-fit-test was used on categorical variables to test if any of the response options were selected significantly more/less compared to others. Pearson's chi-square test or Fisher's

Exact test were used to determine whether a significant relationship existed between risk factors and headache. Binomial tests were applied to questions with only two response options to test whether a significant proportion of respondents selected one of two possible options.

## 3.10 DATA COLLECTION PROCEDURE

Once the study received full ethical approval from the DUT's Institutional Research Ethics Committee (IREC), the questionnaire was sent out to all employed lecturers at DUT via the DUT server. The e-mail included the letter of information and a link to the questionnaire. This questionnaire should have taken between 10-15 minutes to complete. The researchers contact details was provided in the case of respondents experiencing difficulties in completing the questionnaire. Following the initial e-mail requesting participation, two to three follow up e-mails were sent, with a one week interval between each reminder e-mail.

## 3.11 ETHICAL CONSIDERATIONS

To conduct this study, ethical approval from IREC (Appendix 2) was obtained. In addition, Gatekeeper's permission was received prior to the start of the study to access the employees of the institution (Appendix 1). Respondents were required to provide informed consent, after having read the letter of information (Appendix 4), to participate in the study. This is in line with the principle of autonomy which refers to an individual being able to make individual choices with no manipulation of external forces (Christman, 2003).

There are three essential principles of ethics which guide researchers during the research process. This includes beneficence, respect for persons and justice. The principles are based on human rights which need to be secure; the right to autonomy, secrecy and own discretion, confidentiality, fair treatment and to be protected from harm and discomfort (Brink *et a*l., 2012).

### Beneficence

A compiled report was provided to DUT, where pooled data were referred to, ensuring anonymity of the respondents, to allow the institution to address issues that the study may have revealed. Respondents were not forced to participate in this study and were able to withdraw at any point without fear or harm. In addition, the study provided information regarding headaches in lecturers which will be of use to the medical fraternity, fulfilling the principle of beneficence.

#### **Respect for persons**

No personal information was on the survey, such as the names of participants, identification details or contact details. All respondents' information was coded to ensure confidentiality and anonymity. Prior to participation, all information about the study was disclosed to respondents in the letter of information and their initials were required on the informed consent form prior to answering the questionnaire. Participation in this study was voluntary with no remuneration.

#### Justice

The study data obtained will be kept for five years in the Department of Chiropractic and thereafter shredded if printed, and if stored digitally, the flash drive will be formatted. There was a fair and equal chance for all potential respondents to participate. There was no exclusion based on ethnicity or gender, which is in keeping with justice, which is the fair, appropriate and equitable treatment of individuals (Varkey, 2021).

## 3.12 CONCLUSION

The study used a quantitative, descriptive cross-sectional design, where 685 lecturers employed at DUT were invited to participate. This included full-time, part-time and contract staff, of which a minimum response rate of 247 was needed. The HARDSHIP questionnaire was administered electronically via QUESTIONPRO®. Respondents had a choice whether or not to participate in this study, and no remuneration was offered. The responses were automatically coded in QUESTIONPRO® and the data was then imported into IBM SPSS version 28.

The next chapter will present the results obtained from the HARDSHIP questionnaire. This will include the prevalence, selected risk factors and burden of headaches in lecturers.

# **CHAPTER FOUR RESULTS**

## **4.1 INTRODUCTION**

This chapter presents the results of the analysis done on the data collected on lecturers via the HARDSHIP questionnaire. This information is presented in the form of graphs and tabulations.

## 4.2 SAMPLE SIZE AND RESPONSE RATE

For this study, a sample size of 247 was required. Since there were 685 registered lecturers at DUT, 685 questionnaires were sent out electronically. Of the 685 questionnaires sent, 332 questionnaires were submitted, however 60 were excluded because they did not answer any questions. A further 17 answered part of the questionnaire but 12 were excluded because they did not fill in the screening question at the start of Section B. The total number of questionnaires retained for the analysis was 255. Since there were 255 respondents who answered the questionnaire fully, a response rate greater than 100% was achieved.

## **4.3 DEMOGRAPHICS**

The statistics gathered on age, sex, race and marital status are presented as follows:

## 4.3.1 Age

The mean age of respondents was 45.66 years, with a standard deviation of 9.727, and a range from 24 to 76 years (Table 4.1).

#### Table 4.1: Age of respondents

Age in years	
Mean age	45.66
Std. deviation	9.727
Minimum age	24
Maximum age	76

### 4.3.2 Gender

From the 255 respondents, there were 110 males (43.1%) and 145 females (56.9%). More females have answered the questionnaire (Table 4.2).

	Table	4.2:	Gender	of	respo	ondents
--	-------	------	--------	----	-------	---------

Gender	Frequency <i>n</i> =255	Percentage
Male	110	43.1%
Female	145	56.9%
Total	255	100%

#### 4.3.3 Race

Black African (n=99, 38.8%) comprised the most respondents, with Indian lecturers being the second most common (n=92, 36.1%), followed by White respondents (n=48, 18.8%), Coloured lecturers (n=14, 5.5%), and those who identified as Other (n=2, 0.8%) (Table 4.3).

Race	Frequency <i>n</i> = 255	Percentage
Black	99	38.8%
White	48	18.8%
Indian	92	36.1%
Coloured	14	5.5%
Other	2	0.8%
Total	255	100%

#### Table 4.3: Race of respondents

### 4.3.4 Marital Status

Most of the respondents were married (n=168, 65.9%), followed by single respondents (n=64, 25.1%), divorcees (n=18, 7.1%), widowed respondents (n=3, 1.2%) and Other (n=2, 0.8%) (Table 4.4).

#### Table 4.4: Marital status of respondents

Marital status	Frequency <i>n</i> = 255	Percentage
Single	64	25.1%
Married	168	65.9%
Divorced	18	7.1%
Widow	3	1.2%
Other	2	0.8%
Total	255	100%

## **4.4 PREVALENCE OF HEADACHES**

Of the total sample (n=255), 51.8% (n=132) were headache sufferers, whereas the remaining 48.2% (n=123) did not experience headaches (Table 4.5).

Headache sufferer	Frequency <i>n</i> = 255	Percentage
Yes	132	51.8%
No	123	48.2%
Total	255	100%

#### Table 4.5: Headache sufferers

Of the 132 headache sufferers, 97.7% (n=129) of respondents had a headache during the last 12 months, and 2.3% (n=3) did not answer the question (Table 4.6). The three respondents who did not answer this question did not answer any of the remaining questions. They were retained to address objective 2 – risk factors for headaches but had no input for anything else.

Table 4.6: Respondents who had a headach	he during the last 12 months	
Lleadeale during the leat 40 months	<b>F</b> actorian and m. (100	

Headache during the last 12 months	Frequency <i>n</i> = 132	Percentage
Yes	129	97.7%
Unanswered	3	2.3%
Total	132	100%

The following questions were calculated using n=129 headache sufferers.

## **4.5 SCREENING**

This section involved screening to check if a headache disorder was present or not.

Of the 129 headache sufferers, a significant 89.1% (n=115) indicated that they sustained headaches between 1–14 days during the last 30 days. There were 10.9% (n=14) of lecturers who indicated that they had a headache on more than 15 days in the last 30 days. A large number (n=92, 71.3%) reported their headache had worsened during the pandemic, and 103 lecturers (79.8%) indicated that wearing a mask aggravated their headaches. From the 129 respondents, 72.1% (n=93) reported that nobody in their family suffered from similar headaches. Most of the lecturers reported taking medication to treat their headaches, and a majority of them, 89.1% (n=115), indicated that they took medication on less than nine days in the last 30 days. There were 83 (64.3%) lecturers who complained of their headaches being one type, whereas 46 (35.7%) lecturers complained of their headaches being more than one type (Table 4.7).

Lecturers who chose more than one type of headache were asked to focus on the one that bothered them the most in order to answer the diagnostic questions.



Figure 4.1: Screening questions and results

## **4.6 DIAGNOSTIC QUESTIONS**

A number of questions were asked in order to diagnose the headaches, and the results are shown in Figure 4.2.

A significant 78.3% (n=101) said that their headache usually lasted for hours. Only 13 (10.1%) lecturers indicated that their headache lasted for minutes, and this was the same as those who indicated that their headache lasted for days. There were two (1.6%) respondents who reported that their headache never goes away.

A follow up question determining whether the previous answer was when the respondents had/had not taken medication showed that the question about duration of headache was answered keeping in mind that most (65.9%, n=85) of them had taken medication.

There were 63 (48.8%) lecturers who reported that their headache lasted for more than 24 hours when they did not take medication. This was followed by 20 respondents whose headaches lasted between 2–4 hours, with the least number (n=6, 4.7%) of respondents indicating that their headache lasted between 12–24 hours.

From the results presented, it is clear that when the respondents had taken medication then most of them reported their headache duration to be hours. However, when they did not take medication, most of them reported that their headache lasted for more than a day.





A significant majority (69%, n=89) indicated that their headaches were moderate, followed by 33 (25.6%), who indicated they were severe; six (4.7%) reported their headaches to be mild and only 1 (0.8%) reported them to be extremely severe.



Figure 4.3: Severity of headache

Since the questionnaire used did not include cluster related questions, the next three questions were added to diagnose CTH. Figures 4.4 - 4.6 show the results for the specific cluster related headache questions asked.

Just under three quarters of respondents (n=94, 72.9%) did not experience the cluster headache symptoms listed previously. The rest of them (27.1%, n=35) did experience the cluster related symptoms, which included their eye turning red on one side, watering on one side, a running and congested nose on one side.



Figure 4.4: Cluster-type headache question

Of the 129 headache sufferers, 99 (76.7%) respondents did not experience their headaches in cycles. The remaining respondents (23.3%, n=30) did experience cyclical headaches.



Figure 4.5: Cluster-type headache question

Most of the respondents (58.1%, *n*=75) did not experience pain in, behind, or around the eye.



Figure 4.6: Cluster-type headache question

A little over half of the respondents (n=72, 55.8%) described their headache as being throbbing or pulsating, whilst 44.2% (n=57) described their headache as being pressing, squeezing or tightening.



Figure 4.7: Type of pain

Of the 129 respondents, 51.9% (n=67) had unilateral headaches and 48.1% (n=62) had bilateral headaches. More than half of the respondents, 59.7% (n=77) said that exercise did not make their headaches worse.

A significant majority of 68.2% (*n*=88) usually did not feel nauseated and 93.8% (*n*=121) did not vomit during their headache episodes.

A total of 66 respondents (51.2%, *n*=66) reported that light bothers them, and 53.5% (*n*=69) reported that noise bothers them.



Figure 4.8: Type of pain

### Diagnosis according to the HARDSHIP questionnaire algorithm

Of the 129 respondents experiencing headaches, there were more probable tension type headaches (TTH) (41.9%, n=54), followed by migraine type headaches (MTH) (27.9%, n=36) (Table 4.7).

Table 4.7: Diagnosis	of headaches
----------------------	--------------

Headache during the last 12 months	Frequency <i>n</i> = 132	Percentage
Yes	129	97.7%
Unanswered	3	2.3%
Total	132	100%

## 4.7 IMPACT ON LIFE

These questions pertain to the same respondents (n=129) as those included in the prior analysis.

A significant majority (80.6%, n=104) said that their headache interfered with work, and most of the respondents (76.0%, n=98) avoided telling people that they had headaches.

A vast majority also reported that their headaches have never caused a long-term relationship/partnership to break down; this accounted for 91.5%, n=118.



Many of the respondents (73.6%, *n*=95) indicated that their headache has a negative effect on their mental health.

Figure 4.9: The impact of headaches

The majority of the respondents (55.8%, n=72) did not miss work due to their headache, whereas a majority (57.4%, n=74) of respondents missed family/leisure activities between 1–14 days in the last three months.



Figure 4.10: Missed workdays/family activities due to headaches

Of 129 respondents, most of the respondents (54.3%, n=70) said that their family and friends partly understand and accept their headaches. This was followed by those who answered no (26.4%, n=34) and (19.4%, n=25) felt that their headaches are fully understood and accepted by family and friends.



Figure 4.11: Missed workdays/family activities due to headaches

A significant majority (81.4%, n=105) said that they take medication when they have a headache at work. The other respondents had stated that they would go home when they had a headache (6.2%, n=8).

When asked what the respondents do when they have a headache at home, the majority (39.5%, n=55) said they would take medication, followed by sleeping (34.1%, n=44), and others said they would lie down (23.3%, n=30).



It can be concluded that the vast majority take medication when at work or at home.

Figure 4.12: Relieving a headache at home VS at work

## **4.8 SELECTED RISK FACTORS**

The selected risk factors include the results from the social, medical, psychological, work-related and COVID-19 related risk factors. The total sample (n=255) answered the questions on risk factors.

## 4.8.1 Social History Results

There was no significant relationship between smoking and headaches, but there was a greater amount of headache sufferers who smoked daily, compared to smoking weekly or occasionally. There was a significant relationship between how much alcohol they drank and the headaches; having 7–9 units of alcohol a week was significantly related to suffering from headaches.

Compared to those who did not have headaches, a significant number of those who had headaches only had on average 1–4 hours of sleep a night.

Not doing exercise was related to having headaches, and those who exercised for the least number of hours (1–4) had shown a greater percentage of headache sufferers (Table 4.8).

14	Response	Respo Freque	nses as ncy (%)	<b>V</b> 2	-16	
item	categories	Headache YES	Headache NO	- X-	ατ	p-value
De veu emeke?	Yes ( <i>n</i> =40)	24 (60.0)	16 (40.0)	1 220	4	256
Do you smoke?	No ( <i>n</i> =215)	108 (50.2)	107 (49.8)	1.209	I	.200
	Daily ( <i>n</i> =30)	17 (56.7)	13 (43.3)	_		
How often do you smoke?	Weekly (n=7)	5 (71.4)	2 (28.6)	.661#		.856
	Occasionally (n=3)	2 (66.7)	1 (33.3)	-		
Do you drink clochol?	Yes ( <i>n</i> =95)	47 (49.5)	48 (50.5)	- 318	1	572
Do you drink alcohol?	No ( <i>n</i> =160)	85 (53.1)	75 (46.9)	310	I	.575
How many units do you drink per week?	1 – 3 ( <i>n</i> =63)	36 (57.1)	27 (42.9)	_		
	4–6 ( <i>n</i> =24)	6 (25.0)	18 (75.0)	14.763 #		001*
	7 – 9 ( <i>n</i> =5)	5 (100)	0 (0)			.001
	10+ ( <i>n</i> =3)	0 (0)	3 (100)	_		
Do you usually eat	Yes ( <i>n</i> =152)	78 (51.3)	74 (48.7)	020	1	960
three meals a day?	No ( <i>n</i> =103)	54 (52.4)	49 (47.6)	.030	I	.002
On average, how	1 – 4 ( <i>n</i> =37)	25 (67.6)	12 (32.4)	_		
many hours of sleep	5–9 ( <i>n</i> =217)	107 (49.3)	110 (50.7)	5.213 #		.050*
do you get per night?	10+ ( <i>n</i> =1)	0 (0)	1 (100)	-		
Do you exercise?	Yes ( <i>n</i> =139)	57 (41.0)	82 (59.0)	- 14 162	1	< 001*
	No ( <i>n</i> =116)	75 (64.7)	41 (35.3)	- 14.102	I	<.001
For how many hours	1–4 ( <i>n</i> =106)	46 (43.4)	60 (56.6)	_		
per week do you	5–9 ( <i>n</i> =25)	9 (36.0)	16 (64.0)	1.238 #		.568
exercise?	10+ ( <i>n</i> =8)	2 (25.0)	6 (75.0)	=		

#### Table 4.8: Social risk factors

#### Table 4.9: Medical risk factors

14 0	Response	Responses as Frequency (%)		<b>V</b> 2	-16	p-
	categories	Headach e YES	Headach e NO	Χ-	α	value
Have you had a head injury	Yes ( <i>n</i> =11)	6 (54.5)	5 (45.5)	000	4	050
in the last 3 months?	No ( <i>n</i> =244)	126 (51.6)	118 (48.4)	.030	1	.850
	Yes ( <i>n</i> =166)	100 (60.2)	66 (39.8)	12 696	1	000
Do you wear glasses:	No ( <i>n</i> =89)	32 (36.0)	57 (64.0)	- 13.000	I	.000
	Excellent (n=27)	5 (18.5)	22 (81.5)			
	Very good ( <i>n</i> =62)	24 (38.7)	38 (61.3)			
How would you describe your health?	Good ( <i>n</i> =149)	96 (64.4)	53 (35.6)	27.241	1	.000
,	Fair ( <i>n</i> =16)	7 (43.8)	9 (56.3)			
	Poor ( <i>n</i> =1)	0 (0)	1 (100)			
Ligh blood processo	0 ( <i>n</i> =195)	101 (51.8)	94 (48.2)	000	1	096
	Yes ( <i>n</i> =60)	31 (51.7)	29 (48.3)	.000	I	.900
Low blood processo	0 ( <i>n</i> =242)	123 (50.8)	119 (49.2)	1 674	1	106
Low blood pressure	Yes ( <i>n</i> =13)	9 (69.2)	4 (30.8)	1.074	I	.190
Dishataa	0 ( <i>n</i> =230)	123 (53.5)	107 (46.5)	- 2759	1	007
Diabetes	Yes ( <i>n</i> =25)	9 (36.0)	16 (64.0)	2.759	I	.097
Cerebral stroke	0 ( <i>n</i> =255)	132 (51.8)	123 (48.2)			
Anomio	0 ( <i>n</i> =235)	119 (50.6)	116 (49.4)	1 5 2 2	1	217
	Yes ( <i>n</i> =20)	13 (65.0)	7 (35.0)	1.525	I	.217
	0 ( <i>n</i> =251)	131 (52.2)	120 (47.8)	1 166	1	280
	Yes ( <i>n</i> =4)	1 (25.0)	3 (75.0)	1.100	I	.280
Capeer	0 ( <i>n</i> =253)	130 (51.4)	123 (48.6)	1 070	1	171
Cancer	Yes ( <i>n</i> =2)	2 (100)	0 (0)	1.878	1	.171
Seizures	0 ( <i>n</i> =253)	132 (52.2)	121 (47.8)	0.400	1	1 1 1
	Yes ( <i>n</i> =2)	0 (0)	2 (100)	2.103	I	.141
Cipucitio	0 ( <i>n</i> =225)	110 (48.9)	115 (51.1)	6 225	1	012
	Yes ( <i>n</i> =30)	22 (73.3)	8 (26.7)	0.000	I	.012
Havfovor	0 ( <i>n</i> =237)	120 (50.6)	117 (49.4)	1 700	1	100
Hayfever	Yes ( <i>n</i> =18)	12 (66.7)	6 (33.3)	1.122	I	.109

### 4.8.2 Medical History Results

Of those who had a head injury in the last three months, 54.5% (*n*=6) complained of headaches.

Wearing glasses was significantly associated with headaches, as 60.2% (*n*=100) who wore glasses were headache sufferers.

There was a significant association with general health and headaches; most of the respondents indicated their general health as "good" and 64.4% (*n*=96) of this sample were headache sufferers.

### 4.8.2.1 Chronic illnesses

- High blood pressure of the 60 respondents who reported high blood pressure, 31 (51.5%) respondents reported headaches as well.
- Low blood pressure 13 respondents reported having low blood pressure, of which nine (69.2%) respondents reported headaches as well.
- There seemed to be an association between both high blood pressure and headaches, and low blood pressure and headaches.
- Anemia of the 20 respondents who reported anemia, 13 (65.0%) reported headaches.
- Sinusitis was strongly associated with headaches, as 73.3% (*n*=22) who reported sinusitis reported headaches.
- Hayfever of the 18 respondents who reported hayfever, 12 (66.7%) reported headaches.

## 4.8.3 Psychological Results

Of the total sample (*n*=255), the majority of the respondents reported their stress levels as moderate, followed by high stress levels.

Of the 162 respondents who reported their stress levels as moderate, 93 (57.4%) had headaches.

Of the 45 respondents who reported high stress levels, 28 (62.2%) had headaches.

There was a link between depression and headaches: 33 (12.9%) respondents were diagnosed with depression and of the 33 respondents, 18 (54.5%) were headache sufferers.

There was a link between anxiety and headaches, 39 (15.3%) respondents were diagnosed with anxiety and of the 39 respondents, 23 (59.0%) were headache sufferers.

Itom	Response	Responses as Frequency (%)		<b>v</b> <sup>2</sup>	٩t	p-
nem	categories	Headache YES	Headache NO	~	u	value
How would you rate your stress levels?	High ( <i>n</i> =45)	28 (62.2)	17 (37.8)	_		000
	Moderate (n=162)	93 (57.4)	69 (42.6)	- 22 200 #		
	Mild ( <i>n</i> =39)	11 (28.2)	28 (71.8)	23.200 *		.000
	None ( <i>n</i> =9)	0 (0)	9 (100)			
Have you been diagnosed with depression?	Yes ( <i>n</i> =33)	18 (54.5)	15 (45.5)	- 447	1	.732
	No ( <i>n</i> =222)	114 (51.4)	108 (48.6)	.117		
Have you been diagnosed with anxiety?	Yes ( <i>n</i> =39)	23 (59.0)	16 (41.0)			.328
	No ( <i>n</i> =216)	109 (50.5)	107 (49.5)	.958	1	

#### Table 4.10: Psychological risk factors

Item	Response	Responses as	Frequency (%)	<b>X</b> <sup>2</sup>	df	p-value
	categories	Headache YES	Headache NO			•
	0-3 years ( <i>n</i> =29)	14 (48.3)	15 (51.7)			
How long have you been a	4-8 years ( <i>n</i> =74)	42 (56.8)	32 (43.2)	1.207 #		.764
lecturer?	9-13 years ( <i>n</i> =54)	26 (48.1)	28 (51.9)	- -		
	+ 14 years ( <i>n</i> =98)	50 (51.0)	48 (49.0)			
	Full-time ( <i>n</i> =227)	118 (52.0)	109 (48.0)			
appointment	Part-time (n=8)	3 (37.5)	5 (62.5)	.766#		.700
	Contract (n=20)	11 (55.0)	9 (45.0)	-		
	1-3 ( <i>n</i> =89)	38 (42.7)	51 (57.3)			
How many days per week do	4-5 ( <i>n</i> =158)	89 (56.3)	69 (43.7)	4.608 #		.104
you lecture?	6 – 7 ( <i>n</i> =8)	5 (62.5)	3 (37.5)	•		
	< 1 ( <i>n</i> =16)	7 (43.8)	9 (56.3)			
How many hours do you	1-3 ( <i>n</i> =177)	93 (52.5)	84 (47.5)	.480 #		.827
	> 4 ( <i>n</i> =62)	32 (51.6)	30 (48.4)	•		
Online lectures	Yes ( <i>n</i> =231)	119 (51.5)	112 (48.5)			
Face to face theory lectures	Yes ( <i>n</i> =91)	37 (40.7)	54 (59.3)			
Face to face practical lectures	Yes ( <i>n</i> =63)	31 (49.2)	32 (50.8)			
Line manager support	Yes ( <i>n</i> =144)	54 (37.5)	90 (62.5)	26 050	1	000
	No ( <i>n</i> =111)	78 (70.3)	33 (29.7)	20.939		.000
	Remotely (n=11)	2 (18.2)	9 (81.8)	_		
campus or both?	On campus (n=94)	47 (50.0)	47 (50.0)	5.776 #		.051
	Both ( <i>n</i> =150)	83 (55.3)	67 (44.7)			
Did you experience any	Yes ( <i>n</i> =151)	104 (68.9)	47 (31.1)	42 407	4	000
difficulty with the transition?	No ( <i>n</i> =104)	28 (26.9)	76 (73.1)	43.407	1	.000
Now that this go are alowly	Well ( <i>n</i> =108)	35 (32.4)	73 (67.6)			
setting back to normal, how	Moderately (n=140)	95 (67.9)	45 (32.1)	32.561 #		.000
	Not well (n=7)	2 ( 28.6)	5 (71.4)			
	< 1 ( <i>n</i> =4)	0 (0)	4 (100)			
How many hours do you	1 - 4 ( <i>n</i> =110)	58 (52.7)	52 (47.3)	4 662#		.197
day?	5–9 ( <i>n</i> =127)	68 (53.5)	59 (46.5)	4.002		
2	> 10 ( <i>n</i> =14)	6 (42.9)	8 (57.1)			
Do you work at a desk when	Yes ( <i>n</i> =167)	77 (46.1)	90 (53.9)	6 202	1	012
working from home?	No ( <i>n</i> =88)	55 (62.5)	33 (37.5)	0.202	I	.013
la vour chair comfortable?	Yes ( <i>n</i> =203)	105 (51.7)	98 (48.3)	001	1	000
Is your chair connortable?	No ( <i>n</i> =52)	27 (51.9)	25 (48.1)	.001	I	.900
Is your laptop/computer screen	Yes ( <i>n</i> =101)	31 (30.7)	70 (69.3)			
at the same level as your eyes?	No ( <i>n</i> =154)	101 (65.6)	53 (34.4)	29.740	1	.000
	< 1 ( <i>n</i> =4)	2 (50)	2 (50)	_		
How many hours do you	1 – 4 ( <i>n</i> =65)	27 (41.5)	38 (58.5)	0 071 #		000*
per dav?	5 – 9 ( <i>n</i> =158)	93 (58.9)	65 (41.1)	0.071″		.023
por day:	> 10 ( <i>n</i> =28)	10 (35.7)	18 (64.3)	•		
	< 1 ( <i>n</i> =38)	10 (26.3)	28 (73.7)			
How many hours do you	1 - 4 ( <i>n</i> =100)	44 (44.0)	56 (56.0)			
spend using a phone per day?	5 - 9 (n=93)	58 (62 4)	35 (37 6)	26.036	3	.000
<u> </u>	> 10 (n=24)	20 (83 3)	4 (16 7)			
	Often $(n=53)$	22 (11 5)	31 (58 5)			
Does your work interfere with	Sometimes	400 (00 0)	70 (00.0)		~	
family/leisure time?	( <i>n</i> =181)	109 (60.2)	(2 (39.8)	25.997	2	.000
	Never ( <i>n</i> =21)	1 (4.8)	20 (95.2)			

#### Table 4.11: Work-related risk factors

#### 4.8.4 Work-Related Risk Factors

Of the total sample (n=255), most of the lecturers had lectured for more than 14 years and most of them were full-time staff. The work-related factors are seen in Table 4.11.

Of the 255 lecturers, more than half (n=158, 61.9%) reported lecturing 4–5 days per week. Of these, 56.3% (n=89) suffered with headache. When asked about the amount of time spent lecturing, 177 (69.4%) lecturers reported lecturing 1–3 hours a day, of which 52.5% (n=93) reported headaches.

A majority of lecturers (90.6%, n=231) reported being involved in online lecturing, with 51.5% (n=119) being headache sufferers.

There was a significant relationship between line manager support and headaches. Of the 144 lecturers who had support, 90 (69.5%) respondents did not experience headaches. Of the 111 who did not have support, 78 (70.3%) respondents experienced headaches.

Most of the lecturers (n=150) reported working both remotely and on campus, and 55.3% (n=83) of them reported headaches.

A significant relationship between WFH and headaches were reported, as there were 151 respondents who indicated that they experienced difficulty with the transition, of which 68.9% (n=104) reported headaches. There were 40.8% (n=104) of lecturers who did not experience difficulty and 73.1% (n=76) did not experience headaches.

There was a significant association between coping and headaches, since of the respondents who indicated that they were coping well, 67.6% (n=73) did not experience headaches. However, 67.9% (n=95) who were coping moderately experienced headaches.

Most of the respondents (49.8%; n=127) indicated that they sat at a desk between 5-9 hours, of which 53.5% (n=68) had headaches.

A significant association between working at a desk and headaches were reported, as majority of those who worked at a desk when WFH did not have headaches (53.9%, n=90), and 62.5% (n=55) of respondents who did not work at a desk when WFH had headaches.

There was a significant association between the laptop/computer screen level and headaches; of those who reported that their laptop/computer screen is at the same level as their eyes, 69.3% (n=70) were not headache sufferers. From the lecturers who indicated that it was not at the same level, 65.6% (n=101) reported headaches.

A significant relationship existed between the duration of laptop/computer usage and headaches, as using a laptop/computer for 5–9 hours per day was strongly related to suffering from headaches (58.9%, n=93)

There was a significant relationship between the duration of phone usage and headaches: of those who used a phone for 5-9 hours per day, 62.4% (n=58) reported headaches. Those who used a phone for >10 hours per day reported 83.3% (n=20) of headache sufferers.

A significant association between work interfering with family/leisure time and headaches were revealed since those who reported that work never interferes with family/leisure time, 95.2% (*n*=20) did not have headaches. Those who indicated that work sometimes interferes with family/leisure time, 60.2% (*n*=109) had headaches.

## 4.8.5 COVID-19 Related Risk Factors

There was a significant relationship between COVID-19 testing and headaches. The majority of the respondents who tested positive for COVID-19 were headache sufferers: 63 (64.9%) of 97 respondents had tested positive and reported headaches. The majority of those who did not test positive were not headache sufferers, 89 (56.3%) of 158 did not test positive and did not report headaches.

There was a significant difference between experiencing loneliness during the lockdown and headaches. There were 123 respondents who reported loneliness and 69.1% (n=85) reported headaches, whereas there were 132 respondents who did not report loneliness and 64.4% (n=85) did not experience headaches.

A vast majority of indicated that their workload increased during the pandemic, of those respondents 116 (55.8%) reported headaches.

ltom	Response _ categories	Responses as Frequency (%)		<b>v</b> 2	તા	n voluo
nem		Headache YES	Headache NO	- ^-	u	p-value
Have you ever tested	Yes ( <i>n</i> =97)	63 (64.9)	34 (35.1)	- 10.898	1	.001
positive for COVID-19?	No ( <i>n</i> =158)	69 (43.7)	89 (56.3)			
Did you ever experience	Yes ( <i>n</i> =123)	85 (69.1)	38 (30.9)	- 28.617	1	.000
teelings of loneliness during the lockdown?	No ( <i>n</i> =132)	47 (35.6)	85 (64.4)			
Has your workload increased	Yes ( <i>n</i> =208)	116 (55.8)	92 (44.2)	- 7.248	1	.007
during COVID-19?	No ( <i>n</i> =47)	16 (34.0)	31 (66.0)			

#### Table 4.12: COVID-19 related risk factors

## 4.9 CONCLUSION

Approximately half of those sampled in this study were headache sufferers, and a significant number of lecturers reported headaches between 1–14 days in the last 30 days. A TTH was the most common, followed by MTHs. Of the questions about social history, the risk factors

which were significantly related to headaches included alcohol, lack of sleep and lack of exercise. According to the medical-related questions, wearing glasses, those who described their general health as good, and sinusitis, had a significant relationship with headaches.

Not having line manager support, transitioning to WFH, coping at work, working at a desk when at home, screen being at eye level, duration of using a computer/laptop and phone and work interfering with family/leisure time were all said to have been significantly related with headaches.

From the COVID-19 related risk factors, feelings of loneliness during the lockdown had a significant relationship with headaches. Upon impact questions, it was revealed by most respondents that headaches affect their work and mental health.

The next chapter will present an interpretation, discussion and summary of findings and guidelines.

# **CHAPTER FIVE DISCUSSION**

In this chapter, the significant findings from the results in the previous chapter are discussed and will be compared to studies of a similar nature. This includes the prevalence, selected risk factors and burden of primary headaches.

## **5.1 DEMOGRAPHIC FACTORS**

## 5.1.1 Age

The mean age of respondents was 45.66 years with a standard deviation of 9.727 and a range from 24–76 years (Table 4.1). This study showed no relationship between age and headaches.

Contrary to that, Molarius, Tegelberg and Ohrvik (2008) revealed that headaches were more common in individuals between 22–55 years. According to Radtke (2009), headaches in a German population increased with age until 50, and declined thereafter. A study on the adult population in Jordan revealed that the age groups of 18–29 and 30–39 (82.3%) had the highest headache sufferers, followed by those who were 50 years and older (81.7%) (Alzoubi *et al.* 2009). Headaches were said to reduce in individuals older than 65 (Ruiz *et al.*, 2014). According to Straube and Andreou (2019), primary headaches usually start around five years old and are most common around 20–40 years, thereafter declining with older age.

## 5.1.2 Sex

Gender was shown to be significantly related to headaches. Of the 132 headache sufferers, 63.4% were females and 36.4% were males. Females were more prone to headaches, whereas males were more likely to not suffer from headaches, as 63.6% were not headache sufferers.

There are studies in support of the statement which show that there were more female headache sufferers compared to male sufferers (Molarius, Tegelberg and Ohrvik, 2008; Radtke, 2009; Yu *et al.* 2012; Mengistu and Alemayehu 2013).

According to Chowdhury (2012), TTH is not gender inclined. However, a study done in the Turkish population revealed that TTH was more common in females with the male: female ratio being 1: 6 respectively (Bayraktutan *et al.*, 2014).

MTH is three times more common in females than males (GBD, 2018), and periodic fluctuations of oestrogen levels during a female's ovarian cycle makes them more prone to migraines (Vetvik and MacGregor, 2017; Delaruelle *et al.*, 2018).

CTH was considered a male disorder, but in recent times this has been described more frequently in females (Allena *et al.*, 2019).

## 5.1.3 Race

Of 255 respondents, the most common race was Black (38.8%), followed by Indian (36.1%), White (18.1%), Coloured (5.5%) and Other (0.8%) (Table 4.3). In this study, there was no significant relationship between race and headache, but of the 48 Whites who answered the questionnaire, 64.6% were headache sufferers compared to 35.4% who were not headache sufferers.

In contrast, a study conducted in Ohio found that headaches were more common and of a more severe nature in the Black populace when compared to Whites (Heckman, Merrill and Anderson, 2013). According to Loder, Sheikh and Loder (2015), MTH in the United States is most common in Native Americans (17.7%), followed by Whites (15.5%), Hispanics (14.5%), Blacks (14.45) and then Asians (9.2%).

## 5.1.4 Marital Status

This study showed a significant relationship between marital status and headaches. Of the 64 single respondents, 62.5% did not have headaches, but of the 168 married respondents 59.5% were headache sufferers. Those who were married were more prone to headaches.

This finding was consistent with other studies. A study conducted among medical staff in South China revealed a higher prevalence of MTH in married women compare to unmarried women (Xie *et al.*, 2020). In an earlier study, having migraines daily appeared to be due to marital stress (Featherstone and Beitman, 1984). A study conducted on 500 people in Yazd showed that the prevalence of MTH and TTH was higher in married individuals compared to unmarried individuals, which may be caused by stress in marital life due to concerns regarding the future of children, economic problems and routine disputes (Momayyezi, Fallahzadeh and Momayyezi, 2014). Ayatollahi and Cheraghian (2005) showed a significance between marital status and headache.

## **5.2 PREVALENCE OF HEADACHES**

This study indicated that 51.8% of lecturers were headache sufferers. A similar prevalence was identified in a study done in Oman which was 45% (Deleu and Hanssens, 1999), but this was on university students. In support of these findings, a study that aimed to update the documentation of headache epidemiology, from 357 publications globally, found that the prevalence of active headache was 52.0% (Stovner *et al.*, 2022).

In this study, probable TTH (41.9%) was most prevalent followed by MTH (27.9%) and CTH (10.1%). This is comparable to a study done among adults in Jordan which revealed that TTH was the most prevalent among other types (Alzoubi *et al.*, 2009). A study conducted in the Asian region also had a higher prevalence in TTH and reported the general prevalence of TTH to range between 15.6%–25.7% (Wang, 2003). In adults, the mean global prevalence for TTH is 42% (Ferrante *et al.*, 2013).

Contrary to the MTH findings, an Asian study revealed MTH to be between 8.4%–12.7% (Wang, 2003). An estimated prevalence of MTH in the world was said to be 11% (Stovner *et al.*, 2007), which was much lower than the findings in this study.

## **5.3 RISK FACTORS**

The selected risk factors are discussed in the following sections.

### 5.3.1 Social Risk Factors

### 5.3.1.1 Smoking and Headaches

This study did not show a significant relationship between smoking and headaches, but, of the 40 respondents who smoked cigarettes, 60% (*n*=24) were headache sufferers. According to Volans and Castleden (1976), their study showed no relationship between smoking and MTH. Occasional/current smokers were said to be one of the lifestyle factors that increases the prevalence of headaches (Molarius *et al.*, 2008). A study conducted on 51 383 subjects found that headache prevalence was higher in smokers, especially in those under 40 years who smoked more than 10 cigarettes a day (Aamodt *et al.*, 2006). Another study found that the prevalence of active MTH was greater in migraineurs compared to non-migraineurs, and there was a connection between migraine attacks and the number of smoked cigarettes (Lopez-Mesonero *et al.*, 2009). Population studies have shown strong evidence of an association between CTHs and cigarette smoking (Raucci *et al.*, 2020). In contrast, more recently, the findings from a study were not in support of a strong causal relationship between any type of headache and smoking intensity (Johnsen *et al.*, 2018), as supported by this study.

## 5.3.1.2 Alcohol and Headaches

This study showed a significant relationship between alcohol and headaches, those who drank 7-9 units a week were more likely to suffer with headaches.

Alcohol consumption, followed by a few other factors, were noted as the trigger in 10 patients who presented with CTH (Xie e*t al.,* 2013). In about one-third of MTH sufferers in retrospective studies, alcohol has been reported as a trigger for MTH. Some studies have

found that other primary headaches are also triggered by alcohol (Panconesi, 2016). In contrast, migraine prevalence was said to decrease with an increased amount of alcohol, in comparison to alcohol abstinence (Aamodt *et al.*, 2006). More recently, alcohol has been shown to increase the development of migraine by up to 51% (Peroutka, 2014).

This study did not address whether or not the respondents thought alcohol was related to their headache and, thus, this warrants further investigation.

### 5.3.1.3 Sleep and Headaches

A significant relationship existed between lack of sleep and headaches. In support of this, there was a significant relationship shown between MTH and sleep patterns (Momayyezi *et al., 2*015). Population based studies have also shown a correlation between headaches and sleep problems (Momayyezi *et al., 2*015).

A recent study among lecturers continuing their postgraduate education, revealed that of 41 lecturers, the percentage of insomniacs was 24.4% (Daradila *et al.*, 2021). According to Kelman and Rains (2005), MTHs were reduced with an improvement of duration of sleep. In patients who suffer from MTH or TTH, the headache-related impact may be increased due to poor sleep quality (Cho *et al.*, 2020). Studies have shown a relationship between headache frequency and poor quality of sleep in MTH sufferers (Song, 2018; Walters, Hamer and Smitherman, 2014). Other studies have suggested that deprivation of sleep was associated with an increased hyper-excitability of the brain, which, in turn, increased the susceptibility of the brain to MTH attack (Scalise *et al.*, 2006; Lang *et al.*, 2004).

### 5.3.1.4 Exercise and headaches

In this study, a lack of involvement in exercise was related to headaches. Of the 116 respondents who did not exercise, 64.7% (*n*=75) were headache sufferers.

These findings correlate with other studies which suggest that physical inactivity has an association with headache disorders (Molarius, Tegelberg and Ohrvik, 2008; Rasmussen, 1993; Varkey *et al.*, 2008). Another study has also shown a relationship between physical inactivity and an increased headache attack frequency (Varkey *et al.*, 2008). It has been shown that exercise has a protective influence on mental health, especially in relieving stress (Churchill *et al.*, 2021). This may be a likely explanation for physical inactivity having an association with headaches.

In contrast, experimental studies have reported exercise as a trigger factor for MTH, mechanisms including acute release of neuropeptides (Amin et al., 2018). However, although exercise may be a trigger for MTH, regular exercise could possibly have a prophylactic effect on the frequency of MTH (Amin et al., 2018).

### 5.3.2 Medical Risk Factors

#### 5.3.2.1 Head Injury and Headache

The most common reported symptom following a head injury is a headache (Lucas *et al.*, 2014). In this study, of those who had a head injury, 54.5% (*n*=6) complained of headaches but a significant relationship was not noted. A population-based study found that the risk of a new onset of headache suffering, and exacerbation was increased by exposure to head injury (Nordhaug *el al.*, 2018).

#### 5.3.2.2 Glasses and Headache

The findings of this study revealed a significant correlation between wearing glasses and headaches, as 60.2% (*n*=100) who wore glasses were headache sufferers. A population-based study in India had shown that 43.8% of respondents had discontinued wearing glasses. The reasons given were that the spectacles were not comfortable, or they thought that the prescription was not correct, which suggested that the refractive services was poor (Dandona *et al.*, 2002). However, according to Jayanand (2002), an ophthalmologist who responded to the findings of the article, revealed that over the years there has in fact been an increase in the number of people who use glasses. The article further mentions that approximately 90% of individuals who had a refraction test were headache sufferers, among other symptoms, and that there is a poor understanding between the difference of eye strain and headache. There is a poor conception that refraction errors may be the leading cause of headaches and the patient is given glasses, but when the headache persists, the patient stops using glasses thinking it is the wrong prescription, when refraction errors were not even the cause of their headache in the first place (Jayanand, 2002).

### 5.3.2.3 Blood pressure and headache

Although not significant, there seemed to be an association between blood pressure and headache in this study. There have been several studies which have shown that high blood pressure is associated with headache (Badran, Weir and McGuiness, 1970; Finocchi and Sassos, 2017; Middeke *et al.*, 2008). There were 31 (51.5%) out of 60 respondents who reported high blood pressure. Other studies have suggested that those with a headache were always at risk for hypertension (Bigal *et al.*, 2010; Scher *et al.*, 2005). There is conflicting literature on the relationship between headache and blood pressure. Some studies have shown no correlation between the two (Couch and Hassanein, 1989; Ho and Benjamin, 2001), whereas another study found a correlation between MTH and blood pressure (Gardner, Mountain and Hines, 1940). A prospective study of 22 685 adults
revealed that high diastolic and systolic pressure was related with a decreased risk of nonmigrainous headache (Hagen et al., 2002).

#### 5.3.2.4 Sinusitis and Headache

The results of this study presented a significant relationship between sinusitis and headache: 73.3% (*n*=22) who reported sinusitis reported headaches. Since the trigeminal sensory nerves innervate all the paranasal sinuses, many noxious stimuli trigger afferent sensory volleys and this results in a headache (Bluementhal, 2001). A study on "Sinus Headache" showed that many of the patients who had chronic rhinosinusitis suffered from headaches (Maurya *et al., 2*019). Similar findings were revealed in another study that showed patients with chronic rhinosinusitis had a higher risk of chronic headache by ninefold (Aaseth *et al., 2*010). Primary headaches including MTH, CTH, TTH, and headaches from refractive errors are often misdiagnosed as headache due to sinus (Maurya *et al., 2*019). Results from a previous study found that of a total of 130 MTH sufferers, 106 were misdiagnosed with sinus headache (AI-Hashel *et al., 2*013).

#### 5.3.3 Psychological Risk Factors

According to headache sufferers, stress is reported as one of the main aggravating factors for headache (Swanson *et al.*, 2013; Boardman *et al.*, 2005). In this study, there were 93 respondents who had moderate levels of stress and headache, and 28 who had high stress levels and headache. A population-based study showed that an increased amount of stress was associated with an increased frequency of headache for all subtypes (Schramm *el al.*, 2014). However, published literature is not sufficient to determine whether primary headache sufferers experience more life stressors or if they are more reactive compared to those who are not headache sufferers (Nash and Thebarge, 2006). A study that observed 114 headache sufferers for 28 days showed a positive correlation between frequency of both headache and stressful events (Marlowe, 1998). According to Smith (2007), the higher education system reports a high level of stress. This is consistent with the findings of this study, as there were more lecturers who reported their stress levels as either high or moderate (*n*=207), compared to those who reported it as mild or none (*n*=48).

The relationship between depression/anxiety and headache is thought to be bidirectional (Malmberg-Ceder *et al.*, 2019). The results of this study show similar findings as 18 respondents who were diagnosed with anxiety were headache sufferers and 23 who were diagnosed with depression were headache sufferers. A large cross-sectional study showed that anxiety and depression were significantly associated with headaches, compared to non-headache sufferers (Zwart, 2003). However, the results of this study were slightly different as there was no significant relationship. Respondents diagnosed with anxiety and

depression were 39 and 33 respectively, but headache sufferers were 23 respondents for anxiety and 18 for depression.

### 5.3.4 Work-Related Risk Factors

Due to the COVID-19 pandemic, there was a shift to online teaching and learning. Since online learning started, the prevalence of headache had increased from 18%–42% (DiSabella, 2020). In this study, of the 90.6% who reported online lecturing, 51.5% reported headaches.

## 5.3.5 COVID-19 Related Risk Factors

The lives of many were changed during the COVID-19 pandemic. Headaches were reported as the most common neurological manifestation of the virus (Mao, 2020; Sharma and Menon, 2022). According to a meta-analysis and systematic review, headaches were more prevalent in patients with COVID-19, by two-fold (Mutiawati *et al.*, 2020). This is consistent with the findings from this study as 97 respondents tested positive for COVID-19, and majority of them (*n*=63, 64.9%) were headache sufferers. In many studies, the rates of headache during the pandemic ranged from 6.5%–71% (Bolay, Gül and Baykan, 2020; Tostmann *et al.*, 2020).

As a result of social distancing and the imposed lockdown, loneliness became a growing concern (Lampraki *et al.*, 2022). In this study, 123 respondents reported loneliness during the lockdown. A study on a Dutch population also reported that the respondents had feelings of loneliness during the lockdown (van Tillburg *et al.*, 2021). In previous studies, there was a significant relationship between social isolation and psychological distress (Kim and Jung, 2021; Taylor *et al.*, 2018) and it is important to note that primary headaches are strongly associated with psychological distress (Kristoffersen *et al.*, 2018).

# **5.4 BURDEN OF HEADACHE**

Headaches can be extremely burdensome for individuals and may have a significant impact on both social and personal lives, causing a reduced quality of life and productivity (Lipton *et al.,* 2007). In this study, 80.6% (*n*=104), a significant majority, reported that their headache interfered with work. In support of this, a study revealed that 21.7% of employees, who were MTH sufferers, and 8.5%, who were TTH sufferers, missed work (Ayatollahi, Sahebi and Borhani-Haghighi, 2009). Another study revealed a greater number of TTH sufferers who missed work, compared to MTH (Oshinaike *et al.*, 2014). However, another study showed the days when productivity at work was reduced was more significant in MTH compared to TTH (Shimizu *et al.*, 2021). In contrast to this, most of the respondents of another study did not miss work, regardless of the type of headache. The respondents with TTH had the least amount of absenteeism, whereas there was more occasional absenteeism for MTH sufferers (Simic *et al.*, 2020).

A majority of respondents (57.4%, n=74) missed family/leisure activities between 1–14 days in the last three months. A study showed that both personal and work life may be affected equally (Bussone *et al.* 2004). In support of this study's findings, it was revealed that in the past month there was reduced participation in family activities in approximately half of those who were MTH sufferers (Buse *et al.*, 2016). MTH sufferers and their spouses reported that the headache caused stress in their relationships (Buse *et al.*, 2016). This may have negatively affected their mental health. In this study, 73.6%, (n=95) respondents indicated that their headache had a negative effect on their mental health.

# **5.5 CONCLUSION**

In summary, a high prevalence of primary headaches (51.8%) in lecturers was noted. Many risk factors were strongly associated with headaches (i.e alcohol, lack of sleep, wearing glasses, sinusitis etc), and this was comparable to other studies. The impact of headaches caused an interference with work and negatively affected the suffers mental health. Family and/or social activities were also disrupted when a headache occurred.

Chapter Six will discuss the conclusions of the study, the limitations, and recommendations for future studies.

# CHAPTER SIX CONCLUSION, LIMITATIONS AND RECOMMENDATIONS

## **6.1 INTRODUCTION**

In this chapter, the study conclusion are presented, followed by the limitations encountered and the recommendations that stem from the project.

## **6.2 CONCLUSION**

For this study, a sample size of 247 was required; however, 255 respondents answered the questionnaire fully, which was a a response rate greater than 100%.

The mean age of respondents was 45.66 years, with a range from 24–76 years. This study showed no relationship between age and headache. There was a significant relationship between gender and headache. Of the 132 headache sufferers, 63.4% were female. There was no relationship between race and headache. However, most of the respondents were Black (38.8%), followed by Indian (36.1%), White (18.1%), Coloured (5.5%) and Other (0.8%). A significant relationship was shown between marital status and headache — of the 168 married participants, 59.5% were headache sufferers.

There was a high prevalence of primary headaches (51.8%) in lecturers across all the faculties at the Durban University of Technology. Probable tension-type headaches were the most prevalent (41.9%), followed by migraine-type headaches (27.9%), with only 10.1% experiencing cluster-type headaches. They reported that their headaches were moderate (69%) to severe (25.6%).

There were many risk factors which were strongly associated with headaches. The main social risk factors included alcohol intake, lack of sleep and lack of exercise. Having 7–9 units of alcohol a week was significantly related to suffering from headaches (100%, n=5). Compared to those who did not have headaches, a significant number (67.6%, n=25) of those who had headaches only had on average 1–4 hours of sleep a night. Not doing exercise was related to having headaches (64.7%, n=75). Those who exercised for the least number of hours (1–4) had shown a greater percentage of headache sufferers (43.4%, n=46).

According to the medical-related questions, wearing glasses (60.2%, n=100) and sinusitis (73.3%, n=22) had a significant relationship with headaches.

Not having line manager support (37.5%, n=54); difficulty with transitioning to WFH (68.9%, n=104); coping moderately at work (67.9%, n=95); not working at a desk when working at home (62.5%, n=55); laptop/computer screen not being at eye level (65.6%, n=101); duration of using a computer/laptop and phone; and work interfering with family/leisure time, were the work-related factors that were significantly related to headaches. Feelings of loneliness during the lockdown had a significant relationship with headaches.

The impact of headaches caused an interference with work and negatively affected the suffers mental health. Family, social or leisure activities were also neglected when a headache occurred.

This study contributes to a growing body of literature on headaches and their prevalence in university lecturers. The study highlights the disability and negative impact that headaches the sufferer. Professions like chiropractic, that can offer holistic and conservative therapies can assist in alleviating this burden.

# 6.3 LIMITATIONS

- This study was conducted at the Durban University of Technology and does not include information on headaches in lecturers from other universities.
- The reliability of this study depended on the respondents and how honest they were when answering the questionnaire.
- Headache features were determined by a questionnaire, which is not as robust as a clinical assessment with a headache diary.
- The reliability of the diagnosis of headaches were dependent on the validity of the HARDSHIP algorithm.

# **6.4 RECOMMENDATIONS**

#### 6.4.1 Recommendation for Research into Headaches

- Future studies should include/assess lecturers from other universities to determine if the findings in this study are similar in other populations.
- Since there were many dropouts, future studies may want to utilise a shorter questionnaire to yield a better response or administering the questionnaire in person.
- Future research could utilise a different methodology, whereby participants are examined in order to reach more accurate diagnosis, rather than utilising a self-report tool like the one used in this study.

Researchers should consider using a mixed methods or a qualitative research paradigm to allow a greater understanding of the impact of the headaches on the respondent.

## 6.4.2 Recommendation for the Durban University of Technology

- The university should invest in mental health programmes to assist staff to deal with stress, anxiety and depression. These can consist of checks or stress management workshops, held regularly, to allow staff to attend.
- The chiropractic, homoeopathy and somatology clinics at DUT could provide information about the treatments they offer, and benefits that headaches suffers may have from utilising their treatments.

In this chapter, the limitations encountered, and recommendations for future studies in the field of headaches and lecturers have been discussed.

# REFERENCES

Aamodt, A.H., Stovner, L.J., Hagen, K., Brathen, G. and Zwart, J. 2006. Headache prevalence related to smoking and alcohol use. The Head-HUNT Study. *European journal of neurology*, 13(11): 1233–1238. Available: <u>https://doi.org/10.1111/j.1468-1331.2006.01492.x</u> (Accessed 7 October 2022).

Aaseth, K., Grande, R.B., Kvaerner, K., Lundqvist, C. and Russell, M.B. 2010. Chronic rhinosinusitis gives a ninefold increased risk of chronic headache. The Akershus study of chronic headache. *Cephalalgia: an international journal of headache,* 30(2): 152–160. Available: <u>https://doi.org/10.1111/j.1468-</u> (Accessed 10 October 2022)

Abboud, H., Abboud, F.Z., Kharbouch, H., Arkha, Y., El Abbadi, N. and El Ouahabi, A. 2020. COVID-19 and SARS-CoV-2 Infection: Pathophysiology and Clinical Effects on the Nervous System. *World Neurosurgery*, 140: 49–53.

Abu-Arafeh, I., Razak, S., Sivaraman, B. and Graham, C. 2010. Prevalence of headache and migraine in children and adolescents: a systematic review of population-based studies. *Developmental Medicine and Child Neurology*; 52(12): 1088–1097

Abou Hashish, E.A., Baatiah, N.Y., Bashaweeh, A.H. and Kattan, A.M. 2022. The online learning experience and reported headaches associated with screen exposure time among Saudi health sciences students during the COVID-19 pandemic. *BMC Medical Education*, 22(1): 226.

Adney, J., Koehler, S., Tian, L. and Maliakkal, J. 2021. Headache and Hypertension in a 9year-old girl. *Pediatrics in review*, 42(7): 383-386. Available: <u>https://doi.org/10.1542/pir.2019-0119</u> (Accessed 20 October 2022).

Al-Hashel, J.Y., Abokalawa, F., Alenzi, M., Alroughani, R. and Farouk, A. 2021. Coronavirus disease-19 and headache; impact on pre-existing and characteristics of de nono: a cross sectional study. *The journal of headache and pain*, 97(22). Available: <u>https://doi.org/10.1186/s10194-021-01314-7</u> (Accessed 5 January 2022).

Al-Hashel, J.Y., Ahmed, S.F., Alroughani, R. and Goadsby, P.J. 2013. Migraine misdiagnosis as a sinusitis, a delay that can last for many years. *The Journal of Headache and Pain*, 14(1): 97. Available: <u>https://doi.org/10.1186/1129-2377-14-97</u> (Accessed 20 September 2022).

Ali, M., Ahmed, K. and Omer. 2022. Prevalence of migraine headaches and their impact on the academic performance of Sudanese medical students using ID-Migraine test as a screening tool: A cross-sectional study. *Brain and behaviour*, 12(5). Available: <u>https://doi.org/10.1002/brb3.2588</u> (Accessed 10 September 2022).

Allena, M., De Icco, R., Sances, G., Ahmad, L., Putorti, A., Pucci, E., Greco, R. and Tassorelli, C. 2019. Gender Differences in the Clinical Presentation of Cluster Headache: A Role for Sexual Hormones? *Frontiers in Neurology*, 10: 1220.

Al Lily, A.E., Ismail, A.F., Abunasser, F.M. and Alhajhoj Alqahtani, R.H. 2020. Distance Education as a Response to Pandemics: Coronavirus and Arab Culture. *Technology in Society*, 63. Available: <u>https://doi.org/10.1016/j.techsoc.2020.101317</u> (Accessed 20 October 2022).

Alzoubi, K. H., Mhaidat, N., Azzam, S. A., Khader, Y., Salem, S., Issaifan, H. and Haddadin, R. 2009. Prevalence of migraine and tension type headache among adults in Jordan. *The Journal of Headache and Pain*, 10: 265–270.

Amin, F.M., Aristeidou, S., Baraldi, C., Czapinska-Ciepiela, E.K., Ariadni, D.D., Lenola, D.D., Fenech, C., Kampouris, K., Karagiorgis, G., Braschinsky, M., Linde, M. and European Headache Federation School of Advanced Studies (EHF-SAS). 2018. *The Journal of Headache and Pain*, 19(1): 83. Available: <u>https://doi.org/10.1186/s10194-018-0902-y</u> (Accessed 8 October 2022).

Amiri, P., Kazeminasab, S., Nejadghaderi, S., Mohammadinasab, R., Pourfathi, H., Araj-Khodaei, M., Sullman, M., Kolahi, A. and Safiri, S. 2021. Migraine: A Review on Its History, Global Epidemiology, Risk Factors, and Comorbidities. *Frontiers in Neurology*, 12: 800605. Available: <u>https://doi.org/10.3389/fneur.2021.800605</u> (Accessed 10 July 2022). Andreou, A.P and Edvinsson, L. 2019. Mechanisms of migraine as a chronic evolutive condition. *The journal of headache and pain*, 20(1): 117. Available: <u>https://doi.org/10.1186/s10194-019-1066-0</u> (Accessed 7 October 2022).

Arnold, M. 2018. Headache classification committee of the international headache society (IHS) the international classification of headache disorders. *Cephalalgia*, 38: 1–211.

Ashina, S., Bendtsen, L. and Ashina, M. 2005. Pathophysiology of tension-type headache. *Current Pain and Headache Reports*, 9(6): 415–422.

Ashina, S., Mitsikostas, D.D., Lee, M.J., Yamani, N., Wang, S., Messina, R., Ashina, H., Buse, D.C., Pozo-Rosich, P., Jensen, R., Diener, H. and Lipton, R. 2021. Tension-type headache. *Nature Reviews Disease Primers*, 7(1): 1–21.

Aurora, S.K. 2004. Pathophysiology of Migraine and Cluster Headaches. *Seminars in Pain Medicine*, 2(2): 62–71.

Ayatollahi, S.M.T. and Cheraghian, B. 2005. An epidemiologic model for risk factors of migraine and tension type headaches among primary schools teachers of Shiraz. *Journal of Kerman University of Medical Sciences*, 12(2): 85–92.

Ayatollahi, S.M.T., Sahebi, L. and Borhani-Haghighi, A. 2009. Epidemiologic and clinical characteristics of migraine and tension-type headaches and estimating their prevalence in Shiraz (southern IRAN) hospitals staff. *Iran Journal of Neurology*, 47(2): 304–313.

Badran, R.H., Weir, R.J. and McGuiness, J.B. 1970. Hypertension and headache. *Scottish medical journal*, 15(2): 48–51. Available: <u>https://doi.org/10.1177/003693307001500202</u> (Accessed 20 October 2022).

Bahra, A. 2020. Other primary headaches-thunderclap-, cough-, exertional-, and sexual headache. *Journal of neurology*, 267: 1554–1566. Available: <u>https://doi.org/10.1007/s00415-020-09728-0</u> (Accessed 12 July 2022).

Bahra, A., May, A. and Goadsby, P.J. 2002. Cluster headache: A prospective clinical study with diagnostic implications. *Neurology*, 58: 354–61.

Bayraktutan, O.F., Demir, R., Ozel, L. Ozdemir, G. and Ertekin, A. 2014. Prevalence of Tension-Type Headache in Individuals Aged between 18-65 Years in the Eastern Parts of Turkey. *The Eurasian Journal of Medicine*, 46(2): 78–83.

Bendsten, L. 2009. Drug and nondrug treatment in tension-type headache. *Therapeutic Advances in Neurological Disorders*, 2(3): 155-161.

Benoliel, R. and Eliav, E. 2013. Primary headache disorders. *Dental clinics of North America*, 57(3): 513–539. Available: <u>https://doi.org/10.1016/j.cden.2013.04.005</u> (Accessed 1 August 2022).

Bigal, M.E., Kurth, T., Santanello, N., Buse, D., Golden, W., Robbins, M. and Lipton, R.B. 2010. Migraine and cardiovascular disease, a population-based study. *Neurology*, 74(8): 628–635. Available: <u>https://doi.org/10.1212/WNL.0b013e3181d0cc8b</u> (Accessed 8 October 2022).

Bigal, M.E. and Lipton R.B. 2006. Modifiable risk factors for migraine progression. *The Journal of Head and Face Pain*, 46(9); 1334–1343.

Blanda, M. and Wright, J.T. 2009. *Headache, Migraine*. Available at: <u>http://emedicine.medscape.com/article/792267</u> (Accessed 6 September 2022).

Blumenfeld, A., Schim, J. and Brower, J. 2010. Pure Tension-type Headache versus Tension-Type Headache in the Migraineur. *Current Pain and Headache Reports*, 14(6): 465–496.

Blumenthal, H.J. 2001. Headaches and sinus disease. *Headache: The Journal of Head and Face Pain*, 41(9): 883–888.

Boardman, H.F., Thomas, E., Millson, D.S. and Croft, P.R. 2005. Psychological, sleep, lifestyle, and comorbid associations with headache. *Headache*, 45(6): 657–669.

Bobker, S.M. and Robbins, M. 2020. COVID-19 and Headache: A Primer for Trainees. *Headache*, 60(8): 1806–1811.

Bolay, H., Gül, A., & Baykan, B. 2020. COVID-19 is a Real Headache! *Headache*, 60(7), 1415–1421. Available: <u>https://doi.org/10.1111/head.13856</u> (Accessed 20 August 2022).

Bove, G. and Nilsson, N. 1998. Spinal manipulation in the treatment of episodic tensiontype headache: A randomized controlled trial. *JAMA*, 280(18): 1576–1579.

Brink, H., van der Walt, C. and van Rensburg, G. 2012. *Fundamentals of Research Methodology for Healthcare Professionals*. 3<sup>rd</sup> ed. Cape Town Juta & Co. LTD.

Broadbent, A., Combrink, H. and Smart, B. 2020. COVID-19 in South Africa. *Global Epidemiology*, 2: 100034. Available: <u>https://doi.org/10.1016/j.gloepi.2020.100034</u> (Accessed 9 July 2022).

Bryman, A. 2012. Social Research Methods. 4th ed. New York: Oxford University Press.

Burch, R., Rizzoli, P. and Loder, E. 2020. The prevalence and impact of migraine and severe headache in the United States: Updated age, sex, and socioeconomic-specific estimates from government health surveys. *The Journal of Head and Face Pain*, 61: 60–88. Available: <u>https://doi.org/10.1111/head.14024</u> (Accessed 9 August 2022).

Burns, N. and Grove, S. K. 2009. *The practice of nursing research: appraisal, synthesis and generation of evidence*. 6<sup>th</sup> ed. Philadelphia: Saunders.

Buse, D.C., Scher, A.I., Dodick, D.W., Reed, M.L., Fanning, K.M., Adams, A.M. Lipton, R.B. 2016. Impact of Migraine on the Family: Perspectives of People with Migraine and Their Spouse/Domestic Partner in the CaMEO Study. *Mayo Clinic Proceedings*, 91(5): 596–611. Available: <a href="https://doi.org/10.1016/j.mayocp.2016.02.013">https://doi.org/10.1016/j.mayocp.2016.02.013</a> (Accessed 9 October 2022).

Bussone, G., S, U., Grazzi, Rigamonti, A. and D'Amico, D. 2004. Disability and quality of life in different headaches: results from Italian studies. *Journal of Neurological Sciences*, 25: S105–S107.

Castaldo, M., Waliszewska-Prosol, M., Koutsokera, M., Robotti, M., Straburzynski, M., Apostolakopoulou, L., Capizzi, M., Cibuku, O., Ambat, F., Frattale, I., Gadzhieva, Z., Gallo, E., Gryglas-Dworak, A., Halili, G., Jusupova, A., Koperskaya, Y., Leheste, A.R. Manzo, M.L., Marcinno, A. and Marino, A. 2022. Headache onset after vaccination against SARS-CoV-2: a systematic literature review and meta-analysis. *The Journal of Headache and Pain*, 23(1): 41.

Castien, R., van der Windt, D., Grooten, A. and Dekker. J. 2011. Effectiveness of manual therapy for chronic tension-type headache: a pragmatic, randomised, clinical trial. *Cephalalgia*, 31(2): 133–143.

Charleston, L. 2021. Headache Disparities in African-Americans in the United States: A Narrative Review. *Journal of the National Medical Association*, 113(2): 223–229.

Chawla, J. 2009. *Migraine Headache*. Available at: <u>http://emedicine.medscape.com/article/114255</u> (Accessed 10 July 2022).

Check, J. and Schutt, R. 2011. *Research methods in education*. Thousand Oaks, California: Sage Publications, 159–185.

Cho, S., Lee, M.J., Park, H.R., Kim, S., Joo, E.Y. and Chung, C. 2020. Effect of Sleep Quality on Headache-Related Impact in Primary Headache Disorders. *Journal of Clinical Neurology*, 16(2): 237–244.

Chowdhury, D. 2012. Tension type headache. *Annals of Indian Academy of Neurology*, 15(1): 83–88.

Christman, J. 2003. Autonomy in moral and political philosophy. *Stanford encyclopaedia of philosophy*. Available: <u>https://plato.stanford.edu/entries/autonomy-moral/</u>(Accessed 4 February 2022).

Chu, H.T., Liang, C.S., Lee, J.T., Yeh, T.C., Lee, M.S., Sung, Y.F. and Yang, F.C. 2018. Associations Between Depression/Anxiety and Headache Frequency in Migraineurs: A Cross-Sectional Study. *Headache*, 58: 407–415.

Churchill, R., Riadi, I., Kervin, L., Teo, K. and Cosco, T. 2021. Deciphering the role of physical activity in stress management during a global pandemic in older adult populations: a systematic review protocol. *Systematic Reviews*, 10: 140. Available: <u>https://doi.org/10.1186/s13643-021-01678-6</u> (Accessed 15 December 2022).

Clarke, C., Howard, R., Rossor, M. and Shorvon, S. 2009. *Neurology: A Queen Square Textbook*. 1<sup>st</sup> ed. UK: Wiley-Blackwell.

Couch, J.R. and Hassanein, R.S. 1989. Headache as a risk factor for in atherosclerosisrelated diseases. *Headache*, 29(1): 49–54.

Crystal, S. and Grosberg, B. 2009. Tension-type headache in the elderly. *Current Pain and Headache Reports*, 13: 474–478.

Dandona, R., Dandona, L., Kovai, V., Giridhar, P., Prasad, M.N. and Srinivas, M. 2002. Population-based study of spectacles use in southern India. *Indian journal of ophthalmology*, 50(2): 145–155.

Daradila, N.P., Utami, D.K., Tini, K. and Widyadharma, I.P. 2021. Proportion and Characteristics of Insomnia among Lecturers Continuing Postgraduate Education at the Faculty of Medicine Udayana University. *International Journal of Research and Review*, 8(12): 438–446. Available: <u>https://doi.org/10.52403/ijrr.20211254</u> (Accessed 20 December 2022).

Delaruelle, Z., Ivanova, T.A., Khan, S., Negro, A., Ornello, R., Raffaelli, B., Terrin, A., Mitsikostas, D.D., Reuter, U. and European Headeache Federation School of Advanced Studies (EHF-SAS). 2018. Male and female sex hormones in primary headaches. *The* 

*Journal of Headache and Pain*, 19(1): 117. Available: <u>https://doi.org/10.1186/s10194-018-0922-7</u> (Accessed 1 November 2022).

Deleu, D. and Hanssens, Y. 1999. Primary chronic daily headache: clinical and pharmacological aspects. A clinic-based study in Oman. *Headache*, 39(6): 432–436. Available: <u>https://doi.org/10/1046/j.1526-4610.1999.3906432.x</u> (Accessed 28 October 2022).

Dell'Isola, G.B., Tulli, E., Sica, R., Vinti, V., Mencaroni, E., Di Cara, G., Striano, P. and Verrotti, A. 2021. The Vitamin D Role in Preventing Primary Headache in Adult and Pediatric Population. *The Journal of Clinical Medicine*, 10(24): 5983. Available: <u>https://doi.org/10.3390/jcm10245983</u> (Accessed 9 November 2022).

Demir, Y.P. and Sumer, M.M. 2019. Effects of smartphone overuse on headache, sleep and quality of life in migraine patients. *Neurosciences*, 24(2): 115–21. Available: <u>https://doi.org/10.17712/nsj.2019.2.20180037</u> (Accessed 9 August 2022).

Dent, W., Spiss, H.K., Helbok, R., Matuja, W.B.P., Scheunemann, S. and Schmutzhard, E. 2004. Prevalence of migraine in a rural area in South Tanzania: a door-to-door survey. *Cephalalgia*, 24(11): 960–966.

Derry, S., Wiffen, P.J., Moore, R.A. and Bendtsen, L. 2015. Ibuprofen for acute treatment of episodic tension-type headache in adults. *The Cochrane database of systematic reviews*, 7. Available: <u>https://doi.org/10.1002/14651858.CD011474.pub2</u> (Accessed 1 July 2022).

de Vries, B., Anttila, V., Freilinger, T., Wessman, M., Kaunisto, M.A., Kallela, M., Artto, V., Vijfhuizen, L.S., Gobel, H., Dichgans, M., Kubisch, C., Ferrari, M.D., Palotie, A., Terwindt, G.M. and van den Maagdenberg, A.M. 2016. International Headache Genetics Consortium. Systematic re-evaluation of genes from candidate gene association studies in migraine using a large genome-wide association data set. *Cephalalgia*, 36(7): 604–614.

Diener, H.C., Dodick, D., Evers, S., Holle, D., Jense R.H., Lipton, R.B., Porreca, F., Silberstein, S. and Schwedt, T. 2019. Pathophysiology, prevention and treatment of

medication overuse headache. *The Lancet Neurology*, 18(9): 891–902. Available: https://doi.org/10.1016/S1474-4422(19)30146-2 (Accessed 9 November 2022).

DiSabella, M. 2020. Headaches and online learning. *Rise and Shine by Children's national.* Available: <u>https://riseandshine.childrensnational.org/headaches-and-online-learning</u> (Accessed 1 November 2022).

Donnelly, B., 2014. *University lecturers 'suffering high levels of stress'. Herald Scotland*. Available: <u>https://www.heraldscotland.com/news/13184888.university-</u> <u>lecturers-suffering-high-level-stress/</u> (Accessed 22 July 2021).

Dowson, A. 2015. The burden of headache: global and regional prevalence of headache and its impact. *International Journal of Clinical Practice*, 69: 3–7. (Available 20 April 2021).

Drummond, P.D. 2006. Mechanisms of autonomic disturbance in the face during and between attacks of cluster headache. *Cephalalgia: an international journal of headache*, 26(6): 633–641. Available: <u>https://doi.org/10.1111/j.1468-2982.2006.01106.x</u> (Accessed 1 August 2022).

Dueland, A.N. 2015. Headache and alcohol. Headache, 55(7): 1045–1049.

Espí-López, G. V., Bermell-Salvador, C. and Cortés-Amador, S. 2017. Manual therapy as a proposed treatment for chronic migraine. *Journal of Physiotherapy Research*, 1(1): 1–3.

Falavigna, A., Teles, A. R., Velho, M. C., Vedana, V. M., Silva, R. C. d., Mazzocchin, T., Basso, M. and Braga, G. L. D. 2010. Prevalence and impact of headache in undergraduate students in Southern Brazil. *Arq Neuro-Psiquiatr*, 68(6).

Farooq, K. and Williams, P. 2008. Headache and chronic facial pain. *Continuing Education in Anaesthesia Critical Care and Pain*, 8(4): 138–142. Available: <u>https://doi.org/10.1093/bjaceaccp/mkn025</u> (Accessed 12 July 2022).

Featherstone, H.J. and Beitman, B.D. 1984. Marital migraine: A refractory daily headache. *Psychosomatics*, 25(1): 30–31, 35, 38. Available: <u>https://doi.org/10.1016/S0033-3182(84)73094-5</u> (Accessed 28 October 2022).

Ferrante, T., Manzoni, M., Russo, M., Camarda, C., Taga, A., Veronesi, L., Pasquarella, C., Sansebastiano, G. and Torelli, P. 2013. Prevalence of tension-type headache in adult general population: the PACE study and review of the literature. *Neurological Sciences*, 34(1): 137–138.

Finocchi, C. and Sassos, D. 2017. Headache and arterial hypertension. *Neurological sciences: official journal of the Italian Society of Clinical Neurophysiology*, 38(1): 67–72. Available: <a href="https://doi.org/10.1007/s10072-017-2893-x">https://doi.org/10.1007/s10072-017-2893-x</a> (Accessed 9 October 2022)

Fumal, A. and Schoenen, J. 2008. Tension-type headache: current research and clinical management. *The Lancet Neurology*, 7(1): 70–83.

Gardner, J.W., Mountain, G.E. and Hines, E.A. 1940. The relationship of migraine to hypertension headaches. *American Journal of Medical Sciences*, 200(5): 0–53.

Garg, R. 2016. Methodology for research. *Indian Journal of Anaesthesia*, 60(9): 640–645. Available: <u>https://doi.org/10.413/0019-5049.190619</u> (Accessed 20 November 2022).

GBD 2016 Headache Collaborators. 2018. Global, regional, and national burden of migraine and tension-type headache, 1990-2016: a systematic analysis for the Global Burden of Disease Study 2016. *The Lancet. Neurology*, 17(11): 954–976.

GBD 2016 Neurology Collaborators. 2019. Global, regional, and national burden of neurological disorders, 1990-2016: a systematic analysis for the Global Burden of Disease Study 2016. *The Lancet. Neurology*, 18(5): 459–480.

Gil-Gouveia, R. and Martins, I.P. 2002. Headaches associated with refractive errors: Myth or reality? *Headache*, 42(4): 256–262.

Greasley, P. 2008. *Quantitative Data Analysis Using SPSS An Introduction for Health & Social Science*. Open University Press: McGraw-Hill Education.

Gunes, A., Demirci, S., Tok, L., Tok, O., Koyuncuoglu, H. and Yurekli, V.A. 2016. Refractive Errors in Patients with Migraine Headache. *Seminars in ophthalmology*, 31(5): 492–494.

Gurnani, B. and Kaur, K. 2021. Headaches Associated with Face Mask Use During COVID-19 Pandemic-Are We Seeing a Headache Epidemic? *Indian Journal of Otolaryngology and Head and Neck Surgery.* Available: <u>https://doi.org/10.1007/s12070-021-02621-w</u> (Accessed 19 August 2022).

Hagen, K., Stovner, L., Vatten, L., Holmen, J., Zwart, J. and Bovim, G. 2002. Blood pressure and risk of headache: a prospective study of 22 685 adults in Norway. *Journal of Neurology, Neurosurgery* and *Psychiatry*, 72(4): 463–466. Available: <u>https://doi.org/10.1136/jnnp.72.4.463</u> (Accessed 20 October 2022).

Hamelsky, S.W., Lipton, R.B. and Stewart, W.F. 2005. An assessment of the burden of migraine using the willingness to pay model. *Cephalalgia*, 25(2): 87–100.

Han, Q., Lin, Q., Ni, Z. and You, L. 2020. Uncertainties about the transmission routes of 2019 novel coronavirus. *Influenza and other respiratory viruses*, 14(4): 470–471.

Haque, B., Rahman, K.M., Hoque, A., Hasan, A., Chowdhury, R.N. and Uddin, S. 2012. Precipitating and relieving factors of migraine versus tension type headache. *BMC Neurology*, 12(1): 1–4.

Harapan, H., Itoh, N., Yufika, A., Winardi, W., Keam, S., Te, H., Megawati, D., Hayati, Z., Wagner, A. and Mudatsir, M. 2020. Coronavirus disease 2019 (COVID-19): A literature review. Journal of Infection and Public Health, 13(5): 667–673.

Harder, A., Winsvold, B.S., Noordam, R., Vijfhuizen, L.S., Borte, S., Kogelman, L., de Boer, I., Tronvik, E., Rosendaal, F.R., Willems van Dijk, K., O'Connor, E., Fourier, C., Thomas, L.F., Kristoffersen, E.S., Cluster Headache Genetics Working Group., Fronczek, R., Pozo-Risich, P., Jensen, R.H., Ferrari, M.D., Hansen, T.F. and van den Maagdenberg, A. 2021. Genetic Susceptibility Loci in Genomewide Association Study of Cluster Headache. *Annals of neurology*, 90(2): 203: 216.

Headache Classification Committee of the International Headache Society (IHS) 2018. The International Classification of Headache Disorders. *Cephalgia: an international journal of headache*, 38(1): 1–211. Available: <u>https://doi.org/10.1177/0333102417738202</u> (Accessed 10 July 2022).

Heale, R. and Twycross, A. 2015. Validity and reliability in quantitative studies. *Evidence-Based Nursing*, 18(3): 66–67.

Heckman, B.D., Merrill, J.C. and Anderson, T. 2013. Race, psychiatric comorbidity, and headache characteristics in patients in headache subspeciality treatment clinics. *Ethnicity and health*, 18(1): 34–52. Available: <u>https://doi.org/10.1080/13557858.2012.682219</u> (Accessed 2 November 2022).

Ho, K.H. and Benjamin, K.C. 2001. Perceived headache associations in Singapore: results of a randomized national survey. *Headache*, 41(2): 164–170. Available: <u>https://doi.org/10.1046/j.1526-4610.2001.111006164.x</u> (Accessed 12 October 2022).

Houle, M., Lessard, A., Marineau-Belanger, E., Lardon, A., Marchand, A., Descarreaux, M. and Abboud, J. 2021. Factors associated with headache and neck pain among telecommuters – a five days follow-up. *BMC Public Health*, 21: 1086.

IHS—International Headache Society 2004. *Introduction to the classification*. Available: <u>http://ihs-classification.org/en/02\_klassifikation/</u> (Accessed 20 August 2022).

Ihudiebube-Splendor, C. and Chikeme, P. 2020. A descriptive Cross-Sectional Study: *Practical and Feasible Design in Investing Health Care-Seeking Behaviors of Undergraduates*. London: SAGE Publications Limited.

Jairath, N., Hogerney, M. and Parsons, C. 2000. The role of the Pilot Study: A Case Illustration from Cardiac Nursing Research. *Applied Nursing Research*, 13(2): 92–96

Jenkins, B. 2020. Migraine management. Australian Prescriber, 43(5): 148–151.

Jensen, R. 2003. Diagnosis, epidemiology, and impact of tension-type headache. *Current Science Inc*, 7: 455–459. Available: <u>https://doi.org/10.1007/s11916-003-0061-x</u> (Accessed 1 July 2022).

Jensen, R. and Roth, J.M. 2005. Physiotherapy of tension-type headaches. In Olesen, J., Goadsby, P.J., Ramadan, N., Tfelt-Hansen, P. and Welch, K.M. (eds). *The Headaches*. Philadelphia: Lippincott Williams Wilkins, 721–726.

Jensen, R. and Stovner, L. 2008. Epidemiology and comorbidity of headache. *The Lancet Neurology*, 7(4): 354–361.

Johnsen, M. B., Winsvold, B. S., Børte, S., Vie, G. Å., Pedersen, L. M., Storheim, K., Skorpen, F., Hagen, K., Bjørngaard, J. H., Åsvold, B. O., & Zwart, J. A. (2018). The causal role of smoking on the risk of headache. A Mendelian randomization analysis in the HUNT study. *European journal of neurology*, *25*(9), 1148–e102. Available: <u>https://doi.org/10.1111/ene.13675</u> (Accessed 20 September 2022).

Jones, T. L., Baxter, M. A. and Khanduja, V. 2013. A quick guide to survey research. *Annals of the Royal College of Surgeons of England*, 95(1): 5–7. Available: https://doi.org/10.1308/003588413X13511609956372 (Accessed 19 December 2021).

Kandel, S.A. and Mandiga, P. 2022. *Cluster Headache* [online]. Available: <u>https://www.ncbi.nlm.nih.gov/books/NBK544241/# NBK544241 pubdet</u> > (Accessed 19 September 2022).

Kelman, L. and Rains, J.C. 2005. Headache and sleep: examination of sleep patterns and complaints in a large clinical sample of migraineurs. *Headache*, 45(7): 904–10.

Khan, J., Asoom, L., Sunni, A., Rafique, N., Latif, R., Saif, S., Almandil, N., Almohazey, D., AbdulAzeez, S. and Borgio, J. 2021. Genetics, pathophysiology, diagnosis, treatment, management, and prevention of migraine. *Biomedicine and Pharmacotherapy*, 139: 111557. Available: <u>https://doi.org/10.1016/j.biopha.2021.111557</u> (Accessed 9 July 2022).

Kim, H.H. and Jung, J.H. 2021. Social Isolation and Psychological Distress During the COVID-19 Pandemic: A Cross-National Analysis. *The Gerontologist*, 61(1): 103–113. Available: <u>https://doi.org/10.1093/geront/gnaa168</u> (Accessed 20 December 2022).

Kratěnová, J., Žejglicová, K., Malý, M. and Filipová, V. 2007. Prevalence and risk factors of poor posture in school children in the Czech Republic. *The Journal of school health*, 77(3): 131–137. Available: <u>https://doi.org/10.1111/j.1746-1561.2007.00182.x</u> (Available 1 July 2022).

Kristoffersen, E.S., Aaseth, K., Grande, R.B., Lundqvist, C. and Russell, M.B. 2018. Psychological distress, neuroticism and disability associated with secondary chronic headache in the general population- the Akershus study of chronic headache. *The Journal of Headache and Pain*, 19: 62. Available: <u>https://doi.org/10.1186/s10194-018-0894-7</u> (Accessed 22 December 2022).

Kowacs, P.A., Piovesan, E.J., Werneck, L.C., Fameli, H. and Pereira da Silva, H. 2004. Headache related to a specific screen flickering frequency band. *Cephalalgia*, 24(5): 408–410.

Kurt, S. and Kaplan, Y. 2008. Epidemiology and clinical characteristics of headaches in university students. *Clinical Neurology and Neurosurgery*, 110(1): 46–50.

Lampraki, C., Hoffman, A., Roquet. A. and Jopp, D.S. 2022. Loneliness during COVID-19: Development and influencing factors. *PLOS ONE*, 17(3): e0265900. Available: <u>https://doi.org/10.1371/journal.pone.0265900</u> (Accessed 28 December 2022).

Lang, E., Kaltenhäuser, M., Neundörfer, B., & Seidler, S. (2004). Hyperexcitability of the primary somatosensory cortex in migraine — a magnetoencephalographic study. Brain: a journal of neurology, 127(Pt 11), 2459–2469. <u>https://doi.org/10.1093/brain/awh295</u> (Accessed 9 October 2022).

Lee, H, Lee, J., Cho, E., Kim, S. and Y, S. 2019. Efficacy of psychological treatment for headache disorder: a systematic review and meta-analysis. *The Journal of Headache and Pain*, 20(1): 17. Available: <u>https://doi.org/10.1186/s10194-019-0965-4</u> (Accessed 20 July 2022).

Leonardi, M. and Raggi, A. 2019. A narrative review on the burden of migraine: when the burden is the impact on people's life. *The journal of headache and pain*, 20(1): 41. Available: <u>https://doi.org/10.1186/s10194-019-0993-0</u> (Accessed 3 October 2022).

Li, C., Zhang, L., Zhou, J., Fan, Z., Wang, Y., Wang, X., Wang, W. and Yu, S., 2020. Prevalence of primary headache disorders among information technology staff in China: the negative effects of computer use and other correlative factors. *BMC Public Health*, 20(1).

Li, Q., Guan, X., Wu, P., Wang, X., Zhou, L., Tong, Y., Ren, R., Leung, K., Lau, E., Wong, J.Y., Xing, X., Xiang, N., Wu, Y., Li, C., Chen, Q., Li, D., Liu, T., Zhao, J., Liu, M., Tu, W., and Feng, Z. 2020. Early Transmission Dynamics in Wuhan, China, of Novel Coronavirus-Infected Pneumonia. *The New England journal of medicine*, 382(13): 1199–1207.

Lipton, R.B and Bigal, M.E. 2005. Migraine: epidemiology, impact, and risk factors for progression. *The Journal of Head and Face Pain*, 45(1): S3-S13. Available: <u>https://doi.org.10.1111/j.1526-4610.2005.4501001.x</u>. (Accessed 4 July 2022).

Lipton, R. B., & Bigal, M. E. (2007). Ten lessons on the epidemiology of migraine. *Headache*, 47 Suppl 1, S2–S9. Available: <u>https://doi.org/10.1111/j.1526-4610.2007.00671.x</u> (Accessed 7 October 2022).

Lipton, R.B., Bigal, M.E., Diamond, M., Freitag, F., Reed, M.L., Stewart, W.F. and AMPP Advisory Group. 2007. Migraine prevalence, disease burden, and the need for preventative therapy. *Neurology*, 68(5): 343–349. Available: <u>https://doi.org/10.1212/01.wnl.0000252808.97649.21</u> (Accessed 10 October 2022).

Lipton, R.B., Liberman, K., Kolodner, B., Bigal, M.E., Dowson, A. and Stewart, W.F. 2003. Migraine headache disability and health-related quality-of-life: a population-based casecontrol study from England. *Cephalalgia*, 23(6): 441–450.

Loder, S., Sheikh, H.U. and Loder, E. 2015. The prevalence, burden, and treatment of severe, frequent, and migraine headaches in US minority populations: statistics from National Survey studies. *Headache*; 55(2): 214–228.

Lopez-Mesonero, L., Marquez, S., Parra, P., Gomez-Leyva, G., Munaz, P. and Pascual, J. 2009. Smoking as a precipitating factor for migraine: a survey in medical students. *The Journal of Headache and Pain*, 10: 101–103.

Lucas, S., Hoffman, J.M., Bell, K.R. and Dikmen, S. 2014. A prospective study of prevalence and characterization of headache following mild traumatic brain injury. *Cephalalgia: an international journal of headache*, 32(2): 93–102.

Lyngberg, A.C., Rasmussen, B.K., Jorgensen, T. and Jensen, R. 2005. Has the prevalence of migraine and tension-type headache changed over a 12-year period? A Danish population survey. *European journal of epidemiology*, 20(3): 243–249. Available: <u>https://doi.org/10.1007/s10654-004-6519-2</u> (Accessed 2 July 2022).

Maboe, K. 2019. Students' Support in an ODel Context. *Modern Technologies for Teaching and Learning in Socio-Humanitarian Disciplines*, 114–137. Available: <a href="https://doi.org/10.4018/978-1-5225-7841-3.ch006">https://doi.org/10.4018/978-1-5225-7841-3.ch006</a> (Accessed 12 September 2022).

Magnavita, N. 2014. Health surveillance of workers in indoor environments. Application of the Italian version of the MM040/IAQ questionnaire. *La Medicina del lavoro*, 105(3): 174–186.

Magnavita, N. 2015. Work-related symptoms in indoor environments: a puzzling problem for the occupational physician. *International archives of occupational and environmental health*, 88(2): 185–196.

Magnavita, N. 2022. Headache in the Workplace: Analysis of Factors Influencing Headaches in Terms of productivity and Health. *International journal of environmental research and public health*, 19(6): 3712.

Mahmoud, A.N., Mentias, A., Elgendy, A.Y., Qazi, A., Barakat, A.F., Saad, M., Mohsen, A., Abuzaid, A., Mansoor, H., Mojadidi, M.K. and Elgendy, I. 2018. Migraine and the risk of cardiovascular and cerebrovascular events: A meta-analysis of 16 cohort studies including 1,152,407 subjects. *BMJ Open*, 8: e020498.

Malmberg-Ceder, K., Haanpaa, M., Korhonen, P., Kautiainen, H., Veromaa, V. and Soinila,

S. 2019. The role of psychological risk factors in the burden of headache. *Journal of pain research*,12: 1733–1741.

Manzoni, G. C. and Torelli, P. 2008. Migraine with and without aura: a single entity? *Neurological Sciences*, 29(1): 40–43

Mao, L., Jin, H., Wang, M., Hu, Y., Chen, S., He, Q., Chang, J., Hong, C., Zhou, Y., Wang, D., Miao, X., Li, Y. and Hu, B. 2020. Neurologic Manifestations of Hospitalized Patients With Coronavirus Disease 2019 in Wuhan, China. *JAMA neurology*, 77(6): 683–690. Available: <u>https://doi.org/10.1001/jamaneurol.2020.1127</u> (Accessed 4 November 2022).

Marlowe, N. 1998. Self-efficacy moderates the impact of stressful events on headache. *Headache*, 38(9): 662–667. Available: <u>https://doi.org/10.1046/j.1526-4610.1998.3809662.x</u> (Accessed 21 October 2022)>

Martami, F., Ghorbani, Z., Abolhasani, M., Togha, M., Meysamie, A., Sharifi, A. and Jahromi, S.R. 2018. Comorbidity of gastrointestinal disorders, migraine, and tension-type headache: a cross-sectional study in Iran. *Neurological Sciences*, 39(1): 63–70.

Mashige, K. P. 2014. Computer-related symptoms in the workplace: causes and preventative stratagies. *Occupational Health Southern African*, 20 (3): 13–17.

Mateen, F.J., Dua, T., Steiner, T. and Saxena, S. 2008. Headache disorders in developing countries; research over the past decade. *Cephalalgia*, 28(11): 1107–1114.

Maurya, A., Qureshi, S., Jadia, S. and Maurya, M. 2019. "Sinus Headache": Diagnosis and Dilemma?? An Analytical and Prospective Study. *Indian Journal of Otolaryngology and Head and Neck Surgery*, 71(3): 367–370. Available: <u>https://doi.org/10.1007/s12070-019-01603-3</u> (Accessed 20 October 2022).

Mayans, L. and Walling, A. 2018. Acute Migraine Headache: Treatment Strategies. *American Family Physician*, 97(4): 243–251.

MayoClinic.2015.Headachecauses.Available:http://www.mayoclinic.org/symptoms/headache/basics/causes/sym-20050800(Accessed20 July 2022).

Mayo Clinic. 2022. *Migraine symptoms and causes*. Available: <u>https://www.mayoclinic.org/diseases-conditions/migraine-headache/symptoms-causes/syc-20360201</u> (Accessed 20 August 2022).

Mayo Clinic. 2021. *Tension Headache*. Available: <u>https://www.mayoclinic.org/diseases-conditions/tension-headache/symptoms-causes/syc-20353977</u> (Accessed 3 January 2022).

Mengistu, G. and Alemayehu, S. 2013. Prevalence and burden of primary headache disorders among a local community in Addis Ababa, Ethiopia. *The Journal of Headache and Pain*, 14(30).

Mennini, F. S., Gitto, L. and Martelletti, P. 2008. Improving care through health economics analyses: cost of illness and headache. *The Journal of Headache and Pain*, 9(4): 199–206 (Accessed 8 June 2021).

Middeke, M., Lemmer, B., Schaaf, B. Eckes, L. 2008. Prevalence of hypertension-attributed symptoms in routine clinical practice: a general practitioners-based study. *Journal of human hypertension*, 22(4): 252–258. Available: <u>https://doi.org/10.1038/sj.jhh.1002305</u> (Accessed 20 October 2022).

Middleton, F. 2019. *The 4 types of validity | Explained with Easy Examples*. Available: <u>https://www.scribbr.com/methodology/types-of-validity/ (Accessed 25 July 2021)</u>.

Min, J.Y., Park, S.G., Kim, S.S. and Min, K.B. 2014. Workplace injustice and self-reported disease and absenteeism in South Korea. *American journal of industrial medicine*, 57(1): 87–96.

Molarius, A., Tegelberg, A. and Ohrvik, J. 2008. Socio-Economic Factors, Lifestyle and Headache Disorders — A Population-Based Study in Sweden. *Headache journal*, 48(10): 1426–1437. Available: <u>https://doi.org/10.1111/j.1526-4610.2008.01178.x</u> (Accessed 16 October 2022).

Momayyezi, M., Fallahzadeh, H. and Momayyezi, M. 2014. Prevalence of Migraine and Tension-Type Headache in Yazd, Iran. *Zahedan Journal of Research in Medical Sciences*, 17(4). Available: <u>https://doi.org/10.17795/zjrms966</u> (Accessed 1 November 2022).

Monzani, L., Zurriaga, R. and Espi Lopez, G.V. 2018. Anxiety and severity of Tension-Type Headache mediate the relation between headache presenteeism and workers' productivity. *PLOS ONE*, 13(7): e0201189.

Moonasar, D., Pillay, A., Leonard, E., Naidoo, R., Mngemane, S., Ramkrishna, W., Jamaloodien, K., Lebese, L., Chetty, K., Bamford, L., Tanna, G., Ntuli, N., Mlisana, K., Madikizela, L., Modisenyane, M., Engelbrecht, C., Maja, P., Bongweni, F., Furumele, T., Mayet, N., Goga, A., Talisuna, A., Ramadan, O. and Pillay, Y. 2021. COVID-19: lessons and experiences from South Africa's first surge. *BMJ Global Health*, 6(2). Available: https://gh.bmj.com/content/6/2/e004393 (Accessed 20 July 2022).

Mpungose, C.B. 2020. Emergent transition from face-to-face to online learning in a South African University in the context of the Coronavirus pandemic. *Humanities and Social Sciences Communications*, 7(1). Available: <u>https://doi.org/10.1057/s41599-020-00603-x</u> (Accessed 8 January 2022).

Murungi, C. and Gitonga, R. 2015. Active Learning with Technology Tools in the Blended/Hybrid Classes. *Handbook of Research on Educational Technology Integration and Active Learning*, 346–357. Available: <u>https://doi.org/10.4018/978-1-4666-8363-1.ch017</u> (Accessed 10 August 2022).

Mutiawati, E., Syahrul, S., Fahriani, M., Fajar, J. K., Mamada, S. S., Maliga, H. A., Samsu, N., Ilmawan, M., Purnamasari, Y., Asmiragani, A. A., Ichsan, I., Emran, T. B., Rabaan, A. A., Masyeni, S., Nainu, F., & Harapan, H. 2020. Global prevalence and pathogenesis of headache in COVID-19: A systematic review and meta-analysis. *F1000Research*, 9: 1316. Available: <u>https://doi.org/10.12688/f1000research.27334.2</u> (Accessed 16 August 2022).

Nagalakhmi, L.R. and Trivedi, M.J. 2010. Utilization of online survey tools for academic research: A practical approach to survey monkey. *The International Journal Research Publication's*, 4(3): 21–28.

Nash, J.M. and Thebarge, R.W. 2006. Understanding psychological stress, its biological processes, and impact on primary headache. *Headache*, 46(9): 1377–1386. Available: <u>https://doi.org/10.1111/j.1526-4610.2006.00580.x</u> (Accessed 20 October 2022).

National Institute of Neurological Disorders and Stroke. 2016. *Headache: Hope Through Research.* Available: <u>https://www.ninnds.nih.gov/disorders/patient-caregiver-education/hope-through-research/headache-hope-through-research</u> (Accessed 20 April 2021).

Nicholson, R.A., Houle, T.T., Rhudy, J.L. and Norton, P.J. 2007. Psychological Risk Factors in Headache. *Headache*, 47(3): 413–426. Available: <u>https://doi.org/10.1111/j.1526-4610.2006.00716.x</u> (Accessed 18 October 2022).

Nordhaug, L. H., Hagen, K., Vik, A., Stovner, L. J., Follestad, T., Pedersen, T., Gravdahl, G. B., & Linde, M. (2018). Headache following head injury: a population-based longitudinal cohort study (HUNT). *The journal of headache and pain*, 19(1), 8. Available: <u>https://doi.org/10.1186/s10194-018-0838-2</u> (Accessed 9 November 2022).

Mier, R.W. and Dhadwal, S. 2018. Primary headaches. *Dental clinics of North America*, 62(4): 611–628. Available: <u>https://doi.org/10.1016/j.cden.2018.06.006</u> (Accessed 1 September 2022).

Noseda, R. and Burstein, R. 2013. Migraine pathophysiology: anatomy of the trigeminovascular pathway and associated neurological symptoms, cortical spreading depression, sensitization, and modulation of pain. *Pain*, *154(1):* S44–S53.

O'Connor, E., Fourier, C., Ran, C., Sivakumar, P., Liesecke, F., Southgate, L., Harder, A., Vijfhuizen, L.S. Yip, J., Giffin, N., Silver, N., Ahmed, F., Hostettler, I.C., Davies, B., Cader, M.Z., Simpson, B.S., Sullivan, R., Efthymiou, S., Adebimpe, J., Quinn, O., Campbell, C., Cavalleri, G.L., Vikelis, M., Kelderman, T., Paemeleire, K., Kilbride, E., Grangeon, L.,

Lagrata, S., Danno, D., Trembath, R., Wood, N.W., Kockum, I., Winsvold, B.S., Steinberg, A., Sjostrand, C., Waldenlind, E., Vandrovcova, J., Houlden, H., Matharu, M. and Belin, A.C. 2021. Genome-Wide Association Study Identifies Risk Loci for Cluster Headache. *Annals of Neurology*, 90(2): 193–202.

Oshinaike, O., Ojo, O., Okubadejo, N., Ojelabi, O., Dada, A. 2014. Primary Headache Disorders at a Tertiary Health Facility in Lagos, Nigeria: Prevalence and Consultation Patterns. *BioMed Research International.* Available: <u>https://doi.org/10.1155/2014/782915</u> (Accessed 16 September 2022).

Osuntokun, B.O., Adeuja, A.O., Nottidge, V.A., Bademosi, O., Alumide, A.O., Ige, O., Yaria, F., Schoenberg, B.S. and Bolis, C.L. 1992. Prevalence of headache and migrainous headache in Nigerian Africans: a community-based study. *East African medical journal*, 69(4): 196–199.

Panconesi, A. 2016. Alcohol-induced headaches: Evidence for a central mechanism? *Journal of Neurosciences in Rural Practice*, 7(2): 269–275.

Patel, M. and Patel, N. 2019. Exploring Research Methodology: Review Article. *International Journal of Research and Review*, 6(3): 48–55.

Penzien, D.B., Rains, J.C. and Andrasik, F. 2002. Behavioral management of recurrent headache: three decades of experience and empiricism. *Applied psychophysiology and biofeedback*, 27(2): 163–181. Available: <u>https://doi.org/10.1023/a:1016247811416</u> (Accessed 9 July 2022).

Perini, F., D'Andrea, G., Galloni, E., Pignatelli, F., Billo, G., Alba, S., Bussone, G. and Toso. 2005. Plasma cytokine levels in migraineurs and controls. *Headache*, 45(7): 926–931.

Peroutka S. J. 2014. What turns on a migraine? A systematic review of migraine precipitating factors. *Current pain and headache reports*, 18(10): 454. Accessed: <u>https://doi.org/10.1007/s11916-014-0454-z</u> (Accessed 5 October 2022).

Petersen, A.S., Lund, N., Snoer, A., Jensen, R.H. and Barloese, M. 2022. The economic and personal burden of cluster headache: a controlled cross-sectional study. *The Journal of Headache and Pain*, 23(1): 58. Available: <u>https://doi.org/10.1186/s10194-022-01427-7</u> (Accessed 9 September 2022).

Planchuelo-Gomez, A., Trigo, J., de Luis-Garcia, R., Guerrero, A.L., Porta-Etessam, J. and Garcia-Azorin, D. Deep phenotyping of headache in hospitalized COVID-19 patients via principal component analysis. *Frontiers in Neurology*, 11: 583870. Available: <u>https://doi.org/10.3389/fneur.2020.583870</u> (Accessed 9 August 2022).

Prakash, S., Rathore, C., Makwana, P., Dave, A., Joshi, H. and Parekh, H. 2017. Vitamin D Deficiency in Patients With Chronic Tension-Type Headache: A Case-Control Study. *Headache*, 57(7): 1096-1108. Available: <u>https://doi.org/10.1111/head.13096</u> (Accessed 1 July 2022).

Rab, S., Javaid, M. and Haleem, A. 2020. Face masks are new normal after COVID-19 pandemic. *Diabetes and Metabolic Syndrome: Clinical Research*, 14(6): 1617–1619.

Radtke, A. 2009. Prevalence and Burden of Headache and Migraine in Germany. *Headache*, 49: 79–89.

Ramirez-Moreno, J., Ceberino, D., Plata, A.G., Rebollo, B., Sedas, P.M., Hariramani, R., Roa, A.M. and Constantino, A. 2020. *Mask-associated 'de novo' headache in healthcare workers during the COVID-19 pandemic*, 78(8).

Rasmussen B. K. (1993). Migraine and tension-type headache in a general population: precipitating factors, female hormones, sleep pattern and relation to lifestyle. *Pain*, 53(1), 65–72. Available: <u>https://doi.org/10.1016/0304-3959(93)90057-V</u> (Accessed 9 October 2022).

Raucci, U., Boni, A., Evangelisti, M., Vecchia, N.D., Velardi, M., Ursitti, F., Terrin, G., Nardo, G.D., Reale, A., Villani, A. and Parisi, P. 2020. Lifestyle Modifications to Help Prevent Headache at a Developmental Age. *Frontiers in Neurology*, 11: 618357. Available: <u>https://doi.org/10.3389/fneur.2020.618375</u> (Accessed 10 July 2022).

Rizzoli, P. and Mullaly, W.J. 2017. Headache. *The American Journal of Medicine*, 131(1): 17–24. Available: <u>https://doi.org/10.1016/j.amjmed.2017.09.005</u> (Accessed 28 October 2022).

Robbins, M.S. and Lipton, R.B. 2010. The epidemiology of primary headache disorders. *Seminars in Neurology*, 30(2): 107–119. Available: <u>https://doi.org/10.1055/s-0030-1249220</u> (Accessed 12 October 2022).

Robbins, M.S. 2021. Diagnosis and Management of Headache: A Review. *JAMA*, 325(18): 1874–1885. Available: <u>https://doi.org/10.1001/jama.2021.1640</u> (Accessed 29 October 2022).

RSA. 2013. White paper for post-school education and training: Building an expanded, effective and integrated post-school system. Department of Higher Education and Training. Available: <u>https://www.dhet.gov.za/</u> (Accessed 29 October 2022).

Ruschel, M. and Jesus, O. 2022. *Migraine Headache*. StatPearls Publishing. Available: <u>https://pubmed.ncbi.nlm.nih.gov/32809622/</u> (Accessed 9 August 2022).

Ranasinghe, P., Wathurapatha, W.S., Perera, Y.S., Lamabadusuriya, D.A., Kulatunga, S., Jayawardana, N. and Katulanda, P. 2016. Computer vision syndrome among computer office workers in a developing country: an evaluation of prevalence and risk factors. *BMC Research Notes*, 9(1): 1–9.

Rosner, E. 2020. Adverse effects of prolonged mask use among healthcare professionals during COVID-19. *Journal of Infectious Diseases and Epidemiology*, 6: 130.

Rubin, D.B., Danish, H.H., Ali, A.B., Li, K., LaRose, S., Monk, A.D., Cote, D.J., Spendley, L., Kim, A.H., Robertson, M.S., Torre, M., Smith, T.R., Izzy, S., Jacobson, C.A., Lee, J.W. and Vaitkevicius, H. 2019. Neurological toxicities associated with chimeric antigen receptor T-cell therapy. *Brain: a journal of neurology*, 142(5): 1334–1348.

Rudra, A., Chatterjee, S. and Ray, S. 2015. Common primary headaches in geriatrics. *Indian Journal of Pain*, 29(1): 15–20.

Ruiz, M., Pedraza, M.I., de la Cruz, C., Barón, J., Muñoz, I., Rodríguez, C., Celorrio, M., Mulero, P., Herrero, S. and Guerrero, A.L. 2014. Headache in the elderly: a series of 262 patients. *Neurología*, 29(6): 321–326.

Russell, M.B., Saltyte-Benth, J. and Levi, N. 2006. Are infrequent episodic, frequent episodic and chronic tension-type headache inherited? A population-based study of 11 199 twin pairs. *The Journal of Headache and Pain*, 7: 119–126. Available: <u>https://doi.org/10.1007/s10194-006-0299-x</u> (Accessed 20 July 2022).

Safiri, S., Pourfathi, H., Eagan, A., Mansournia, M.A., Khodayari, M.T., Sullman, M.J.M., Kaufman, J., Collins, G., Dai, H., Bragazzi, N.L. and Kolahi, A.A. 2022. Global, regional and national burden of migraine in 204 countries and territories, 1990–2019. *Pain*, 163(2): e293-e309. Available: <u>https://www.doi.org/10.1097/j.pain.00000000002275</u> (Accessed 19 September 2022).

Scalise, A., Desiato, M. T., Gigli, G. L., Romigi, A., Tombini, M., Marciani, M. G., Izzi, F. and Placidi, F. 2006. Increasing cortical excitability: a possible explanation for the proconvulsant role of sleep deprivation. *Sleep*, 29(12), 1595–1598. Available: <u>https://doi.org/10.1093/sleep/29.12.1595</u> (Accessed 9 October 2022).

Scher, A.I., Terwindt, G.M., Picavet, H.S., Vershuren, W.M., Ferrari, M.D. and Launer, L.J.2005. Cardiovascular risk factors and migraine: the GEM population-based study.*Neurology*,64(4):614–620.Available:https://doi.org/10.1212/01.WNL.0000151857.43225.49(Accessed 10 October 2022).

Schramm, S., Moebus, S., Lehmann, N., Galli, U., Obermann, M., Bock, E., Yoon, M., Diener, H. and Katsarava, Z. 2014. The association between stress and headache: A longitudional population-based study. *Cephalalgia*, 35(10): 853–863. Available: <u>https://doi.org/10.1177/0333102414563087</u> (Accessed 20 October 2022).

88

Shah, N. and Hameed, S. 2021. *Muscle Contraction Tension Headache*. StatPearls Publishing [online]. Available: <u>https://www.ncbi.nlm.nih.gov/books/NBK562274/</u> (Accessed 20 July 2022).

Shah, B. and Pandey, D.R. 2017. Migraine Review Article. European Journal of BiomedicalAndPharmaceuticalSciences,4(4):226–230.Available:https://www.researchgate.net/publication/315729535(Accessed 4 June 2022).

Sharma, M. and Menon, B. 2022. Headache Incidence and Characteristics in COVID-19 Patients: A Hospital-Based Study. *Annals of Indian Academy of Neurology*, *2*5(1): 88–91. Available: <u>https://doi.org/10.4103/aian.aian\_244\_21</u> (Accessed 20 November 2022).

Shereen, M.A., Khan, S., Kazmi, A., Bashir. and Siddique, R. 2020. COVID-19 infection: Emergence, transmission, and characteristics of human coronaviruses. *Journal of Advanced Research*, 24: 91–98.

Shimizu, T., Sakai, F., Miyake, H., Sone, T., Sato, M., Tanabe, S., Azuma, Y. and Dodick, D. 2021. Disability, quality of life, productivity impairment and employer costs of migraine in the workplace. *The Journal of Headache and Pain*, 22(1).

Shuaibi, S., AlAshqar, A., Ahmed, S.F., Alroughani, R., AlThufairi, H., Owayed, S., AlHamdan, F. and Al-Hashel, J. 2021. Primary Headache Disorder Among School Students in Kuwait. *Frontiers in Neurology*. Available: <u>https://doi.org/10.3389/fneur.2021.621017</u> (Accessed 28 October 2022).

Simić, S., Rabi-Žikić, T., Villar, J.R., Calvo-Rolle, J.L., Simić, D. and Simić, S.D. 2020. Impact of Individual Headache Types on the Work and Work Efficiency of Headache Sufferers. *International Journal of Environmental Research and Public Health*, 17(18): 6918. Available: <u>https://doi.org/10.3390/ijerph17186918</u> (Accessed 20 October 2022). Sinclair, A.J., Sturrock, A., Davies, B. and Matharu, M. 2015. Headache management: pharmacological approaches. *Practical Neurology*, 15(6): 411–423. Available: <u>https://pn.bmj.com/content/15/6/411</u> (Accessed 1 August 2022).

Solomon, S. 2009. Post-traumatic headache: commentary: an overview. *Headache*, 49(7): 1112–1115.

Song, T.J., Cho, S.J., Kim, W.J., Yang, K.I., Yun, C.H. and Chu, M.K. 2018. Poor sleep quality in migraine and probable migraine: a population study. *The journal of headache and pain*, 19(1): 58. Available: <u>https://doi.org/10.1186/s10194-018-0887-6</u> (Accessed 20 October 2022).

Souto, I., Pereira, A., Brito, E., Sancho, L. and Jardim, J. 2018. Psychosocial Risk Factors and Distress in Higher Education Teachers. *The European Proceedings of Social and Behavioural Sciences*, 40: 127–140. Available: <u>https://doi.org/10.15405/epsbs.2018.11.14</u> (Accessed 28 November 2021).

Stalin, P., Abraham, S.B., Kanimozhy, K., Prasad, R.V., Singh, Z. and Purty, A.J. 2016. Mobile Phone Usage and its Health Effects Among Adults in a Semi-Urban Area of Southern India. *Journal of clinical and diagnostic research*, 10(1): LC14-LC16.

Statistics South Africa. 2022. South Africa's youth continues to bear the burden of unemployment. Available: <a href="https://www.statssa.gov.za/?p=15407#:~:text=Q1%3A2022%20saw%20the%20total,rate">https://www.statssa.gov.za/?p=15407#:~:text=Q1%3A2022%20saw%20the%20total,rate</a> <a href="https://www.statssa.gov.za/?p=15407#:~:text=Q1%3A2022%20saw%20the%20total,rate">https://www.statssa.gov.za/?p=15407#:~:text=Q1%3A2022%20saw%20the%20total,rate</a> <a href="https://www.statssa.gov.za/?p=15407#:~:text=Q1%3A2022%20saw%20the%20total,rate">https://www.statssa.gov.za/?p=15407#:~:text=Q1%3A2022%20saw%20the%20total,rate</a> <a href="https://www.statssa.gov.za/?p=15407#:~:text=Q1%3A2022%20saw%20the%20total">https://www.statssa.gov.za/?p=15407#:~:text=Q1%3A2022%20saw%20the%20total,rate</a> <a href="https://www.statssa.gov.za/?p=15407#:~:text=Q1%3A2022%20saw%20the%20total">https://www.statssa.gov.za/?p=15407#:~:text=Q1%3A2022%20saw%20the%20total,rate</a> <a href="https://www.statssa.gov.za/?p=15407#:~:text=Q1%3A2022%20saw%20the%20total">https://www.statssa.gov.za/?p=15407#:~:text=Q1%3A2022%20saw%20the%20total</a>, <a href="https://www.statssa.gov.za/?p=15407#:~:text=Q1%3A2022%20saw%20the%20total">https://www.statssa.gov.za/?p=15407#:~:text=Q1%3A2022%20saw%20the%20total</a>, <a href="https://www.statssa.gov.za/?p=15407#:~:text=Q1%3A2022%20saw%20the%20total">https://www.statssa.gov.za/?p=15407#:~:text=Q1%3A2022%20saw%20the%20total</a>, <a href="https://www.statssawwd.gov.za/?p=15407#">https://www.statssawwd.gov.za/?p=15407#">https://www.statssawwd.gov.za/?p=15407#</a>, <a href="https://www.statssawwd.gov.za/?p=15407#">https://www.statssawwd.gov.za/?p=15407#</a>, <a href="https://www.statssawwd.gov.za/?p=15407#">https://www.statssawwd.gov.za/?p=15407#</a>, <a href="https://www.statswawwd.gov.za/?p=15407#">https://www.statswawwd.gov.za/?p=15407#</a>, <a href="https://www.statswawwd.gov.za/?p=15407#">https://www.statswawwd.gov.za/?p=15407#</a>, <a href="https://www.statswawwd.gov.za/?p=15407#">https://www.statswawwd.gov.za/?p

Steiner, T.J., Gururaj, G., Andrée, C., Katsarava, Z., Ayzenberg, I., Yu, S., Al Jumah, M., Tekle-Haimanot, R., Birbeck, G., Herekar, A., Linde, M., Mbewe, E., Manandhar, K., Risal, A., Jensen, R., Queiroz, L., Scher, A., Wang, S. and Stovner, L. 2014. Diagnosis, prevalence estimation and burden measurement in population surveys of headache: presenting the HARDSHIP questionnaire. *The Journal of Headache and Pain*, 15(1). Available: <u>https://doi.org/10.1186/1129-2377-15-3</u> (Accessed 10 January 2022).

Steiner, T.J., Stovner, L.J., Dua, T., Birbeck, G.L., Jensen, R., Katsarava, Z., Martelletti, P. and Saxena, S. 2011. Time to act on headache disorders. *The Journal of Headache and Pain*, 12: 501-503. Available: <u>https://doi.org/10.1007/s10194-011-0368-7</u> (Accessed 27 October 2022).

Steiner, T.J., Stovner, L.J., Vos, T., Jensen, R. and Katsarava, Z. 2018. Migraine is first cause of disability in under 50s: will health politicians now take notice? *The Journal of Headache and Pain*, 19(1): 17. Available: <u>https://doi.org/10.1186/s10194-018-0846-2</u> (Available 20 October 2022).

Stovner, L.J, Hagen, K., Jensen, R., Katsarava, Z., Lipton, R., Scher, A., Steiner, T. and Zwart, J. 2007. The global burden of headache: a documentation of headache prevalence and disability worldwide. *Cephalalgia: An International Journal of Headache*, 27(3): 193–210. (Accessed 17 January 2021).

Stovner, L.J., Hagen, K., Linde, M. and Steiner, T. J. 2022. The global prevalence of headache: an update, with analysis of the influences of methodological factors on prevalence estimates. *The journal of headache and pain*, *23*(1), 34. Available: <u>https://doi.org/10.1186/s10194-022-01402-2</u> (Accessed 5 September 2022).

Straube, A. and Andreou, A. 2019. Primary headaches during lifespan. *The Journal of Headache and Pain,* 20 (35). Available: <u>https://doi.org/10.1186/s10194-019-0985-0</u> (Accessed 28 October 2022).

Swanson, S.A., Zeng, Y., Weeks, M. and Colman, I. 2013. The contribution of stress to the comorbidity of migraine and major depression: results from a prospective cohort study. *BMJ Open*, 3(3): e002057.

Takeshima, T. and Kikui, S. 2013. Headache disorders. *Brain and Nerve: Shinkei kenkyu no Shinpo*, 65(9): 1043–1055 (Accessed 18 June 2021).

Taylor, H.O., Taylor, R.J., Nguyen, A.W. and Chatters, L. 2018. Social Isolation, Depression, and Psychological Distress Among Older Adults. *Journal of aging and health*, 30(2): 229–246. Available: <u>https://doi.org/10.1177/089826431667351</u> (Accessed 22 December 2022).

Tekle Haimanot, R., Seraw, B., Forsgren, L., Ekbom, K. and Ekstedt, J. 1995. Migraine, chronic tension-type headache, and cluster headache in an Ethiopian rural community. *Cephalalgia*, 15(6): 482–488.

Tewari, D. and Ilesanmi, K. 2020. Teaching and learning interaction in South Africa's higher education: Some weak links. *Cogent Social Sciences*, 6(1). Available: <u>https://doi.org/10.1080/23311886.2020.1740519</u> (Accessed 10 January 2022).

Theeler, B., Lucas, S. Riechers, R. 2nd. and Ruff, R.L. 2013. Post-traumatic headaches in civilians and military personnel: a comparative, clinical review. *Headache*, 53(6), 881–900.

Thorud, H., Aurjord, R.and Falkenberg, H. 2021. Headache and musculoskeletal pain in school children are associated with uncorrected vision problems and need for glasses: a case-control study. *Scientific Reports*, 11: 2093.

Togha, M., Hashemi, S.M., Yamani, N., Martami, F. and Salami, Z. 2022. A Review on Headaches Due to COVID-19 Infection. *Frontiers in Neurology*, 13: 942956.

Tostmann, A., Bradley, J., Bousema, T., Yiek, W. K., Holwerda, M., Bleeker-Rovers, C., Ten Oever, J., Meijer, C., Rahamat-Langendoen, J., Hopman, J., van der Geest-Blankert, N., & Wertheim, H. 2020. Strong associations and moderate predictive value of early symptoms for SARS-CoV-2 test positivity among healthcare workers, the Netherlands, March 2020. *Euro surveillance: bulletin Europeen sur les maladies transmissibles = European communicable disease bulletin*, 25(16): 2000508. Available: https://doi.org/10.2807/1560-7917.ES.2020.25.16.2000508 (Accessed 19 September 2022). Uygun, Ö., Ertaş, M., Ekizoğlu, E., Bolay, H., Özge, A., Kocasoy Orhan, E., Çağatay, A. A. and Baykan, B. 2020. Headache characteristics in COVID-19 pandemic — a survey study. *The journal of headache and pain*, *21*(1), 121.

Van Deursen, A.J. and Van Dijk, J.A. 2019. The first-level digital divide shifts from inequalities in physical access to inequalities in material access. *New media and society*, 21(2): 354–375.

van Tillburg, T.G., Steinmetz, S., Stolte, E., van der Roest, H. and de Vries, D.H. 2021. Loneliness and Mental Health During the COVID-19 Pandemic: A Study Among Dutch Older Adults. *The journals of gerontology. Series B, Psychological sciences and social sciences*, 76(7): e249–e255. Available: <u>https://doi.org/10.1093/geronb/gbaa111</u> (Accessed 20 October 2022).

Varkey, B. 2021. Principles of clinical ethics and their application to practice. *Medical principles and practice*, 30: 17–28. Available: <u>https://doi.org/10.1159/000509119</u> (Accessed 6 February 2022).

Varkey, E., Hagen, K., Zwart, J. A. and Linde, M. 2008. Physical activity and headache: results from the Nord-Trøndelag Health Study (HUNT). *Cephalalgia: an international journal of headache*, 28(12): 1292–1297. Available: <u>https://doi.org/10.1111/j.1468-2982.2008.01678.x</u> (Accessed 6 October 2022).

Vetvik, K.G. and MacGregor, E.A. 2017. Sex differences in the epidemiology, clinical features, and pathophysiology of migraine. *Lancet Neurology*, 16(1): 76–87.

Vicente Herrero, M.T., Ramirez Iniguez de la Torre, M.V., Reinoso Barbero, L. and Ruiz de la Torre, E. 2021. Preventative Aspects for Migraine and the Workplace: A European Study, *Archivos de Prevencion de Riesgos Laborales*, 24(1): 20–33.
Villar-Martinez, M.D., Puledda, F. Goadsby, P.J. 2020. Recent Advances in the Management of Cluster Headache. *Current Treatment Options in Neurology*, 22(12). Available: <u>https://doi.org/10.1007/s11940-020-00655-z</u> (Accessed 28 October 2022).

Volans, G.N. and Castleden, C.M. 1976. The relationship between smoking and migraine. *Postgraduate Medical Journal*, 52(604): 80–82.

Vosoughi, K., Stovner, L.J., Steiner, T.J., Moradi-Lakeh, M., Fereshtehnejad, S., Farzadfar, F., Heydarpour, P., Malekzadeh, R., Naghavi, M., Sahraian, M.A., Sepanlou, S.G., Tehrani-Banihashemi, A., Majdzadeh, R., Feigin, V.L., Vos, T., Mokdad, A.H. and Murray, C.J. 2019. The burden of headache disorders in the Eastern Mediterranean Region, 1990–2016: findings from the Global Burden of Disease study 2016. *The Journal of Headache and Pain*, 20(40). Available: <u>https://doi.org/10.1186/s10194-019-0990-3</u> (Accessed 19 October 2022)

Wagner, R. and Myers, R.R. 1996. Endoneurial injection of TNF-alpha produces neuropathic pain behaviors. *Neuroreport*, 7(18): 2897–2901.

Walters, A.B., Hamer, J.D. and Smitherman, T.A. 2014. Sleep disturbance and affective comorbidity among episodic migraineurs. *Headache*, 54(1): 116–124.

Wang, J., Su, H., Xie, W. and Yu, S. 2017. Mobile Phone Use and The Risk of Headache: A Systematic Review and Meta-analysis of Cross-sectional Studies. *Scientific Reports*, 7: 12595. Available: <u>https://doi.org/10.1038/s41598-017-12802-9</u> (Accessed 20 September 2022).

Wang, S.J. 2003. Epidemiology of migraine and other types of headache in Asia. *Current neurology and neuroscience reports*, 3(2): 104–108. Available: <u>https://doi.org/10.1007/s11910-003-0060-7</u> (Accessed 8 November 2022).

Wang, Y.F. and Wang, S.J. 2021. Hypertension and migraine: Time to revisit the evidence. *Current Pain and Headache Reports*, 25(9): 1–9.

Waung, M.W., Taylor, A., Qualmann, K.J. and Burish, M.J. 2020. Family History of Cluster Headache: A Systematic Review. *JAMA neurology*, 77(7); 887–896.

Wei, D.Y., Yuan Ong, J.J. Goadsby, P.J. 2018. Cluster Headache: Epidemiology, Pathophysiology, Clinical Features, and Diagnosis. *Annals of Indian Academy of Neurology*, 21(1): S3–S8.

Winter, A.C., Hoffmann, W., Meisinger, C., Evers, S., Vennemann, M., Pfaffenrath, V., Fendrich, K., Baumeister, S.E., Kurth, T. and Berger, K. 2011. Association between lifestyle factors and headache. *The Journal of Headache and Pain*, 12: 147–155. Available: <u>https://doi.org/10.1007/s10194-010-0286-0</u> (Accessed 28 October 2022).

World Health Organization (WHO). 2016. *Headache disorders*. Available: <u>https://www.int/news-room/fact-sheets/detail/headache-disorders</u> (Accessed 20 June 2021).

Wu, T., Shuntong, K., Peng, W., Zuo, C., Zhu, Y., Pan, L., Fu, K., You, Y., Yang, X., Luo, X., Jiang, L. and Deng, M. 2021. Original Hosts, Clinical Features, Transmission Routes, and Vaccine Development for Coronavirus Disease (COVID-19). *Frontiers in Medicine*, 8: 702066.

Xie, Q., Huang, Q., Wang, J., Li, N., Tan, G. and Zhou, J. 2013. Clinical features of cluster headache: an outpatient clinic study from China. *Pain Medicine*, 14(6): 802–807.

Xie, W., Li, R., He, M., Cui, F., Sun, T., Xiong, J., Zhao, D., Na, W., Liu, R. and Yu, S. 2020. Prevalence and risk factors associated with headache amongst medical staff in South China. *The Journal of Headache and Pain*, 21(5). Available: <u>https://doi.org/10.1186/s10194-</u> <u>020-1075-z</u> (Accessed 20 June 2021) Yu, S., Liu, R., Zhao, G., Yang, X., Qiao, X., Feng, J., Fang, Y., Cao, X., He, M. and Steiner, T. 2012. The prevalence and burden of primary headaches in China: a population-based door-to-door survey. *Headache*, *52(4):* 582–591.

Zhu, F.C., Li, Y.H., Guan, X.H., Hou, L.H., Wang, W.J., Li, J.X., Wu, S.P., Wang, B.S., Wang, Z., Wang, L., Jia, S.Y., Jiang, H.D., Wang, L., Jiang, T., Hu, Y., Gou, J.B., Xu, S.B., Xu, J, J., Wang, X.W., Wang, W. and Chen, W. 2020. Safety, tolerability, and immunogenicity of a recombinant adenovirus type-5 vectored COVID-19 vaccine: a dose-escalation, open-label, non-randomised, first-in-human trial. *Lancet*, 395(10240): 1845–1854.

Zwart, J. A., Dyb, G., Hagen, K., Ødegård, K. J., Dahl, A. A., Bovim, G., & Stovner, L. J. 2003. Depression and anxiety disorders associated with headache frequency. *The Nord-Trøndelag Health Study. European journal of neurology*, 10(2): 147–152. Available: <u>https://doi.org/10.1046/j.1468-1331.2003.00551.x</u> (Accessed 20 September 2022).

# **APPENDICES**

## Appendix 1: Permission from the Director of Research

Miss D Chetty 141 Bayat Road Orient Heights Pietermaritzburg

Prof K Motaung Director: Research and Post Graduate Support Research, Innovation and Engagement Durban University of Technology Durban

#### **Request for Permission to Conduct Research**

Dear Prof Motaung

My name is Deidré Chetty and I am currently pursuing my Master's degree in Chiropractic at the Durban University of Technology. For my dissertation, the research I wish to conduct is titled: The prevalence, selected risk factors and burden of primary headaches in lecturers at a University of Technology in KwaZulu-Natal during the COVID-19 pandemic."

I hereby seek your consent to access lecturers at the Durban University of Technology, in order to collect data. Due to COVID-19, the data collection will be done using an online survey.

I have provided you with a copy of my proposal which includes copies of the data collection tools and consent forms to be used in the research process, as well as a copy of the approval letter which I received from the Institutional Research Ethics Committee (IREC).

If you require any further information, please do not hesitate to contact me on 072 261 3621 or deidrechetty@outlook.com..

Yours sincerely,

Deidré Chetty Durban University of Technology

## Appendix 2: Ethics Clearance Certificate



Institutional Research Ethics Committee Research and Postgraduate Support DDUT-IRECtorate 2<sup>nd</sup> Floor, Berwyn Court Gate I, Steve Biko Campus Durban University of Technology

P O Box 1334, Durban, South Africa, 4001

Tel: 031 373 2375 Email: lavishad@dut.ac.za http://www.dut.ac.za/research/institutional\_research\_ethics

ww.dut.ac.za

12 August 2022

Ms D Chetty 141 Bayat Road Orient Heights Pietermaritzburg

Dear Ms Chetty

The prevalence, selected risk factors and burden of primary headaches in lecturers at a university of Technology (UoT) in KwaZulu-Natal during the COVID-19 pandemic Ethical Clearance number IREC 122/22

The DUT-Institutional Research Ethics Committee acknowledges receipt of your final data collection tool for review.

We are pleased to inform you that the data collection tool has been approved. Kindly ensure that participants used for the pilot study are not part of the main study.

In addition, the DUT-IREC acknowledges receipt of your gatekeeper permission letter.

Please note that FULL APPROVAL is granted to your research proposal. You may proceed with data collection.

Any adverse events [serious or minor] which occur in connection with this study and/or which may alter its ethical consideration must be reported to the DUT-IREC according to the DUT-IREC Standard Operating Procedures (SOP's).

Please note that any deviations from the approved proposal require the approval of the DUT-IREC as outlined in the DUT-IREC SOP's.

**Yours Sincerely** 

Prof J K Adam Chairperson: DUT-IREC



ENVISION 2030 transparency · honesty · integrity · respect · accountability fairness · professionalism · commitment · compassion · excellence



## Appendix 3: Permission Letter from Professor Timothy Steiner

Re: DUT Chiropractic Research Steiner, Timothy J <u>t.steiner@imperial.ac.uk</u> To deidre Chetty <u>deidrechetty@outlook.com</u> Tuesday 2022/02/01 12: 20

Hello. You have permission to use this questionnaire for the purposes of your research. Please acknowledge the original manuscript, and explicitly describe any adaptations and modifications, in your publications.

I hope all goes well.

Best wishes

**Timothy Steiner** 

From: Deidre Chetty <<u>deidrechetty@outlook.com</u>> Sent: 01 February 2022 07: 52 To: Steiner, Timothy J <<u>t.steiner@imperial.ac.uk</u>> Subject: DUT Chiropractic Research

Dear Professor Steiner

My name is Deidré Chetty and I am a Master's student at the Durban University of Technology in South Africa. My current research topic is titled : "The prevalence, selected risk factors and burden of primary headaches in lecturers at a University of Technology in KwaZulu-Natal during the COVID-19 pandemic."

My research objectives are:

- 1. To determine the prevalence and types of primary headaches suffered by lecturers at the University
- 2. To determine the socio-demographic, medical history, work-related and COVID-19 related risk factors for the primary headaches
- 3. To determine the personal impact of the primary headaches in terms of work, family and social life

I hereby request your permission to use the Headache-attributed restriction, disability, social handicap and impaired participation (HARDSHIP) questionnaire for the purposes of my research. If your permission is granted, your questionnaire will be adapted and modified to suit my research study.

Kind regards

Deidré Chetty

Get Outlook for Android

## Appendix 4: Letter of Information



#### LETTER OF INFORMATION

#### Dear Participant, I would like to welcome you and thank you for your interest in participating

**Title of the Research Study:** The prevalence, selected risk factors and the burden of primary headaches in lecturers at a University of Technology in KwaZulu-Natal during the COVID-19 pandemic

Principal Investigator/s/researcher: Deidre Chetty, MHsc Chiropractic student

Supervisor: Dr Penny Orton - PhD Nursing Collaborator: Dr Laura O'Connor - MTech: Chiropractic

#### Brief Introduction and Purpose of the Study:

Headaches in lecturers have not been well documented in South Africa, yet they play such an integral role in the lives of students. During the last two years, COVID-19 has forced lecturers to transition to online learning and adapt to a "new normal", which has placed an extra burden on them. This study is a quantitative descriptive cross-sectional survey which will provide information on the prevalence and the socio-demographic, medical history, workrelated and COVID-19 related risk factors for the primary headaches within the lecturer population at the Durban University of Technology (DUT). The personal impact that headaches have in terms of work, family and social life will also be investigated.

#### Aim

The aim of this study is to determine the prevalence, selected risk factors and burden of primary headaches in lecturers at a University of Technology (UoT) in KwaZulu-Natal during the COVID-19 pandemic

#### **Outline of the Procedures**

All willing participants who have completed the informed consent (see below) are encouraged to complete the questionnaire. Participation in the study will not interrupt participants as it will take 10-15 minutes to complete. Participants are free to withdrawn from the study at any point, even while completing the questionnaire, if they feel any discomfort at any stage

#### **Risks or Discomforts to the Participant:**

There are no foreseeable risks or discomfort if you participate in this study

#### Explain to the participant the reasons he/she may be withdraw from the Study:

You may withdraw from the study if any of the questions make you feel uncomfortable. There will be no adverse consequences if you choose to withdraw from the study

#### **Benefits:**

You may find comfort in talking about your headache experiences. Your full co-operation will also add to the existing body of knowledge about headaches, therefore this will benefit lecturers in terms of creating awareness and the chiropractic profession and other health care providers.

#### **Remuneration:**

No remuneration will be offered for participating in this study

#### Costs of the Study:

There are no costs associated with participating in this study

#### **Confidentiality:**

All information gathered by this study is confidential. The questionnaire will be analysed by a statistician and all information will only be used for research purposes. No personal information will be published in the final dissertation of this study. All personal information that relates to your participation in this study will be treated confidentially and answers to the questionnaire will be processed anonymously.

#### **Research-related Injury:**

Not applicable to this study as it is an online questionnaire based study

#### Storage of all electronic and hard copies including tape recordings

This will be kept for five years and discarded afterwards

#### Persons to contact in the Event of Any Problems or Queries:

The researcher, Deidre Chetty	deidrechetty@outlook.com or 072 261 3621
My supervisor, Dr Penny Orton	- pennyo@dut.ac.za
My collaborator, Dr O'Connor	lauraw@dut.ac.za

You may also contact the DUT Institutional Research Ethics administrator on 031 373 2375. Any related complaints may be reported to the Director: Research and Postgraduate Support Dr L Linganiso on 031 373 2577 or researchdirector@dut.ac.za.

Your assistance is greatly appreciated

Deidre Chetty

## Appendix 5: Informed Consent



## CONSENT

**Full Title of the Study:** The prevalence, selected risk factors and the burden of primary headaches in lecturers at a University of Technology in KwaZulu-Natal during the COVID-19 pandemic

Names of Researcher/s: Deidre Chetty

#### Statement of Agreement to Participate in the Research Study:

□ I hereby confirm that I have been informed by the researcher, <u>Deidre Chetty</u> researcher), about the nature, conduct, benefits and risks of this study - Research Ethics Clearance Number: I<u>REC 122/22</u>,

□ I have also received, read and understood the above written information (Participant Letter of Information) regarding the study.

□ I am aware that the results of the study, including personal details regarding my sex, age, date of birth, initials and diagnosis will be anonymously processed into a study report.

□ In view of the requirements of research, I agree that the data collected during this study can be processed in a computerised system by the researcher. □ I may, at any stage, without prejudice, withdraw my consent and participation in the study.

□ I have had sufficient opportunity to ask questions and (of my own free will) declare myself prepared to participate in the study.

□ I understand that significant new findings developed during the course of this research which may relate to my participation will be made available to me.

Full Name of Participant	Date	Date			Time			Initials	
I. Deidre Chetty (name of researche	r) herewith	confirm	that	the	above	participant	has	been	

Deidré Chetty

Full Name of Researcher

Date

fully informed about the nature, conduct and risks of the above study.

Signature

# Appendix 6: Questionnaire Focus Group

## **SECTION A**

Demographics: Please	e fill in or plac	e an X	where r	ecess	ary					
What is your sex?	Male		Fema	ale						
What is your age?	years									
What is your weight?	kg									
What is your height?	m									
What is your race?	Black V	/hite	Indiar	۱		Coloure	ed		Other	
Social history: Please	place an X wh	nere ne	cessary	1						•
Do you smoke tobacco?	?	Ye	es			No				
If yes, how often?		D	Daily		١	Veekly		Occasi	onally	
Do you drink alcohol?		Ye	es		1	No				
If yes, how many units do you drink per week? 1 unit = 350ml of beer or 1 small glass of wine or 1 tot of spiritunit (a double tot = 2 units)							units			
Do you usually eat three	e meals a day?	Ye	es			No				
How many hours of sl per night?	eep do you ge	et 1-	4		5	5-9		>10	)	
Medical history: Pleas	se place an X o	or fill in	where	necess	sary					
Have you had a head in 3 months?	njury in the last	Ye	es				No			
If yes, have you had tre	atment?	Ye	es				No			
Do you wear glasses?		Ye	es				No			
When last were your ey	es tested?									
Please tick the appro	priate box/es	High pres	sure	blood			Low press	ure	bloo	d
have a history with any	the following:	Diab	etes				Cerek	oral stro	ke	
(you may tick more that	n 1 box)	Anae	emia				Thyrc	oid disea	ase	
		Cano	cer				Seizu	res		
Other?										
Please specify										
Psychological: Please	e place an X w	here ne	ecessary	/			1			
Are you usually highly s	stressed?				<u> </u>	es		No		
Do you nave panic atta	CKS?	aiana			<u>Y</u>	es		NO		
Have you been diagnos	sed with appres	sion ?			Y V	es loc		NO		
nave you been diagnos	sed with anxiety	<u>'</u>			ľ	62		NO		
Have you had treatment by a psychiatrist or counsellor?			Y	es		No				

Work related: Please answer the following q	uestions and	d place >	K where ne	ecess	ary		
How long have you been a lecturer?	years						
Indicate the nature of your lecturing appointment	Full time Part time Contract						
How many days per week do you lecture?			days/we	ek			
How many hours do you lecture per day?			hours/d	lay			
Describe the manner in which you have lecture	d over	Online I	ectures				
the last year:		Face to	face theor	y lectu	ures		
		Face to	face pract	ical le	ctures		
lecture preparation?	Der week on						
Do you hold another occupation?		Yes		No			
If yes, please state the occupation:							
If you are feeling unwell/ in pain, do you have I support?	ine manager	Yes		No			
Are you working remotely, on campus or both?	Remotely		On campus		Both		
Did you experience any difficulty working from h	Yes		No				
How many hours do you spend sitting at a desk	hours/day						
Do you work at a desk when working at home?		Yes No					
If yes, is your desk height appropriate for your w	Yes No						
lf no: where do you work:							
Dining room table		Yes		No			
Sofa		Yes		No			
On the bed		Yes		No			
Other:				•			
Is your chair comfortable for you?		Yes		No			
Are your feet supported?		Yes		No			
Is your laptop/computer screen at the same le eyes?	evel as your	Yes		No	1		
How many hours do you spend using a laptop/c day?	computer per	fhours/day					
How many hours do you spend using a phone a		hc	ours/d	lay			
Does your work interfere with family or leisure ti	me?	Yes	N	0			
COVID-19 related questions: Please place ar	n X where ne	cessary					
Have you tested positive for COVID-19?		Yes		Ν	0		
If yes, do you have any side effects from COVIE	)-19?	Yes		N	0		
Have you ever experienced feelings of lonelines	ss during the	Yes		N	0		
Has your workload increased since COVID-19?		Yes		Ν	0		

## Section B

## Screening: Please tick or fill in where necessary

Have you ever had a headache? Yes 
No 
If yes, please answer the rest of the questionnaire. If no, you are now complete. Thank you for participating

Have you had a headache during the last 12 months?	Yes 🗆 No 🗆	
On how many days in the last 3 months did you have a head	lache?	days
Have your headaches worsened since COVID-19?	Yes □	No 🗆
Do you think wearing a mask aggravates your headache?	Yes 🗆	No 🗆

Daily headache questions: Please fill in where necessary

How long do these headaches usually last?

(Please enter the amount of minutes or hours or tick the box)

\_\_\_\_\_minutes or \_\_\_hours or □ never goes away

Do you take medication to treat it? Yes Do No Do

If yes, what medication do you use the most often to treat your headache? Please list it here

In the last 30 days, how many days did you take medication for your headache?

\_\_\_\_Days

Diagnostic questions: Please fill in or tick	where necessary
How often do you have <b>this type of</b> <b>headache</b> ? (please tick box or enter the number of days per month or per year)	every days/mont days/year day
How long does this type of headache usually last? (please enter the number of minutes, hours or days, or tick the box)	mins,hours ordays never goes away □
Is your last answer with or without medication?	with  without
How long would it last <b>if you did not take</b> <b>medication</b> ? (please enter the number of minutes, hours or days)	mins,hours ordays
How bad is <b>this type of headache</b> usually? (please tick one box)	not quite bad bad very bad
<ul> <li>During a headache, do one or more of these happen to you?</li> <li>Your eye turns red on one side only</li> <li>Your eye waters on one side only</li> <li>Your nose runs on one side only</li> <li>Your nose gets congested on one side only</li> </ul>	Yes No
There are many ways of describing a headache, but most are either throbbing or pressing. Thinking still of <b>this type of headache</b> , which best describes the pain? (please tick one box)	Throbbing pressing, or pulsating squeezing (this means varying in or time with the heart beat) tightening
Is this the worst pain you have ever experienced?	no 🗆 yes 🗆
Is the pain of <b>this type of headache</b> usually on only one side of the head? (please tick one box)	no 🗆 yes 🗆
Does exercise (like walking or climbing stairs) tend to make it worse? (please tick one box)	no 🛛 yes 🗆
Thinking still of <b>this type of headache</b> , how does it affect your ability to do day-to-day activities? (please tick one box)	can do cannot do can do everything some things nothing as normal
With this type of headache, do you usually feel nauseated (as though you may vomit or throw up)? (please tick one box)	no 🗆 yes 🗆
With <b>this type of headache</b> , do you usually actually vomit (throw up)? (please tick one box)	no 🗆 yes 🗆
When you have <b>this type of headache</b> , does daylight or other lighting bother you? In other words, do you prefer to be in the dark? (please tick one box)	no not sure yes (this question refers to <u>ordinary</u> levels of light, not bright lighting)
When you have <b>this type of headache</b> , does noise bother you? In other words, do you prefer to be in the quiet? (please tick one box)	no not sure yes (this question refers to <u>ordinary</u>
n na ta ta se no na sel la stata de la tata d	levels of holse, not very loud holse)

## Section C

Impact questions The next questions are about the effects your headaches have on y	your own	life.		
Have your headaches interfered with your work?	no 🗆	yes□		
On how many days in the last 3 months did you skip work due to your headache?	days			
On how many days in the last 3 months could you do less than half of your usual work due to your headache?	days			
Do you believe your headaches have made you less successful in your career?	no⊡ yes⊡			
Do you feel that your family and friends understand and accept your headaches?	🗆 no	□ partly	□ yes, fully	
Do you avoid telling people that you have headaches?		no 🗆	yes 🗆	
Have your headaches ever caused a long-term relationship or partnership to break down?		no 🗆	yes 🗆	
On how many days in the last 3 months did you miss family or leisure activities because of your headache?		da	ays	
When you have a headache at work, what do you usually do?				
When you have a headache at home, what do you usually do?				

Thank you for taking the time to answer all the questions!

# Appendix 7: Final Questionnaire

## Section A

Demographics: Plea an appropriate optic	ise com on	plete	e the fol	low	ving questions	s k	by filling in an ans	wer or choosing
What is your age?	)	/ears	5					
What is your sex?	Male				Female			
What is your race?	Black		White		Indian		Coloured	Other
What is your marital status?	Single	М	arried	[	Divorced		Widowed	Other

Social History: Please choose an appropriate option							
Do you smoke tobacco?	Yes		No				
If yes, how often do you smoke?	Daily		Weekly		Occasionally		
Do you drink alcohol?	Yes		No				
If yes, how many units do you drink per week? 1 unit = 350ml of beer or 1 small glass of wine or 1 tot of spirit (a	1-3 units 4-6 units 7-9 units >10 unit	S S					
double tot = 2 units)							
Do you usually eat three meals a day?	Y e s		No				
On average how many hours of sleep do you get per night?	1 - 4		5-9		> 1 0		
Do you exercise?	Yes		No		Occasionally		
If yes, how many hours per week do you exercise for?	1-4		5-9		>10		

Medical History: Please complete the following questions by filling in an answer or choos	ing
an appropriate option	

an appropriate option					
Have you had a head injury in the last 3 months?	Yes		No	5	
Do you wear glasses?	Yes		N		
In general, how would you describe your health?	Excellent	Very good	Good	Fair	Poor
	High blood pressure		Low blo	·e 🗆	
Please tick the appropriate	Diabetes		Cerebra		
with or if you have a history with	Anaemia		Thyroid	disease	
any of the following: (you may tick more than 1 box)	Cancer		Seizures	3	
,	Sinusitis		Hayfeve	r	
Other? Please specify	N/A 🗆				
1 ,					

Psychological: Please choose ONE option				
How would you rate your stress levels?	High	Moderate	Mild	Non e
Have you been diagnosed with depression?	Yes		No	
Have you been diagnosed with anxiety?	Yes		No	

Work related: Please choose an appropriate option where necessary								
How long have you been a lecturer?	0-3 years	4-8	years	9-1:	3 years	>14 years		
Indicate the nature of your lecturing appointment?	Full time		Pa	rt tii	me	Contract		
How many days per week do you lecture? 1-3 days 4-5 days 6-7 days								

How many hours do you lecture per day?       < 1 hour       1-3 hours       > -         Describe the manner in which you have lectured over       Online lectures       > -         The last year:       Face to face theory lectures	4 hours	
Describe the manner in which you have lectured Online lectures over the last year:		
over Face to face theory lectures		
the last year:		
You may choose more than one option Face to face practical lectures		
If you are feeling unwell/ in pain, do you have line <b>Yes No</b>		
Are you working remotely on campus or both? Remotely On campus B	Both	
Did you experience any difficulty with the transition to working from home during COVID-19?		
Now that things are slowly getting back to normal, how are you coping?	ping well	I
How many hours do you spend sitting at a desk per <1 hou 1-4 hours 5-9 hours > day?	10 hour	rs
Do you work at a desk when working from home? Yes No		
If yes, is your desk height appropriate for your work? Yes No		
Is your chair comfortable for you? Yes No		
Is your laptop/computer screen at the same level as your eyes? No		
How many hours do you spend using a < 1 hour 1-4 hours 5-9 hours > laptop/computer per day?	>10 houi	rs
How many hours do you spend using a phone a < 1 hour 1-4 hours 5-9 hours > day?	10 hour	rs
Does your work interfere with family or leisure time? Often Sometimes N	Never	

COVID-19 related questions: Please choose ONE option								
Have you tested positive for COVID-19?	Yes	No						
Did you ever experience feelings of loneliness during the lockdown?	Yes	No						
Has your workload increased during COVID-19?	Yes	No						

#### Section **B**

Screening: Please choose an appropriate option

Are you a headache sufferer? Yes □ No 🗆 If yes, please answer the rest of the questionnaire. If no, you are now complete. Thank you for participating. Please submit your questionnaire Have you had a headache during the last 12 months? Yes No 🗆 During the last 30 days, on how many of these days did you have a headache? 1-14 days□ >15days□ Have your headaches worsened during the pandemic? Yes 🗆 No 🗆 Do you think wearing a mask aggravates your headache? Yes □ No 🗆 Does anyone in your immediate family suffer from similar headaches? Yes No 🗆 Do you take medication to treat your headache? Yes 🗆 No 🗆

In the last 30 days, how many days did you take medication for your headache?

< 9 days 
□ >10 days □

Please think about your headaches.

Do you think they are all of one type, or are they of more than one type?

One 
More than one

If you answered one, the next questions are to diagnose the headache. If you answered more than one, please focus upon the headache type that on the whole bothers you the most.

Diagnostic questions: Please complete answer or choosing an appropriate op	e the questions below tion	/ by filling i	n an		
How long does this type of headache usually last?	□ <u>m</u> inutes □ hours never goes away [	□ days			
With regards to your last answer, is this when you have taken medication?	Yes 🗆	No 🗆			
How long would it last <b>if you did not take</b> medication?	hours days				
How bad is <b>this type of headache</b> usually?	Mild  Moderate  S	Severe 🗆 E	xtremely severe		
During a headache, do one or more of these - Your eye turns red on one side on - Your eye waters on one side only - Your nose runs on one side only - Your nose gets congested on or	Yes □	No 🗆			
Does your headaches occur in cycles (i.e. pattern)?	occurs in a routine	Yes 🗆 N	o 🗆		
Do you experience excruciating pain that is in, behind or around one eye, but may radia your face, head and neck?	Yes 🗆 N	o 🗆			
There are many ways of describing a headac throbbing or pressing. Thinking still of <b>this type of headache</b> , wh pain?	Throbbin pulsating (this mear time with tightening	g pressing, or squeezing as varying in or the heart beat) g			
Is the pain of <b>this type of headache</b> usually head?	on only one side of the	no 🗆	yes 🗆		
Does exercise (like walking or climbing stai worse?	rs) tend to make it	no 🗆	yes □		
With <b>this type of headache</b> , do you usua though you may vomit or throw up)?	no 🗆	yes □			
With <b>this type of headache</b> , do you usually up)?	no 🗆	yes 🗆			
When you have <b>this type of headache</b> , or lighting bother you? In other words, do you pr (this question refers to <u>ordinary</u> levels of ligh	□ no	□ yes			
When you have <b>this type of headache</b> , doe In other words, do you prefer to be in the qui (this question refers to <u>ordinary</u> levels of nois	□ no	□ yes			

### Section C

Impact questions	
The next questions are about the effects your headac	hes have on <b>your own life</b> .
Have your headaches interfered with your work?	no □ yes□
On how many days in the last 3 months did you miss work due to your headache?	None 1-14 15-30 >31 days
Do you feel that your family and friends understand and accept your headaches?	□ □ □ no partly fully
Do you avoid telling people that you have headaches?	no 🗆 yes 🗆
Have your headaches ever caused a long-term relationship or partnership to break down?	no 🗆 yes 🗆
On how many days in the last 3 months did you miss family or leisure activities because of your headache?	None 1-14 15-30 >31 days
When you have a headache at work, what do you usually do?	Take medication □ Go home□ Other □
When you have a headache at home, what do you usually do?	Take medication □ Lie down □ Sleep □
If other, please specify what you do	Other 🗆
Do you think your headaches have had a negative effect on your mental health?	no 🗆 yes 🗆

## Thank you for taking the time to answer all the questions!

Appendix 8: Diagnostic Algorithm

**Lifting The Burden** In official relations with the World Health Organization The Global Campaign against Headache

# Diagnostic algorithm for headache according to the diagnostic questions in the HARDSHIP questionnaire

Responses to the diagnostic questions of the HARDSHIP questionnaire, for each bothersome headache separately identified by the participant, should be entered into this algorithm.

The algorithm must be applied from the beginning in each case.

#### Begin here:

Flags for manual review	HARDSHIP question	Response	Value	Action
	14	Headache on ≥15	0	Continue with migraine algorithm
		days/month	1	Continue below
	18	Acute medication on ≥10 days/month	0	Not medication-overuse headache (MOH); manually review for final diagnosis
			1	Possible MOH; manually review for final diagnosis

Migraine algorithm	HARDSHIP question	Response	Value	Calculate 1 <sup>st</sup> intermediate score	Calculate 2 <sup>nd</sup> intermediate score	Final score and diagnosis
	21/23	< 4 h	0			0
В		> 3 d (72 h)	0			0
		≥ 4 h and ≤ 3 d (72 h)	1			1
С	24	Not bad	0			
		Quite bad	1			
		Very bad	1			
	25	Pulsating/throbbing yes	1	04.05.00.07 > 0		4
		Pressing/tightening yes	0	24+25+26+27 < 2 24+25+26+27 < 2		0
	26	One side	1			
		Both sides	0			
	27	Worsened by physical activity yes/no	1/0			
D	29	Nausea yes/no	1/0	N+V ≥ 1: 1		
	30	Vomiting yes/no	1/0	N+V = 0: 0	n/v+pt/pn ≥ 1	1
	31	Photophobia (pt) yes/no	1/0	Pt+pn = 2: 1	n/v+pt/pn = 0	0
	32	Phonophobia (pn) yes/no	1/0	Pt+pn ≤ 1: 0		
					B+C+D=3	Migraine
					B+C+D < 3	Not migraine, go to <b>TTH</b> algorithm

TTH algorit hm	HARDSHIP question	Response	Value	Calculate 1 <sup>st</sup> intermediate score	Calculate 2 <sup>nd</sup> intermediate score	Final score and diagnosis
	21/23	< ½ h	0			0
В		> 7 d	0			0
		≥ ½ h and ≤ 7 d (168 h)	1			1
С	24	Not bad	1			
		Quite bad	1			
		Very bad	0			
	25	Pulsating/throbbing yes	0	2/1+25+26+27 > 2		1
		Pressing/tightening yes	1	24+25+26+27 < 2		0
	26	One side	0			
		Both sides	1			
	27	Worsened by physical activity yes/no	0/1			
D	29	Nausea yes/no	1/0	N+V ≥ 1: 0		
	30	Vomiting yes/no	1/0	N+V = 0: 1	n/v+pt/pn = 2	1
	31	Photophobia (pt) yes/no	1/0	Pt+pn ≤ 1: 1	n/v+pt/pn < 2	0
	32	Phonophobia (pn) yes/no	1/0	Pt+pn = 2: 0		
					B+C+D = 3	ТТН
					B+C+D < 3	Not TTH, go to probable migraine algorithm.

Probable migraine algorithm	HARDSHIP question	Response	Value	Calculate 1 <sup>st</sup> intermediate score	Calculate 2 <sup>nd</sup> intermediate score	Final score and diagnosis
	21/23	< 4 h	0			0
В		> 3 d (72 h)	0			0
		≥ 4 h and ≤ 3 d (72 h)	1			1
С	24	Not bad	0			
		Quite bad	1			
		Very bad	1			
	25	Pulsating/throbbing	1			
		Pressing/tightening	0	$24+25+26+27 \ge 2$ 24+25+26+27 < 2		1
	26	One side	1	24+23+20+21 < 2		0
		Both sides	0			
	27	Worsened by	1/0			
		Activity yes/no				
D	29	Nausea yes/no	1/0	N+V ≥ 1: 1		
	30	Vomiting yes/no	1/0	N+V < 1: 0	n/v+pt/pn ≥ 1	1
	31	Photophobia (pt)	1/0	Pt+pn = 2: 1	n/v+pt/pn < 1	0
	32	Phonophobia (pn)	1/0	Pt+pn ≤ 1: 0		
					B+C+D = 2	Probable migraine
					B+C+D < 2	Not pMig, go to probable TTH algorithm

Probable TTH algorithm	HARDSHIP question	Response	Value	Calculate 1 <sup>st</sup> intermediate score	Calculate 2 <sup>nd</sup> intermediate score	Final score and diagnosis
	21/23	< ½ h	0			0
В		> 7 d	0			0
		≥ ½ h and ≤ 7 d (168	1			1
С	24	Not bad	1			
		Quite bad	1			
		Very bad	0			
	25	Pulsating/throbbing	0			
		Pressing/tightening	1	$24+25+26+27 \ge 2$		1
	26	One side	0	24+23+20+27 < 2		0
		Both sides	1			
	27	Worsened by physical activity yes/no	0/1			
D	29	Nausea yes/no	1/0	N+V ≥ 1: 0		
	30	Vomiting yes/no	1/0	N+V = 0: 1	n/v+pt/pn = 2	1
	31	Photophobia (pt) ves/no	1/0	Pt+pn ≤ 1: 1	n/v+pt/pn < 2	0
	32	Phonophobia (pn) ves/no	1/0	Pt+pn = 2: 0		
					B+C+D = 2	Probable TTH
					B+C+D < 2	Not pTTH, undetermined headache